2011 DATA B00K



Illinois Student Assistance Commission



TABLE OF CONTENTS

INTRODUCTION	v
PART ONE - ISAC APPROPRIATION HISTORY	1
Table 1.0 Appropriation History, FY1980-FY2011	3
Table 1.1 Summary of FY2011 Program Expenditures and Recipients	7
PART TWO - MONETARY AWARD PROGRAM	9
Table 2.0a Historical Awards (total applications, announced eligible, enrolled) and Payout Summary, FY1997-FY2011	11
Table 2.0b FY2011 Monetary Award Program Formula	12
Table 2.0c Monetary Award Program Maximum Award History, AY1978-AY2012	13
Table 2.0d MAP/IIA Suspension History AY1978-AY2012	14
Sector Statistics	
Table 2.1 Historical Enrolled Awards and Payout Summary by Sector, FY1980-FY2011	17
Table 2.2 Weighted Mean Tuition and Fees and MAP College Budgets by Sector, FY1980-FY2011	21
Table 2.3a Summary of Awards and Payout by Sector, FY2007 to FY2011	22
Institution Statistics	
Table 2.3b Summary of Awards and Payout by Institution, FY2008 to FY2011	23
Table 2.3c MAP Recipients by Dependency Status and Proportion of Recipients Paid by Institution, FY2011	30
Table 2.3d Average Income by Dependency Status by Institution, FY2011	36
Table 2.3e Tuition and Fees at MAP Approved Institutions, AY2010-AY2011	42

Eligibility by Income and Dependency Status	Page
Table 2.4a Historical Summary, FY2007-FY2011	49
Table 2.4b Eligibility at Public Four-Year Institutions, FY2011	50
Table 2.4c Eligibility at Public Two-Year Institutions, FY2011	51
Table 2.4d Eligibility at Private Institutions, FY2011	52
Table 2.4e Eligibility at Proprietary Institutions, FY2011	53
Applicant Characteristics	
Table 2.5a Historical Summary by Class Level, FY2007-FY2011	55
Table 2.5b Historical Summary by Age, FY2007-FY2011	55
Table 2.6a Historical Summary of Announced Eligible Dependent MAP Applicants, FY2007-FY2011	56
Table 2.6b Historical Summary of Announced Eligible Independent MAP Applicants, FY2007-FY2011	57
Table 2.6c Combined Announced Eligible Dependent/Independent Combined MAP Applicants, FY2007-FY2011	58
Table 2.6d Characteristics of Paid Dependent MAP Applicants, FY2007-FY2011	59
Table 2.6e Characteristics of Paid Independent MAP Applicants, FY2007-FY2011	60
Table 2.6f Characteristics of Paid Dependent/Independent Combined MAP Applicants, FY2007-FY2011	61
PART THREE - FEDERAL FAMILY EDUCATION LOAN PROGRAMS – TABLES DISCONTINUED	63
PART FOUR - ILLINOIS DESIGNATED ACCOUNT PURCHASE PROGRAM — TABLES DISCONTINUED	65

PART FIVE - SPE	CIALIZED SCHOLARSHIP, GRANT, AND WAIVER PROGRAMS
Table 5.0a	Specialized Scholarship and Grant Programs - Award and Payout Summary by Program, FY2007-FY2011
Table 5.0	Specialized Scholarship and Grant Programs - Award and Payout Summary by Sector, FY2007-FY2011
Table 5.0 0	Minority Teachers of Illinois Scholarship Program, ISAC/Future Teachers Corp Program, IFTC/Golden Apple, and the Nurse Educator Scholarship Program - Recipients and Payout by Institution, FY2011
Table 5.0 c	Illinois Veteran Grant Illinois National Guard Grant Program, Grant Program for Dependents of Police/Fire/Correctional Officers, and the Student-to-Student Grant Program - Recipients and Payout by Institution, FY2011
Table 5.1	Historical Summary of the Robert C. Byrd Honors Scholarship Program - Recipients and Payout by Institution, FY2008-FY2011.86
Table 5.2	Historical Summary of the State Scholar Program, AY1979-80-AY2011-12
Table 5.3	Historical Summary of the Bonus Incentive Grant Program - Summary of Recipients and Payout, FY2002-FY201190
Table 5.4a	Silas Purnell Illinois Incentive for Access Program - Summary of Awards and Payout by Sector, FY2007-FY2011
Table 5.4k	Silas Purnell Illinois Incentive for Access Program - Summary of Awards and Payout by Institution, FY2008-FY2011
<u>Table 5.5</u>	Illinois Special Education Teacher Tuition Waiver Program - Summary of Recipients, FY1996-FY2011
Table 5.6	Illinois Teachers Loan Repayment Program, FY2003-FY2011
Table 5.7	Higher Education License Plate Program - Summary of Recipients and Payout, FY1998-FY2011
Table 5.8	Illinois Optometric Education Scholarship Program - Summary of Recipients and Payout, FY2004-FY2011
<u>Table 5.9</u>	Nurse Educator Loan Repayment Program, FY2008-FY2011
Table 5.10	Veterans' Home Nurse Loan Repayment Program, FY2008-FY2011
Table 5.1 1	Nurse Educator Scholarship Program, FY2007-FY2011
Table 5.12	2 John R. Justice Student Loan Repayment Program, FY2011

	Page
PART SIX - COLLEGE ILLINOIS!® PREPAID TUITION PROGRAM	105
Table 6.0 College Illinois/® Prepaid Tuition Program - Enrollment and Payout FY2007-FY2011	107
PART SEVEN – ISAC OUTREACH OVERVIEW	109
Table 7.0 Illinois Student Assistance Commission Statewide Outreach Activities FY2009-FY2011	111



ILLINOIS STUDENT ASSISTANCE COMMISSION FY2011 DATA BOOK

Introduction

The Illinois Student Assistance Commission (ISAC) Data Book is completed annually by the Research, Planning, and Policy Analysis Department. It is a collection of tables containing information regarding student financial aid programs administered by ISAC and is divided into seven sections. Part One provides information pertaining to the appropriation history of the agency and summary data for ISAC administered programs for the previous fiscal year. Part Two focuses on the Monetary Award Program (MAP), the State's largest need-based grant program. Beginning with the 2011 ISAC Data Book, Parts Three and Four have been discontinued. Part Three, provided information on the Federal Family Education Loan Programs and Part Four, contained information on the Illinois Designated Account Purchase Program (IDAPP), ISAC's division for federal and private educational loans. Due to the enactment of the Health Care and Education Reconciliation Act of 2010, ISAC no longer originates or services FFELP loans. Colleges now process new loans through the U.S. Department of Education's William D. Ford Federal Direct Loan Program. The next section of the Data Book, Part Five, focuses on specialized scholarship and grant programs administered by ISAC. These programs include: the Illinois Veteran Grant Program, the Minority Teachers of Illinois Scholarship Program, the Illinois Future Teachers Corp Program, the Illinois National Guard Grant Program, the Dependents Grant Programs, the Student-to-Student Grant Program, the Robert C. Byrd Honors Scholarship Program, the Illinois State Scholar Program, the College Savings Bond Bonus Incentive Grant Program, the Silas Purnell Illinois Incentive for Access Program, the Illinois Special Education Teacher Tuition Waiver Program, the Illinois Teachers Loan Repayment Program, the Golden Apple Illinois Scholars Program, the Illinois Optometric Education Scholarship Program, the Nurse Educator Scholarship Program, the Nurse Educator Loan Repayment Program, Veterans' Home Nurse Loan Repayment Program, and the John R. Justice Student Loan Repayment Program. The next section, Part Six, provides information on the College Illinois!® Prepaid Tuition Program. The last section, Part Seven, provides an overview of ISAC's statewide outreach activities.

Each year ISAC receives a state appropriation with authority to expend funds from state and federal sources, including the Federal Student Loan Fund (FSLF). In FY2011, ISAC's state appropriation totaled \$821,701.6. This appropriation included \$441,324.1 in state funds, authority to spend \$362,977.5 in federal student loan funds, and authority to spend \$17,400,000 in federal program funds. The dollars appropriated from these funds are used by ISAC to provide financial aid for students. Table 1.0 shows ISAC's appropriation history by program from FY1980 through FY2011. Table 1.1 shows a summary of program expenditures, and recipients for state fiscal year 2011 (July 1, 2010 – June 30, 2011).

PART ONE -- APPROPRIATION HISTORY

Table 1.0 of the 2011 ISAC Data Book ISAC Appropriation History (\$ in thousands)

	Monetar	y Award Pr	ogram		Illinois	Higher Ed			Dependent	S	Veteran	Optometric	Merit
Fiscal		SSIG/		•	Incentive	License Plate		Robert	Grant	National	Grant	Ed Schlrship	Recog.
<u>Year</u>	GRF/EAF	(S)LEAP	SLOF	MAP Plus	For Access	Program	Douglas	Byrd	Programs	<u>Guard</u>	Program	Program	<u>Scholar</u>
EV1090	\$70.051.6	¢4.150.0							\$44.2	¢150.0			
FY1980	\$79,051.6	\$4,150.0	•	•	•	•	•	•	\$44.2	\$150.0	•	•	•
FY1981	\$82,166.8	\$4,170.0	•	•	•	•	•	•	\$25.0	\$120.3	•	•	•
FY1982	\$87,496.3	\$4,250.0	•	•	•	•	•	•	\$30.0	\$150.0	•	•	•
FY1983	\$90,863.3	\$4,033.0	•	•	•	•	•	•	\$30.0	\$450.0	•	•	•
FY1984	\$101,155.4	\$3,240.9	•	•	•	•	•	•	\$39.6	\$500.0	•	•	•
FY1985	\$105,779.1	\$4,105.1	•	•	•	•	•	•	\$44.5	\$1,020.0	•	•	Φα του ο
FY1986	\$118,102.0	\$4,200.0	•	•	•	•	• •	•	\$50.0	\$1,400.0	·	•	\$2,500.0
FY1987	\$127,885.7	\$3,928.6	•	•	•	•	\$480.8	•	\$48.5	\$1,455.0	\$4,056.6	•	\$4,656.0
FY1988	\$131,198.4	\$4,200.0	•	•	•	•	\$900.0	•	\$73.5	\$1,396.8	\$4,274.2	•	\$2,256.0
FY1989	\$147,083.5	\$4,000.0	•	•	•	•	\$900.0	•	\$79.0	\$2,900.0	\$8,000.0	•	Φ0.100.0
FY1990	\$171,942.4 a	\$3,900.0	•	•	•	•	\$1,000.0	•	\$85.0	\$3,200.0	\$8,000.0	•	\$8,100.0 a
FY1991	\$178,349.9	\$3,200.0	•	•	•	•	\$1,300.0	•	\$85.0	\$3,600.0	\$10,200.0	•	\$4,500.0
FY1992	\$179,876.3	\$3,630.7	•	•	•	•	\$1,300.0	•	\$87.5	\$3,800.0	\$11,400.0	•	\$4,300.0
FY1993	\$197,731.8	\$4,200.0	•	•	•	•	\$1,300.0	•	\$92.0	\$3,800.0	\$12,000.0	•	\$2,200.0
FY1994	\$209,008.8	\$4,200.0	•	•	•	•	\$1,300.0		\$95.0	\$3,500.0	\$12,000.0	•	\$2,080.0
FY1995	\$239,652.3	\$4,200.0	•	•	•	•	\$1,300.0	\$1,000.0	\$95.0	\$3,800.0	\$19,700.0	•	\$2,200.0
FY1996	\$251,749.6	\$4,200.0	•	•	•	•	\$1,000.0	\$1,350.0	\$95.0	\$3,800.0	\$21,740.0	•	\$2,200.0
FY1997	\$262,407.5	\$2,600.0	•	•	\$9,000.0		\$1,000.0	\$1,750.0	\$95.0	\$3,800.0	\$21,800.0	•	\$2,200.0
FY1998	\$280,265.0	\$2,820.0	•	•	\$9,000.0	\$1,000.0	•	\$1,750.0	\$97.9	\$3,919.7	\$21,800.0	•	\$2,200.0
FY1999	\$308,512.0	\$1,498.0	•	•	\$8,500.0	\$50.0	•	\$1,750.0	\$150.0	\$3,900.0	\$22,000.0	•	\$2,275.0
FY2000	\$335,485.8 c	\$1,500.0	•	•	\$8,000.0	\$70.0	•	\$1,750.0	\$150.0	\$4,050.0	\$21,750.0	•	\$4,700.0
FY2001	\$355,090.8 d	\$2,000.0	•	•	\$8,000.0	\$70.0	•	\$1,800.0	\$220.0	\$4,325.0	\$21,000.0	•	\$7,000.0
FY2002		\$3,100.0	•	•	\$8,000.0		•	\$1,800.0	\$250.0	\$4,500.0	\$19,250.0	•	\$5,300.0 g
	1 \$329,522.8	\$3,700.0	•	•	\$7,200.0	\$70.0	•	\$1,800.0	\$275.0	\$4,500.0	\$19,250.0	•	\$5,400.0
FY2004	\$338,699.8	\$3,700.0 i	•	•	\$7,200.0	\$70.0	•	\$1,800.0	\$275.0	\$4,500.0	\$19,250.0	\$50.0	\$5,400.0
FY2005	\$338,699.8	•	•	•	\$7,200.0	\$70.0	•	\$1,800.0	\$350.0	\$4,500.0	\$19,250.0	\$50.0	\$5,400.0
FY2006	\$346,699.8	\$3,700.0	•	•	\$7,200.0	\$70.0	•	\$1,800.0	\$350.0	\$4,500.0	\$19,250.0	\$50.0	•
FY2007	\$354,259.8	\$3,700.0	\$26,840.0	\$34,400.0	\$8,200.0	\$70.0	•	\$1,800.0	\$470.0	\$4,500.0	\$19,250.0	\$50.0	•
FY2008	\$381,099.8	\$3,700.0	•	•	\$8,200.0	\$70.0	•	\$1,800.0	\$470.0	\$4,480.0	\$19,250.0	\$50.0	•
FY2009	\$381,099.8	\$4,200.0	•	•	\$8,200.0	\$70.0	•	\$3,000.0	\$470.0	\$4,480.0	\$19,250.0	\$50.0	•
FY2010	\$388,102.2	\$4,000.0	•	•	\$4,800.0	\$70.0	•	\$3,000.0	\$710.2	\$4,741.7	\$16,842.5	\$50.0	•
FY2011	\$403,488.7	\$4,000.0	•	•	•	\$70.0	•	\$3,000.0	\$950.0	\$2,700.0	•	\$50.0	•

Table 1.0, Appropriation History (\$ in thousands), continued 2011 ISAC Data Book

Fiscal <u>Year</u>	Academic Scholar Program	Illinois Opportunity <u>Programs</u>	Minority Teachers <u>Scholarships</u>	Arthur F. Quern IT <u>Grant</u>	Bonus Incentive <u>Grant</u>	Student -to- <u>Student</u>	IFTC Tchr Shrtg <u>Scholarship</u>	Teacher Loan <u>Forgive</u>	Nurse Educator Scholarship Prog	Forensic Science Prog Grant	Illinois Scholars
FY1980	•	•	•	•	•	\$227.5		•		•	•
FY1981	\$2,000.0	•	•	•	•	\$250.0	•	•	•	•	•
FY1982	•	•	•	•	•	\$275.0	•	•	•	•	•
FY1983	•	•	•	•	•	\$275.0	•	•	•	•	•
FY1984	•	•	•	•		\$350.0	•	•	•	•	•
FY1985	•	•	•	•	•	\$350.0	•	•	•	•	•
FY1986	•	•	•	•		\$350.0	•	•	•	•	•
FY1987	•	•	•	•	•	\$388.0	•	•	•	•	•
FY1988	•	•	•	•		\$372.5	•	•	•	•	•
FY1989	•	•	•	•	•	\$538.1	•	•	•	•	•
FY1990	•	\$5,000.0 b	•	•		\$550.0	•	•	•	•	•
FY1991	•	\$3,254.6	•	•	•	\$550.0	•	•	•	•	•
FY1992	•	\$1,587.1	\$500.0	•	\$26.0	\$800.0	•	•	•	•	•
FY1993	•	\$1,287.1	\$500.0	•	\$26.0	\$800.0	•	•	•	•	•
FY1994	•	\$902.1	\$1,320.0	•	\$111.0	\$800.0	•	•	•	•	•
FY1995	•	•	\$1,500.0	•	\$175.0	\$800.0	\$997.3 J	•	•	•	•
FY1996	•	•	\$1,850.0	•	\$290.0	\$800.0	\$1,245.8	•	•	•	•
FY1997	•	•	\$1,950.0	•	\$375.0	\$900.0	\$1,345.8	•	•	•	•
FY1998	•	•	\$2,100.0	•	\$440.0	\$1,000.0	\$1,345.8	•	•	•	•
FY1999	•	•	\$2,200.0	•	\$420.0	\$1,000.0	\$1,445.8	•	•	•	•
FY2000	•	•	\$2,400.0	•	\$470.0	\$1,000.0	\$1,645.8	•	•	•	•
FY2001	•	•	\$3,100.0	\$2,600.0	\$525.0	\$1,000.0	\$1,850.0	•	•	•	•
FY2002	•	•	\$2,850.0	\$3,000.0	\$620.0	\$950.0	\$2,750.0	•	•	•	•
FY2003	•	•	\$2,415.0	•	\$650.0	\$950.0	\$2,600.0 x	\$685.0 y	•	•	
FY2004	•	•	\$3,100.0	•	\$650.0	\$950.0	\$7,050.0	\$2,700.0	•	•	\$3,514.0
FY2005	•	•	\$3,100.0	•	\$650.0	\$950.0	\$4,150.0	\$500.0	•	•	\$3,020.0
FY2006	•	•	\$3,100.0	•	\$650.0	\$950.0	\$4,160.0	\$500.0	•	• .	\$3,020.0
FY2007	•	•	\$3,100.0	•	\$650.0	\$950.0	\$4,160.0	\$500.0	\$1,350.0	\$500.0	\$3,160.0
FY2008	•	•	\$3,100.0	•	\$325.0	\$950.0	\$4,100.0	\$500.0	\$1,350.0	•	\$3,160.0
FY2009	•	•	\$3,100.0	•	\$325.0	\$950.0	\$4,100.0	\$500.0	\$1,350.0	•	\$3,160.0
FY2010	•	•	\$2,165.0	•	\$206.4	\$950.0	\$1,935.1	\$500.0	\$574.7	•	\$3,160.0
FY2011	•	•	\$1,393.0	•	\$331.1	\$1,147.3	\$1,804.0	\$500.0	•	•	\$3,160.0

Table 1.0, Appropriation History (\$ in thousands), continued 2011 ISAC Data Book

Fiscal <u>Year</u>	Nurse Educator Ln Repay Prog	Veterans' Home Nurse <u>Ln Repay Prog</u>	Lender <u>Reimb.</u>	State <u>Admin</u>	Federal <u>Admin</u>	Total State <u>Programs</u>	Total <u>SLF*</u>	Total Other Federal <u>Programs</u>	Grand <u>Total</u>
FY1980	•	•	\$12,182.0	\$1,732.3	\$2,156.0	\$81,205.6	\$14,338.0	\$4,150.0	\$99,693.6
FY1981	•	•	\$15,000.0	\$1,815.9	\$3,513.2	\$86,378.0	\$18,513.2	\$4,170.0	\$109,061.2
FY1982	•	•	\$18,000.0	\$1,979.2	\$5,790.5	\$89,930.5	\$23,790.5	\$4,250.0	\$117,971.0
FY1983	•	•	\$30,000.0	\$2,188.6	\$6,386.3	\$93,806.9	\$36,386.3	\$4,033.0	\$134,226.2
FY1984	•	•	\$30,000.0	\$2,048.6	\$5,934.1	\$104,093.6	\$35,934.1	\$3,240.9	\$143,268.6
FY1985	•	•	\$69,000.0	\$2,130.0	\$7,224.3	\$109,323.6	\$76,224.3	\$4,105.1	\$189,653.0
FY1986	•	•	\$104,800.0	\$2,228.5	\$8,251.9	\$124,630.5	\$113,051.9	\$4,200.0	\$241,882.4
FY1987	•	•	\$140,097.6	\$2,433.1	\$10,866.8	\$140,922.9	\$150,964.4	\$4,409.4	\$296,296.7
FY1988	•	•	\$164,564.0	\$2,345.3	\$13,503.8	\$141,916.7	\$178,067.8	\$5,100.0	\$325,084.5
FY1989	•	•	\$136,770.0	\$2,423.4	\$16,144.7	\$161,024.0	\$152,914.7	\$4,900.0	\$318,838.7
FY1990	•	•	\$92,000.0	\$2,958.9 a	\$17,937.9	\$199,836.3 a	\$109,937.9	\$4,900.0	\$314,674.2
FY1991	•	•	\$100,000.0	\$3,042.6	\$19,183.9	\$203,582.1	\$119,183.9	\$4,500.0	\$327,266.0
FY1992	•	•	\$128,000.0	\$3,157.0	\$21,359.1	\$205,533.9	\$149,359.1	\$4,930.7	\$359,823.7
FY1993	•	•	\$150,000.0	\$3,516.0	\$24,716.8	\$221,952.9	\$174,716.8	\$5,500.0	\$402,169.7
FY1994	•	•	\$144,100.0	\$3,455.3	\$28,026.2	\$233,272.2	\$172,126.2	\$5,500.0	\$410,898.4
FY1995	•	•	\$167,265.3	\$4,910.6 k	\$29,410.5 1	\$273,778.7	\$196,675.8	\$6,551.5 m	\$477,006.0
FY1996	•	•	\$174,200.0	\$5,115.5 n	\$29,000.0 o	\$288,885.9	\$202,200.0	\$7,550.0	\$498,635.9
FY1997	•	•	\$174,200.0	\$5,403.4 p	\$26,000.0	\$311,276.7 q	\$200,200.0	\$5,350.0	\$516,826.7
FY1998	•	•	\$174,200.0	\$5,610.2 r	\$26,688.1	\$330,778.6 q	\$200,888.1	\$4,570.0	\$536,236.7
FY1999	•	•	\$188,000.0	\$5,939.7 r	\$27,489.8	\$358,392.5 q	\$215,489.8	\$3,248.0	\$577,130.3
FY2000	•	•	\$160,000.0	\$6,111.9 s	\$29,946.5	\$390,333.5 u	\$275,446.5	\$3,250.0	\$669,030.0
FY2001	•	•	\$160,000.0	\$6,360.7 t	\$30,892.2	\$413,366.5 u	\$381,892.2	\$3,800.0	\$799,058.7
FY2002	•	•	\$150,000.0	\$7,018.1 v	\$32,125.8	\$428,586.4 w	\$295,325.8	\$4,900.0	\$728,812.2
FY2003	•	•	\$160,000.0	\$6,163.3 z	\$33,656.6	\$386,351.1 aa	\$287,156.6	\$5,500.0	\$679,007.7
FY2004	•	•	\$190,000.0	\$5,960.8 ab	\$34,619.8	\$399,369.9	\$302,332.3	\$5,900.0	\$707,602.2
FY2005	•	•	\$190,000.0	\$4,460.0 ac	\$36,146.6	\$393,604.8	\$255,146.6	\$2,200.0	\$650,951.4
FY2006	•	•	\$190,000.0	\$300.0 ad	\$41,638.3	\$390,869.8	\$257,972.7	\$5,900.0	\$654,742.5
FY2007	•	•	\$190,000.0	\$300.0	\$43,203.8	\$401,539.8	\$335,778.2 ae	\$5,900.0	\$743,218.0
FY2008	\$1,000.0	\$1,220.0	\$260,000.0	\$380.0 af	\$44,063.2	\$429,774.8	\$345,397.6	\$5,900.0	\$781,072.4
FY2009	\$1,000.0	\$1,220.0	\$260,000.0	\$380.0 af	\$42,639.5	\$431,204.8	\$337,473.9	\$12,600.0 ag	\$781,278.7
FY2010	\$300.0	\$43.3	\$290,000.0	\$380.0 af	\$44,477.5	\$428,031.1	\$362,977.5	\$12,400.0 ag	\$803,408.6
FY2011	\$300.0	\$50.0	\$290,000.0	\$380.0 af	\$44,477.5	\$441,324.1	\$362,977.5	\$17,400.0 ag	\$821,701.6

Table 1.0 Appropriation History, Footnote Reference 2011 ISAC Data Book

- * Includes Student Loan Operating Fund, Federal Student Loan Fund, and Federal Reserve Recall Fund.
- a Since FY90, General Funds revenue are partially derived from the Educational Assistance Fund.
- b Dollars were appropriated for development and administration of these programs, not for direct aid to students.
- c Includes \$2.6 million for MAP Summer School and Less-Than-Half-Time Demonstration Projects.
- d Includes \$4.0 million for MAP Summer School and Less-Than-Half-Time Demonstration Projects.
- e All FY2002 figures reflect the original appropriation minus reserve requirements.
- f Includes \$2.0 million for the MAP Less-Than-Half Time Demonstration Project.
- g Includes the transfer of \$800,000 from MRS to IIA.
- h All FY2003 figures reflect the original appropriation minus reserve requirements.
- i Although ISAC had \$3.7 million in spending authority in FY2004, no funding was received for (S)LEAP.
- J Includes \$51.5 (federal funds) for Christa McAuliffe Fellowships.
- k Includes \$178.0 for collection activities related to Teacher Ed Programs.
- 1 Includes \$1,303.6 for activities related to the State Postsecondary Review Entity (SPRE).
- m Includes federal SSIG, Robert Byrd, Paul Douglas, and SPRE.
- n Includes \$125.0 for collection activities related to Teacher Ed Programs.
- o Includes \$1,000.0 for activities related to SPRE.
- p Includes \$150.0 for the State Account Receivable Fund and \$250.0 for Higher-EdNet.
- q Includes \$2,000.0 for the MAP Reserve Fund.
- r Includes \$150.0 for the State Account Receivable Fund and \$235.0 for Higher-EdNet.
- s Includes \$100.0 for the State Account Receivable Fund and \$235.0 for Higher-EdNet.
- t Includes \$60.0 for the State Account Receivable Fund and \$235.0 for Higher-EdNet.
- u Includes \$4,500.0 for the MAP Reserve Fund.
- v Includes \$300.0 for the State Account Receivable Fund and \$65.0 for Higher-EdNet.
- w Includes \$6,500.0 for the MAP Reserve Fund.
- x Prior to FY2003, this Program was named the DeBolt Teacher Shortage Scholarship.
- y Includes \$135,000 which could be used for administrative costs.
- z Includes \$300.0 for the State Account Receivable Fund and \$65.0 for Higher-EdNet.
- aa Includes \$6,670.0 for the MAP Reserve Fund.
- ab Includes \$300.0 for the State Account Receivable Fund and \$10.0 for Higher-EdNet; but excludes \$115.0 GRF administrative reserve.
- ac Includes \$300.0 for the State Account Receivable Fund but excludes \$91,000 GRF.
- ad Includes \$300.0 for the State Account Receivable Fund.
- ae Includes SLOF funding appropriated for MAP-Plus and MAP.
- af Includes \$300.0 for the State Account Receivable Fund, \$60.0 for Future Teacher Corps Fund and \$20.0 for National Guard Grant Fund.
- ag Includes Byrd Scholarship, Paul Douglas Refund to ED, Federal S/LEAP and Federal College Access Challenge Grants.

Table 1.1 of the 2011 ISAC Data Book Summary of Program Expenditures and Recipients State Fiscal Year 2011

	7 7	Total #	Average
Student Assistance Programs	Expenditures	<u>Recipients</u>	<u>Award</u>
Monetary Award Program (MAP)	\$403,295,683	147,210	\$2,740
Silas Purnell Illinois Incentive for Access Grant Program (IIA)			
Student-to-Student Grant Program (STS)	\$1,147,260	3,103	\$370
Higher Ed License Plate Program (HELP)*	\$70,000	280	\$250
Illinois Veteran Grant Program (IVG)**		9,869	
Illinois National Guard Grant Program	\$2,691,922	1,294	\$2,080
Minority Teachers of Illinois Scholarship (MTI)	\$1,160,922	241	\$4,817
Illinois Future Teachers Corp Program (IFTC)	\$183,420	26	\$7,055
Illinois Future Teachers Corp Program - Golden Apple (IFTC-GA)	\$1,138,649	163	\$6,986
Illinois Scholars (Golden Apple)#	\$3,361,100	204	
Illinois Teachers Loan Repayment Program	\$495,221	107	
Bonus Incentive Grant Program (BIG)	\$331,100	373	\$888
Grants for Dependents of Police/Fire/Correctional Officers	\$803,741	87	\$9,238
Optometric Education Scholarship Program	\$50,000	10	\$5,000
Nurse Educator Loan Repayment Program (NELR)	\$297,236	61	\$4,873
Veterans' Home Nurse Loan Repayment Program	\$14,319	4	\$3,580
John R. Justice Student Loan Repayment Program	\$363,859	130	\$2,799
Robert C. Byrd Honors Scholarship Program	\$1,942,500	1,316	\$1,476
IL Special Education Teacher Tuition Waiver Program		238	
Illinois State Scholars		17,476	
Total	\$417,346,932	182,192	

Students may participate in more than one program, therefore, recipient totals across programs do not represent total unduplicated students served.

^{*} Recipients are estimated assuming a \$250 average award.

^{**} In FY2011, the IVG Program was not funded. Recipients reflect student beneficiaries of tuition waivers at institutions.

[#] Spending includes administrative and indirect costs which preclude calculation of an average award.

The Monetary Award Program (MAP) provides grants to students to help pay for tuition and mandatory fees at Illinois colleges and universities and certain degree-granting institutions. Illinois is one of the largest providers of state need-based aid in the nation. To receive a MAP grant, a student must demonstrate financial need, be enrolled as an undergraduate for at least three credit hours per term at an ISAC-approved Illinois institution, be a resident of Illinois, and meet several other requirements. The maximum award level is dependent on legislative action and available funding in any given year. The award amount could increase or decrease throughout the academic year. Students apply using the *Free Application for Federal Student Aid* (FAFSA). In FY2011, 351,188 students were eligible to receive MAP grants. Of those eligible, 147,210 students actually enrolled and claimed their awards.

The tables in this section provide a variety of data including historical volume and payout summaries. The Monetary Award Program Payout Summary (Table 2.0a), the 2011 Monetary Award Program formula (Table 2.0b), a history of the Monetary Award Program maximum award (Table 2.0c); and the Monetary Award Program Suspension History (Table 2.0d). Statistical information by sector can be found in Tables 2.1-2.3A, statistical information by institution in Tables 2.3b-2.3e, by student income and dependency status in Tables 2.4a-2.4e, by applicant and announced eligible characteristics in Table 2.5a-2.6c, and by paid applicant characteristics in Tables 2.6d-2.6f.

PART TWO -- MONETARY AWARD PROGRAM

Table 2.0a of the 2011 ISAC Data Book Monetary Award Program Historical Awards and Payout Summary FY1997-FY2011

	FY1997	FY1998	FY1999	FY2000	FY2001
Total Applications	400,056	408,413	412,213	418,757	426,485
# Eligible Awards	190,609	193,480	194,985	194,036	197,889
% Eligible	47.6%	47.4%	47.3%	46.3%	46.4%
Eligible \$	\$484,878,006	\$523,997,445	\$556,283,929	\$582,968,469	\$624,602,410
# Paid Awards	127,607	127,039	136,456	136,697	139,421
% Paid	66.9%	65.7%	70.0%	70.4%	70.5%
Mean Award	\$2,063	\$2,201	\$2,247	\$2,383	\$2,500
Statutory Maximum Award	\$4,000	\$4,120	\$4,320	\$4,530	\$4,740
Payout	\$263,261,997	\$279,672,414	\$306,668,399	\$325,760,232	\$348,572,038
% of Appropriation Expended	99.3%	98.8%	98.9%	96.7%	98.2%
	FY2002	FY2003 *	FY2004 **	FY2005	FY2006
Total Applications	456,252	490,846	537,907	553,962	550,021
# Eligible Awards	210,299	214,179	236,631	241,024	236,168
% Eligible	46.1%	43.6%	44.0%	43.4%	42.9%
Eligible \$	\$682,282,198	\$636,895,172	\$644,644,641	\$647,972,798	\$693,738,054
# Paid Awards	140,744	132,025	140,898	150,311	146,853
% Paid	66.9%	61.6%	59.5%	62.5%	62.2%
Mean Award	\$2,646	\$2,539	\$2,355	\$2,198	\$2,365
Statutory Maximum Award	\$4,968	\$4,968	\$4,968	\$4,968	\$4,968
Payout	\$372,360,503	\$335,155,967	\$331,807,485	\$330,328,687	\$347,380,390
% of Appropriation Expended	99.5%	99.6%	98.0% #	97.5%	99.1%
	FY2007	FY2008	FY2009	FY2010 ***	FY2011
Total Applications	576,557	597,441	661,621	773,930	822,521
# Eligible Awards	236,306	239,455	259,333	314,198	351,188
% Eligible	41.0%	40.1%	39.2%	40.6%	42.7%
Eligible \$	\$760,621,252	\$767,040,027	\$811,519,288	\$950,311,045	\$1,033,321,391
# Paid Awards	146,635	145,543	144,230	141,380	147,210
% Paid	62.1%	60.8%	55.6%	45.0%	41.9%
Mean Award	\$2,613	\$2,637	\$2,662	\$2,762	\$2,740
Statutory Maximum Award	\$4,968	\$4,968	\$4,968	\$4,968	\$4,968
Payout	\$383,192,534	\$383,817,067	\$383,892,090	\$390,465,309	\$403,295,687
% of Appropriation Expended	99.6%	99.7%	99.6%	97.0%	99.1%

^{*}Due to budget cuts in 2003, eligibility was eliminated for 8,486 students who had received the equivalent of four years of full-time MAP.

^{**} In FY2004, eligibility was restored to students who had received up to the equivalent of 4.5 years of full-time MAP. These students were eligible for 50 percent of their annual award. There were 1,980 students who received 4.5 years of full-time MAP that were still ineligible.

^{# (}S)LEAP appropriation excluded in calculation. Spending authority approved but no funds were actually received.

^{***} In FY2010, ISAC received a lump sum for all programs. Funding was then cut in half. It was restored late in 2009 near the end of the first semester.

Table 2.0b of the 2011 ISAC Data Book FY2011 Monetary Award Program Formula

FY2011 MAP Formula

Budget

- 1. Use 2003-2004 reported tuition and fees at all institutions, assessed at 100 percent at all institutions.
- 2. Use one living allowance for all applicants, set to \$4,875.

Resources

- 1. Use 80 percent of Pell Grant eligibility as determined by the 2003-2004 Pell Grant Payment Schedule, which contains a \$4,050 maximum.
- 2. Calculate the ISAC adjusted EFC by inflating the Federal EFC.

Adjusted Dependent Students' Parent Contribution: Adjustment Factor = [PC/11,000 + 1.10] rounded to 2 decimal places Adjusted $PC = PC \times Adjustment$ Factor

 $\label{eq:Adjusted Independent Student Contribution:} Adjustment Factor = [EFC/11,000 + 1.10] rounded to 2 decimal places \\ Adjusted EFC = EFC x Adjustment Factor$

3. Use a minimum self-help expectation of \$1,800 for all students.

Award Amounts

- 1. Set the maximum award equal to the lesser of \$4,968 or the tuition and mandatory fees specified in the budget. Set the minimum award to \$300, and round maximum eligibility in \$150 increments to calculate partial awards.
- 2. Provide no award for applicants who have an EFC equal to or greater than \$9,000.
- 3. Students who have used 75 or more MAP paid credit hours must be a junior or senior to be eligible for MAP. Students who have used 135 or more MAP paid credit hours are not eligible for MAP.

Note: In FY2011, second term awards were reduced 5% at semester schools and second and third term awards were reduced 3.7% at quarter schools to keep claims within the appropriation.

Table 2.0c of the 2011 ISAC Data Book Monetary Award Program - Maximum Award History Academic Year 1977-2012

Academic	Maximum	Effective
Year	Award	Maximum
1977-1978	\$1,550	
1978-1979	\$1,650	
1979-1980	\$1,800	
1980-1981	\$1,900	
1981-1982	\$1,950	
1982-1983	\$2,000	
1983-1984	\$2,200	
1984-1985	\$2,400	
1985-1986	\$2,850	
1986-1987	\$3,100	
1987-1988	\$3,100	
1988-1989	\$3,150	
1989-1990	\$3,500	
1990-1991	\$3,500	
1991-1992	\$3,500	
1992-1993	\$3,500	
1993-1994	\$3,500	
1994-1995	\$3,800	
1995-1996	\$3,900	
1996-1997	\$4,000	
1997-1998	\$4,120	
1998-1999	\$4,320	
1999-2000	\$4,530	
2000-2001	\$4,740	
2001-2002	\$4,968	
2002-2003 *	\$4,968	\$4,720
2003-2004 *	\$4,968	\$4,471
2004-2005 *	\$4,968	\$4,471
2005-2006 *	\$4,968	\$4,521
2006-2007	\$4,968	
2007-2008	\$4,968	
2008-2009	\$4,968	
2009-2010	\$4,968	
2010-2011 *	\$4,968	\$4,844
2011-2012 *	\$4,968	\$4,720

^{*} Reduction factor applied to awards

Table 2.0d of the 2011 ISAC Data Book MAP / IIA Suspension History, FY1978-FY2011

						# Eligible	# Eligible		
Award		Original	Suspension		Shutdown	Including	Left in	\$ Eligible in	
Year	Program	Suspension Date	Release to Date	Final Suspension Date	Date	Suspended	Suspension	Suspension	Notes
1977-78	MAP			10/1/77			n/a		
1978-79	MAP			2/15/79			0		
1979-80	MAP			9/13/79			0		Supplemental Approp
1980-81	MAP			8/28/80			10,000		\$100 spring term cut
1981-82	MAP			2/15/82			0		
1982-83	MAP			3/15/83			0		
1983-84	MAP			12/9/83			5,418		
1984-85	MAP			12/10/84			9,488		
1985-86	MAP			3/15/86			0		\$50 spring term cut
1986-87	MAP			1/31/87			6,708		\$25 spring term cut
1987-88	MAP			3/15/88			0		
1988-89	MAP			3/15/89			0		
1989-90	MAP			3/15/90			0		
1990-91	MAP			12/19/90		158,224	9,006		
				10/2/91 for Continuing					
1991-92	MAP			and 11/13/91 for 1st-time 6/21/92 for Continuing		171,510	19,406		
1992-93	MAP			and 3/3/93 for 1st-time		179,925	33,935		
1993-94	MAP			10/2/93 for Continuing		191,912	5,439		
1994-95	MAP	Continuing	None	10/2/97 for Continuing		187,665	0		
				10/2/95 for Continuing		,			
1995-96	MAP			and 4/4/96 for 1st-time		188,242	5,271		
1996-97	MAP			10/2/96 for Continuing and 12/15/96 for 1st-time		190,609	12,393		
1990-97	WAF			10/2/97 for Continuing		190,009	12,393		
1997-98	MAP			and 1/15/98 for 1st-time		193,480	10,449		
1998-99	MAP			10/2/98 for Continuing		194,985	5,194		
1999-00	MAP	None	None	None		194,036	0		
2000-01	MAP/IIA	None	None	None	EOY	,	0		
2000 01		Tione	Release through	Tione	LOI	177,307	0		
2001-02	MAP	10/27/01	12/7/01	12/8/01	EOY		16,544	\$23,218,285	
	IIA	None	None	None	EOY	42,771	0	\$0	

Note: for FY94 - FY99, continuing apps after 10/1 were not elig for MAP awards.

Table 2.0d of the 2011 ISAC Data Book, continued MAP / IIA Suspension History, FY1978-FY2011

A 3		Outstand	C		Cl 4 J	# Eligible	# Eligible	φ Ε ΙΙ - 1.1	
Award Year	Program	Original Suspension Date	Suspension Release to Date	Final Suspension Date	Shutdown Date	Including Suspended	Left in Suspension	\$ Eligible in Suspension	Notes
2002-03	MAP IIA	8/13/02 10/8/02	None None	8/13/02 10/8/02	3/4/03 3/4/03	214,179 42,197	44,144 8,158	\$32,553,829 \$2,039,500	
2003-04	MAP IIA	8/2/03 8/2/03	None None	8/2/03 8/2/03	EOY EOY	236,631 49,665	51,832 18,436	\$80,601,634 \$5,530,800	
2004-05	MAP IIA	8/16/04 8/16/04	Release through 10/15/04 Removed 2nd/3rd	10/16/04	EOY EOY	241,024 52,293	26,453 17,145	\$30,806,834 \$5,143,500	Removed 2nd/3rd term awards back to 7/15
2005-06	MAP IIA	9/1/05 8/2/05	None None		2/8/06 2/8/06	230,088 48,031	26,375 16,335	\$38,476,394 \$4,900,500	
2006-07	MAP IIA	8/25/06 8/19/06	None None		EOY EOY	236,168 52,340	34,798 15,285	\$56,123,768 \$7,642,500	
2007-08	MAP IIA	8/16/07 8/7/07	None None		EOY EOY	239,455 56,679	43,361 20,234	\$70,716,887 \$10,117,000	
2008-09	MAP IIA	7/26/08 7/12/08	8/1/08		EOY EOY	259,333 65,560	59,846 31,042	\$101,982,409 \$15,521,000	
2009-10	MAP IIA	5/15/09 4/18/09	6/4/09		EOY EOY	314,198 103,213	120,048 68,676		Funding only for first-term awards
2010-11	MAP IIA	4/19/10 Not Funded	None N/A	4/19/2010 N/A	EOY N/A	351,188 N/A	151,367 N/A	\$323,944,215 N/A	Second & third term awards reduced 5%
2011-12	MAP IIA	3/26/11 Not Funded	None N/A	4/8/2011 N/A	N/A	N/A	N/A	N/A	Eligible data not available until EOY

Table 2.1 of the 2011 ISAC Data Book Monetary Award Program Historical Award and Payout Summary by Sector FY1980-FY2011

PUBLIC

		PUBLIC	4		PUBLIC	C 2	ALL PUBLIC			
		Mean			Mean		•	Mean		
Fiscal Year	# Awards	Award	\$ Payout	# Awards	Award	\$ Payout	# Awards	Award	\$ Payout	
1979-1980	30,381	\$631	\$19,177,767	23,391	\$290	\$6,771,919	53,772	\$483	\$25,949,686	
1980-1981	32,880	\$661	\$21,721,264	24,014	\$285	\$6,836,543	56,894	\$502	\$28,557,807	
1981-1982	29,160	\$786	\$22,909,736	27,044	\$319	\$8,615,608	56,204	\$561	\$31,525,344	
1982-1983	34,038	\$754	\$25,646,857	37,447	\$345	\$12,911,839	71,485	\$539	\$38,558,696	
1983-1984	35,091	\$833	\$29,247,528	37,797	\$393	\$14,841,718	72,888	\$605	\$44,089,246	
1984-1985	35,288	\$892	\$31,476,383	34,260	\$358	\$12,248,879	69,548	\$629	\$43,725,262	
1985-1986	34,704	\$999	\$34,653,407	33,078	\$370	\$12,251,705	67,782	\$692	\$46,905,112	
1986-1987	35,004	\$1,140	\$39,911,262	30,394	\$420	\$12,755,175	65,398	\$805	\$52,666,437	
1987-1988	33,470	\$1,189	\$39,804,100	30,176	\$465	\$14,042,816	63,646	\$846	\$53,846,916	
1988-1989	34,497	\$1,275	\$43,982,387	31,058	\$438	\$13,599,931	65,555	\$878	\$57,582,318	
1989-1990	36,739	\$1,472	\$54,072,168	36,811	\$492	\$18,112,141	73,550	\$981	\$72,184,309	
1990-1991	39,612	\$1,647	\$65,237,393	38,139	\$516	\$19,664,380	77,751	\$1,092	\$84,901,773	
1991-1992	40,999	\$1,636	\$67,066,269	38,090	\$551	\$20,985,782	79,089	\$1,113	\$88,052,051	
1992-1993	40,356	\$1,957	\$78,955,892	34,903	\$642	\$22,397,927	75,259	\$1,347	\$101,353,819	
1993-1994	44,386	\$1,935	\$85,877,285	42,316	\$588	\$24,886,083	86,702	\$1,278	\$110,763,368	
1994-1995	44,222	\$2,132	\$94,261,070	44,450	\$684	\$30,409,433	88,672	\$1,406	\$124,670,503	
1995-1996	44,297	\$2,243	\$99,375,316	45,243	\$712	\$32,191,571	89,540	\$1,469	\$131,566,887	
1996-1997	44,749	\$2,353	\$105,282,746	43,179	\$745	\$32,159,287	87,928	\$1,563	\$137,442,033	
1997-1998	45,378	\$2,486	\$112,795,843	41,095	\$804	\$33,039,431	86,473	\$1,686	\$145,835,274	
1998-1999	45,849	\$2,539	\$116,419,385	44,819	\$805	\$36,066,502	90,668	\$1,682	\$152,485,887	
1999-2000	44,280	\$2,729	\$120,842,602	44,688	\$837	\$37,405,512	88,968	\$1,779	\$158,248,114	
2000-2001	44,663	\$2,868	\$128,109,930	46,195	\$888	\$41,016,991	90,858	\$1,861	\$169,126,921	
2001-2002	44,094	\$3,029	\$133,559,601	48,481	\$959	\$46,490,577	92,575	\$1,945	\$180,050,178	
2002-2003	39,844	\$3,011	\$119,989,270	48,421	\$935	\$45,254,556	88,265	\$1,872	\$165,243,826	
2003-2004	43,824	\$2,865	\$125,550,410	51,656	\$837	\$43,213,447	95,480	\$1,768	\$168,763,857	
2004-2005	43,989	\$2,726	\$119,914,304	58,745	\$813	\$47,756,914	102,734	\$1,632	\$167,671,218	
2005-2006	43,361	\$3,127	\$135,600,329	57,967	\$901	\$52,207,890	101,328	\$1,853	\$187,808,219	
2006-2007	43,377	\$3,439	\$149,176,374	57,211	\$989	\$56,595,122	100,588	\$2,046	\$205,771,496	
2007-2008	42,724	\$3,487	\$148,992,788	56,679	\$999	\$56,645,243	99,403	\$2,069	\$205,638,031	
2008-2009	42,372	\$3,542	\$150,100,858	56,326	\$1,008	\$56,789,887	98,698	\$2,096	\$206,890,745	
2009-2010	42,115	\$3,627	\$152,739,701	52,690	\$1,031	\$54,323,319	94,805	\$2,184	\$207,063,020	
2010-2011	44,743	\$3,576	\$159,981,571	53,411	\$1,006	\$53,740,110	98,154	\$2,177	\$213,721,681	

Table 2.1, Monetary Award Program Historical Award and Payout Summary, continued 2011 ISAC Data Book

PRIVATE

	PRIVATE 4			PRIVA	TE 2 & I	HOSPITAL	ALL PRIVATE & HOSPITAL			
		Mean			Mean			Mean		
Fiscal Year	# Awards	Award	\$ Payout	# Awards	Award	\$ Payout	# Awards	Award	\$ Payout	
1979-1980	31,831	\$1,437	\$48,614,000	5,618	\$1,499	\$8,420,000	37,449	\$1,523	\$57,034,000	
1980-1981	32,547	\$1,494	\$48,607,293	6,199	\$1,281	\$7,939,098	38,746	\$1,459	\$56,546,391	
1981-1982	32,173	\$1,546	\$49,730,315	6,460	\$1,372	\$8,862,486	38,633	\$1,517	\$58,592,801	
1982-1983	29,112	\$1,608	\$46,820,592	5,413	\$1,429	\$7,733,901	34,525	\$1,580	\$54,554,493	
1983-1984	29,649	\$1,766	\$52,345,476	5,357	\$1,532	\$8,207,447	35,006	\$1,730	\$60,552,923	
1984-1985	29,571	\$1,991	\$56,875,669	5,429	\$1,674	\$9,088,447	35,000	\$1,885	\$65,964,116	
1985-1986	29,428	\$2,190	\$64,443,433	5,411	\$1,841	\$9,959,078	34,839	\$2,136	\$74,402,511	
1986-1987	28,247	\$2,401	\$67,822,711	5,140	\$2,128	\$10,940,353	33,387	\$2,359	\$78,763,064	
1987-1988	28,907	\$2,414	\$69,782,260	5,358	\$2,147	\$11,503,208	34,265	\$2,372	\$81,285,469	
1988-1989	29,835	\$2,431	\$72,526,014	5,648	\$2,195	\$12,395,553	35,483	\$2,393	\$84,921,567	
1989-1990	31,441	\$2,730	\$85,825,373	5,099	\$2,389	\$12,180,745	36,540	\$2,682	\$98,006,118	
1990-1991	30,860	\$2,784	\$85,925,594	4,595	\$2,520	\$11,580,557	35,455	\$2,750	\$97,506,151	
1991-1992	30,936	\$2,691	\$83,234,025	4,730	\$2,522	\$11,927,035	35,666	\$2,668	\$95,161,060	
1992-1993	29,630	\$2,895	\$85,771,018	5,362	\$2,558	\$13,717,014	34,992	\$2,843	\$99,488,032	
1993-1994	30,600	\$2,671	\$81,743,042	5,839	\$2,408	\$14,058,069	36,439	\$2,629	\$95,801,111	
1994-1995	32,418	\$3,024	\$98,031,188	6,129	\$2,771	\$16,982,525	38,547	\$2,984	\$115,013,713	
1995-1996	36,619	\$3,081	\$112,812,726	3,824	\$2,848	\$10,892,330	40,443	\$3,059	\$123,705,056	
1996-1997	36,336	\$3,186	\$115,761,882	3,343	\$3,009	\$10,058,082	39,679	\$3,171	\$125,819,964	
1997-1998	36,765	\$3,325	\$122,246,472	2,924	\$3,130	\$9,153,411	39,689	\$3,311	\$131,399,883	
1998-1999	38,858	\$3,472	\$134,898,617	3,035	\$3,174	\$9,634,430	41,893	\$3,450	\$144,533,047	
1999-2000	38,474	\$3,646	\$140,294,139	2,965	\$3,344	\$9,915,423	41,439	\$3,625	\$150,209,562	
2000-2001	38,500	\$3,860	\$148,612,040	3,036	\$3,500	\$10,626,012	41,536	\$3,834	\$159,238,052	
2001-2002	38,474	\$4,130	\$158,912,684	2,973	\$3,828	\$11,380,587	41,447	\$4,109	\$170,293,271	
2002-2003	35,343	\$4,006	\$141,598,695	2,651	\$3,744	\$9,926,167	37,994	\$3,988	\$151,524,862	
2003-2004	36,773	\$3,714	\$136,559,508	2,873	\$3,451	\$9,914,766	39,646	\$3,695	\$146,474,274	
2004-2005	38,099	\$3,562	\$135,727,394	2,802	\$3,125	\$8,757,474	40,901	\$3,533	\$144,484,868	
2005-2006	36,391	\$3,665	\$133,367,576	2,721	\$3,160	\$8,597,888	39,112	\$3,630	\$141,965,464	
2006-2007	36,563	\$4,048	\$147,999,179	2,554	\$3,450	\$8,811,025	39,117	\$4,009	\$156,810,204	
2007-2008	36,569	\$4,056	\$148,308,122	2,770	\$3,464	\$9,595,055	39,339	\$4,014	\$157,903,177	
2008-2009	35,434	\$4,098	\$145,221,115	2,636	\$3,537	\$9,323,552	38,070	\$4,059	\$154,544,667	
2009-2010	35,844	\$4,173	\$149,589,723	2,683	\$3,596	\$9,647,372	38,527	\$4,133	\$159,237,095	
2010-2011 *	38,954	\$4,058	\$158,081,497	1,738	\$3,618	\$6,287,773	40,692	\$4,039	\$164,369,270	

^{*} In FY2011, Lexington College and St. Augustine College changed from Private 2-Year to Private 4-Year Institutions which is reflected in Private totals.

Table 2.1, Monetary Award Program Historical Award and Payout Summary, continued 2011 ISAC Data Book

PROPRIETARY

Fiscal Year	# Awards	Mean <u>Award</u>	\$ Payout
1997-1998 *	877	\$2,779	\$2,437,256
1998-1999 **	3,895	\$2,477	\$9,649,465
1999-2000 **	* 6,290	\$2,751	\$17,302,556
2000-2001	7,027	\$2,876	\$20,207,065
2001-2002	6,722	\$3,275	\$22,017,053
2002-2003	5,766	\$3,189	\$18,387,279
2003-2004	5,772	\$2,871	\$16,569,355
2004-2005	6,676	\$2,722	\$18,172,601
2005-2006	6,413	\$2,745	\$17,606,707
2006-2007	6,930	\$2,974	\$20,610,835
2007-2008	6,801	\$2,981	\$20,275,860
2008-2009	7,462	\$3,009	\$22,456,677
2009-2010	8,048	\$3,003	\$24,165,194
2010-2011	8,364	\$3,013	\$25,204,735

 $Beginning\ in\ FY1998, Proprietary\ Institutions\ meeting\ eligiblity\ criteria\ became\ eligible\ for\ participation\ in\ the\ Monetary\ Award\ Program.$

^{*} First-time freshmen only. ** All freshmen and sophomores. *** All class levels from this year forward.

Table 2.1, Monetary Award Program Historical Award and Payout Summary, continued 2011 ISAC Data Book

ALL INSTITUTIONS

	ALL PRIV	ALL PRIVATE & HOSPITAL ALL PUBLIC		LIC	PRO	PRIETA	RY	TOTAL				
		Mean			Mean			Mean			Mean	
Fiscal Year	# Awards	Award		# Awards	Award	\$ Payout	# Awards		\$ Payout	# Awards	Award	\$ Payout
1979-1980	37,449	\$1,523	\$57,034,000	53,772	\$483	\$25,949,686				91,221	\$910	\$82,983,686
1980-1981	38,746	\$1,459	\$56,546,391	56,894	\$502	\$28,557,807				95,640	\$890	\$85,104,198
1981-1982	38,633	\$1,517	\$58,592,801	56,204	\$561	\$31,525,344				94,837	\$950	\$90,118,145
1982-1983	34,525	\$1,580	\$54,554,493	71,485	\$539	\$38,558,696				106,010	\$878	\$93,113,189
1983-1984	35,006	\$1,730	\$60,552,923	72,888	\$605	\$44,089,246				107,894	\$970	\$104,642,169
1984-1985	35,000	\$1,885	\$65,964,116	69,548	\$629	\$43,725,262				104,548	\$1,049	\$109,689,378
1985-1986	34,839	\$2,136	\$74,402,511	67,782	\$692	\$46,905,112				102,621	\$1,182	\$121,307,623
1986-1987	33,387	\$2,359	\$78,763,064	65,398	\$805	\$52,666,437				98,785	\$1,330	\$131,429,501
1987-1988	34,265	\$2,372	\$81,285,469	63,646	\$846	\$53,846,916				97,911	\$1,380	\$135,132,385
1988-1989	35,483	\$2,393	\$84,921,567	65,555	\$878	\$57,582,318				101,038	\$1,410	\$142,503,885
1989-1990	36,540	\$2,682	\$98,006,118	73,550	\$981	\$72,184,309				110,090	\$1,546	\$170,190,429
1990-1991	35,455	\$2,750	\$97,506,151	77,751	\$1,092	\$84,901,773				113,206	\$1,611	\$182,407,924
1991-1992	35,666	\$2,668	\$95,161,060	79,089	\$1,113	\$88,052,051				114,755	\$1,597	\$183,213,111
1992-1993	34,992	\$2,843	\$99,488,032	75,259	\$1,347	\$101,353,819				110,251	\$1,822	\$200,841,851
1993-1994	36,439	\$2,629	\$95,801,111	86,702	\$1,278	\$110,763,368				123,141	\$1,677	\$206,564,479
1994-1995	38,547	\$2,984	\$115,013,713	88,672	\$1,406	\$124,670,503				127,219	\$1,884	\$239,684,216
1995-1996	40,443	\$3,059	\$123,705,056	89,540	\$1,469	\$131,566,887				129,983	\$1,964	\$255,271,943
1996-1997	39,679	\$3,171	\$125,819,964	87,928	\$1,563	\$137,442,033				127,607	\$2,063	\$263,261,997
1997-1998	39,689	. ,	\$131,399,883	86,473	\$1,686	\$145,835,274	877	\$2,779	\$2,437,256	127,039	\$2,201	\$279,672,414
1998-1999	41,893	\$3,450	\$144,533,047	90,668	\$1,682	\$152,485,887	3,895	\$2,477	\$9,649,465	136,456	\$2,247	\$306,668,399
1999-2000	41,439	\$3,625	\$150,209,562	88,968	\$1,779	\$158,248,114	6,290	\$2,751	\$17,302,556	136,697	\$2,383	\$325,760,232
2000-2001	41,536	\$3,834	\$159,238,052	90,858	\$1,861	\$169,126,921	7,027	\$2,876	\$20,207,065	139,421	\$2,500	\$348,572,038
2001-2002	41,447	. ,	\$170,293,271	92,575	. ,	\$180,050,178		\$3,275	\$22,017,053	140,744	\$2,646	\$372,360,502
2002-2003	37,994	\$3,988	\$151,524,862	88,265	\$1,872	\$165,243,826	5,766	\$3,189	\$18,387,279	132,025	\$2,539	\$335,155,967
2003-2004	39,646	\$3,695	\$146,474,274	95,480	\$1,768	\$168,763,857	5,772	\$2,871	\$16,569,355	140,898	\$2,355	\$331,807,485
2004-2005	40,901	\$3,533	\$144,484,868	102,734	\$1,632	\$167,671,218	6,676	\$2,722	\$18,172,601	150,311	\$2,198	\$330,328,687
2005-2006	39,112	\$3,630	\$141,965,464	101,328	\$1,853	\$187,808,219	6,413	\$2,745	\$17,606,707	146,853	\$2,365	\$347,380,390
2006-2007	39,117	\$4,009	\$156,810,204	100,588	\$2,046	\$205,771,496	6,930	\$2,974	\$20,610,835	146,635	\$2,613	\$383,192,535
2007-2008	39,339	\$4,014	\$157,903,177	99,403	\$2,069	\$205,638,031	6,801	\$2,981	\$20,275,860	145,543	\$2,637	\$383,817,068
2008-2009	38,070	\$4,059	\$154,544,667	98,698	\$2,096	\$206,890,745	7,462	\$3,009	\$22,456,677	144,230	. ,	\$383,892,090
2009-2010	38,527	\$4,133	\$159,237,095	94,805	\$2,184	\$207,063,020	8,048	\$3,003	\$24,165,194	141,380	\$2,762	\$390,465,310
2010-2011	40,692	\$4,039	\$164,369,270	98,154	\$2,177	\$213,721,681	8,364	\$3,013	\$25,204,735	147,210	\$2,740	\$403,295,687

Table 2.2 of the 2011 ISAC Data Book Weighted Mean Tuition and Fees and MAP College Budgets by Sector, FY1980-FY2011*

	WEIGHTED MEAN: TUITION AND FEES				S	WEIGHTED MEAN: TOTAL MAP COLLEGE BUDGETS							
Fiscal Year	Public Univ.	Public 2-Year	Private 4-Year	Private 2-Year l	Proprietary **	All Combined	Fiscal Year	Public Univ.	Public 2-Year	Private 4-Year	Private 2-Year	Proprietary	All Combined
1980	\$828	\$469	\$3,521	\$2,221		\$1,399	1980	\$3,449	\$3,019	\$6,459	\$5,070	· ·	\$4,078
1981	\$901	\$501	\$3,899	\$2,571	•	\$1,508	1981	\$3,878	\$3,426	\$7,201	\$5,535	•	\$4,550
1982	\$1,017	\$539	\$4,242	\$2,853	•	\$1,518	1982	\$4,314	\$3,664	\$7,764	\$6,088	•	\$4,789
1983	\$1,156	\$612	\$4,845	\$3,240	•	\$1,699	1983	\$4,253	\$3,537	\$8,170	\$6,291	•	\$4,768
1984	\$1,334	\$704	\$5,266	\$3,508		\$1,926	1984	\$4,576	\$3,629	\$9,018	\$6,696	•	\$5,140
1985	\$1,503	\$732	\$5,748	\$3,836		\$2,122	1985	\$4,857	\$3,832	\$9,616	\$7,135	•	\$5,479
1986	\$1,615	\$766	\$6,267	\$4,312	•	\$2,235	1986	\$5,084	\$3,966	\$10,283	\$7,732	•	\$5,698
1987	\$1,710	\$791	\$6,809	\$4,452	•	\$2,390	1987	\$5,277	\$4,091	\$10,952	\$8,022	•	\$5,958
1988 a	\$1,787	\$853	\$7,132	\$4,914	•	\$2,548	1988 a	\$5,417	\$4,153	\$11,408	\$8,557	•	\$6,172
1988 b	\$1,914	\$853	\$7,132	\$4,914	•	\$2,592	1988 b	\$5,544	\$4,153	\$11,408	\$8,557	•	\$6,216
1989 с	\$2,222	\$896	\$7,693	\$5,192	•	\$2,828	1989 с	\$5,929	\$4,246	\$12,093	\$8,946	•	\$6,525
1990	\$2,330	\$925	\$8,362	\$5,391	•	\$3,014	1990	\$6,185	\$4,475	\$13,005	\$9,060	•	\$6,897
1991	\$2,410	\$954	\$9,110	\$5,816	•	\$3,185	1991	\$6,414	\$4,704	\$14,017	\$9,674	•	\$7,259
1992	\$2,538	\$1,038	\$9,799	\$6,466	•	\$3,378	1992	\$6,654	\$4,788	\$14,963	\$10,355	•	\$7,535
1993	\$2,901	\$1,108	\$10,499	\$6,817	•	\$3,676	1993	\$7,140	\$4,858	\$15,852	\$10,711	•	\$7,912
1994	\$3,134	\$1,201	\$11,054	\$7,246	•	\$3,990	1994 d	\$7,184	\$5,251	\$15,104	\$11,296	•	\$8,040
1995	\$3,303	\$1,259	\$11,716	\$7,615	•	\$4,245	1995	\$7,353	\$5,309	\$15,766	\$11,665	•	\$8,295
1996	\$3,434	\$1,323	\$12,317	\$7,016	•	\$4,519	1996	\$7,584	\$5,473	\$16,467	\$11,166	•	\$8,669
1997	\$3,629	\$1,370	\$13,036	\$7,310	•	\$4,789	1997	\$7,879	\$5,620	\$17,261	\$11,560	•	\$9,033
1998 e	\$3,817	\$1,452	\$13,975	\$7,782	\$7,553	\$5,079	1998 e	\$8,317	\$5,952	\$18,475	\$12,282	\$12,053	\$9,579
1999	\$3,942	\$1,506	\$14,850	\$8,194	\$7,921	\$5,443	1999	\$8,442	\$6,006	\$19,350	\$12,694	\$12,421	\$9,943
2000	\$4,160	\$1,576	\$15,625	\$8,864	\$8,441	\$5,801	2000	\$8,660	\$6,076	\$20,125	\$13,364	\$12,941	\$10,301
2001	\$4,406	\$1,653	\$16,362	\$9,162	\$9,066	\$6,095	2001	\$9,106	\$6,353	\$21,062	\$13,862	\$13,766	\$10,795
2002	\$4,786	\$1,731	\$17,105	\$9,491	\$9,882	\$6,525	2002	\$9,661	\$6,606	\$21,980	\$14,366	\$14,757	\$11,400
2003	\$5,298	\$1,830	\$17,905	\$10,050	\$10,109	\$6,964	2003	\$10,173	\$6,705	\$22,780	\$14,925	\$14,984	\$11,839
2004	\$5,785	\$1,935	\$18,944	\$10,537	\$10,403	\$7,363	2004	\$10,660	\$6,810	\$23,819	\$15,412	\$15,278	\$12,238
2005	\$6,565	\$2,138	\$19,994	\$11,284	\$14,360	\$7,993	2005	\$11,440	\$7,013	\$24,869	\$16,159	\$19,235	\$12,868
2006	\$7,151	\$2,318	\$21,148	\$11,650	\$17,587	\$8,605	2006	\$12,026	\$7,193	\$26,023	\$16,525	\$22,462	\$13,480
2007	\$7,875	\$2,465	\$22,311	\$11,677	\$15,322	\$9,344	2007	\$12,750	\$7,340	\$27,185	\$16,552	\$20,197	\$14,219
2008	\$8,553	\$2,603	\$23,719	\$13,207	\$16,700	\$10,077	2008	\$13,428	\$7,478	\$28,594	\$18,082	\$21,575	\$14,952
2009	\$9,452	\$2,762	\$25,305	\$14,240	\$18,905	\$10,881	2009	\$14,327	\$7,637	\$30,180	\$19,115	\$23,780	\$15,756
2010	\$10,442	\$2,939	\$26,257	\$14,866	\$18,874	\$11,580	2010	\$15,317	\$7,814	\$31,132	\$19,741	\$23,749	\$16,455
2011	\$11,386	\$3,307	\$27,552	\$15,426	\$19,098	\$11,990	2011	\$16,261	\$8,182	\$33,427	\$20,301	\$23,973	\$16,895

^{*} Weights equal Fall Undergraduate FTE. MAP college budgets include: 1) tuition and fees; 2) room and board or an ISAC determined commuter allowance; and, 3) ISAC determined personal allowance.

Sources: ISAC College Budget Records, IBHE Enrollment Reports

^{**} Data reflects changing costs and new schools entering the program.

a As reported June 1, 1987 (Does not include mid-year increases).

b Includes mid-year increases.

c Includes public four-year mid-year tuition increases.

d Beginning in FY1994, MAP college budgets include tuition and fees and a standardized cost of living allowance. Beginning in FY2002, this changed to a uniform living allowance of \$4,875.

e Beginning in FY1998, Proprietary Institutions meeting eligibility criteria became eligible for participation in the Monetary Award Program.

Table 2.3a of the 2011 ISAC Data Book Monetary Award Program Summary of Awards and Payout by Sector FY2007-FY2011

	2006-2007		 2007-2008		2008-2009		200	9-2010	-	201	2010-2011*	
Sector	# Awards	\$ Payout	# Awards	\$ Payout		# Awards	\$ Payout	# Awards	\$ Payout		# Awards	\$ Payout
Public 4-Year	43,377	\$149,176,374	42,724	\$148,992,788		42,372	\$150,100,858	42,115	\$152,739,701		44,743	\$159,981,571
Private 4-Year	36,563	\$147,999,179	36,569	\$148,308,122		35,434	\$145,221,115	35,844	\$149,589,723		38,954	\$158,081,497
Public 2-Year	57,211	\$56,595,122	56,679	\$56,645,243		56,326	\$56,789,887	52,690	\$54,323,319		53,411	\$53,740,110
Private 2-Year	1,836	\$6,305,510	1,921	\$6,632,903		1,787	\$6,243,846	1,847	\$6,616,440		848	\$3,167,258
Hospital	718	\$2,505,515	849	\$2,962,152		849	\$3,079,706	836	\$3,030,932		890	\$3,120,515
Proprietary	6,930	\$20,610,835	6,801	\$20,275,860		7,462	\$22,456,677	8,048	\$24,165,194		8,364	\$25,204,735
All Sector Total	146,635	\$383,192,535	145,543	\$383,817,067		144,230	\$383,892,090	141,380	\$390,465,309		147,210	\$403,295,687

^{*} In FY2011, Lexington College and St. Augustine College changed from Private 2-Year to Private 4-Year Institutions which is reflected in the Private sector totals.

Table 2.3b of the 2011 ISAC Data Book Summary of MAP Awards and Payout by Institution FY2008-FY2011

Public 4-Year

MAP		200	7-2008	200	8-2009	200	9-2010	2010-2011	
Code	<u>Institution</u>	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout
010	Chicago State University	2,986	\$7,787,437	2,957	\$7,753,283	2,835	\$7,645,323	2,754	\$7,442,355
014	Eastern Illinois University	2,746	\$9,640,407	2,808	\$9,972,047	2,751	\$10,111,112	3,040	\$10,947,684
129	Governors State University	815	\$1,822,513	862	\$1,910,776	848	\$1,956,941	976	\$2,175,555
022	Illinois State University	3,615	\$12,837,679	3,508	\$12,553,143	3,840	\$14,034,842	4,239	\$15,315,482
079	Northeastern Illinois Univ.	3,231	\$8,437,111	3,218	\$8,672,315	3,182	\$8,719,389	3,263	\$8,959,610
045	Northern Illinois University	5,521	\$19,692,130	5,518	\$20,145,385	5,584	\$21,024,403	6,066	\$22,346,873
060	Southern Illinois-Carbondale	5,196	\$18,596,885	5,054	\$18,193,019	4,449	\$16,154,878	4,284	\$15,582,637
070	Southern Illinois-Edwardsville	2,658	\$7,688,552	2,623	\$7,607,060	2,442	\$7,236,194	2,676	\$7,850,476
064	University of Illinois-Chicago	5,998	\$23,867,920	6,114	\$24,463,623	6,594	\$26,747,956	7,207	\$28,528,619
127	University of Illinois-Springfield	792	\$2,460,828	818	\$2,659,939	812	\$2,746,463	839	\$2,801,086
065	University of Illinois-Urbana	6,206	\$25,593,126	6,014	\$25,795,793	6,038	\$26,085,611	6,516	\$27,502,999
066	Western Illinois University	2,960	\$10,568,200	2,878	\$10,374,476	2,740	\$10,276,589	2,883	\$10,528,194
	Total Public 4	42,724	<u>\$148,992,788</u>	42,372	<u>\$150,100,858</u>	<u>42,115</u>	<u>\$152,739,701</u>	44,743	<u>\$159,981,571</u>

Private 4-Year

MAP		2007	2007-2008		8-2009	2009-2010		2010-2011	
Code	<u>Institution</u>	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout
001	Augustana College	598	\$2,855,535	566	\$2,687,745	569	\$2,696,181	627	\$2,915,833
002	Aurora University	794	\$3,335,098	801	\$3,378,166	843	\$3,645,861	1,089	\$4,470,536
058	Benedictine University	1,030	\$3,813,316	1,098	\$4,044,377	1,165	\$4,363,062	1,422	\$5,315,957
005	Blackburn College	338	\$1,479,354	340	\$1,418,982	303	\$1,322,005	300	\$1,242,984
006	Bradley University	1,456	\$6,334,224	1,372	\$5,999,981	1,481	\$6,638,425	1,589	\$6,879,084
090	Columbia College	2,745	\$10,538,999	2,738	\$10,624,629	2,747	\$10,845,863	2,815	\$10,826,939

Table 2.3b, Summary of Awards and Payout by Institution, continued 2011 ISAC Data Book

Private 4-Year, continued

MAP		200	2007-2008		8-2009	2009-2010		2010	0-2011
Code	<u>Institution</u>	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout
011	Concordia University	340	\$1,428,812	330	\$1,430,853	372	\$1,622,363	526	\$2,211,886
013	DePaul University	4,441	\$18,104,217	4,549	\$18,647,942	4,795	\$20,155,591	5,213	\$21,286,954
055	Dominican University	606	\$2,646,334	681	\$3,049,443	768	\$3,432,319	885	\$3,945,121
150	East West University	889	\$3,129,727	870	\$2,878,580	636	\$2,294,757	598	\$2,082,108
016	Elmhurst College	899	\$3,928,565	877	\$3,823,081	903	\$4,032,591	997	\$4,293,079
017	Eureka College	316	\$1,401,262	322	\$1,402,054	336	\$1,422,552	354	\$1,523,533
019	Greenville College	495	\$2,067,200	511	\$2,162,360	463	\$1,953,934	505	\$2,107,413
098	Hebrew Theological College	63	\$258,218	63	\$289,597	60	\$265,592	67	\$295,473
020	Illinois College	409	\$1,801,487	344	\$1,557,183	337	\$1,536,918	358	\$1,526,732
021	Illinois Institute of Technology	532	\$2,344,665	516	\$2,341,921	554	\$2,510,993	566	\$2,505,745
023	Illinois Wesleyan University	448	\$2,159,352	439	\$2,127,301	443	\$2,126,794	477	\$2,260,478
083	Judson University	352	\$1,275,546	318	\$1,205,981	330	\$1,324,427	358	\$1,418,974
025	Kendall College*	376	\$1,232,282						
026	Knox College	265	\$1,250,611	248	\$1,185,860	258	\$1,224,612	241	\$1,108,621
027	Lake Forest College	284	\$1,333,600	266	\$1,251,437	309	\$1,500,995	347	\$1,616,486
029	Lewis University	1,211	\$5,018,436	1,189	\$5,003,473	1,154	\$4,966,299	1,340	\$5,577,309
160	Lexington College**							31	\$123,286
091	Lincoln Christian University	293	\$1,090,608	278	\$1,023,807	218	\$772,845	233	\$841,563
031	Loyola University Chicago	2,530	\$11,132,308	2,483	\$11,064,307	2,615	\$11,659,660	2,593	\$11,260,785
034	MacMurray College	364	\$1,531,288	298	\$1,265,739	245	\$996,441	235	\$978,906
033	McKendree University	706	\$2,948,495	656	\$2,771,993	719	\$3,023,339	717	\$2,979,191
036	Millikin University	938	\$4,023,117	918	\$4,001,558	948	\$4,091,911	939	\$4,077,409
038	Monmouth College	562	\$2,568,486	535	\$2,473,058	594	\$2,741,778	633	\$2,896,618
043	National Louis University	971	\$2,781,630	838	\$2,372,494	667	\$2,003,934	619	\$1,744,197
044	North Central College	639	\$2,851,976	656	\$2,986,151	746	\$3,434,559	748	\$3,330,564
046	North Park University	540	\$2,136,783	532	\$2,112,450	524	\$2,062,484	577	\$2,246,994
048	Northwestern University	517	\$2,388,612	486	\$2,274,953	527	\$2,430,601	542	\$2,445,298

^{*} In FY2009, Kendall College changed from a Private 4-year Institution to a for-profit Proprietary School. ** In FY2011, Lexington College changed from a Private 2-Year to a Private 4-Year Institution.

Table 2.3b, Summary of Awards and Payout by Institution, continued 2011 ISAC Data Book

Private 4-Year, continued

MAP		200	7-2008	200	8-2009	2009-2010		2010-2011	
Code	<u>Institution</u>	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout
049	Olivet Nazarene College	755	\$3,143,212	725	\$3,034,674	734	\$3,140,947	823	\$3,389,899
052	Quincy University	365	\$1,529,191	415	\$1,798,706	451	\$1,951,846	483	\$2,005,549
007	Robert Morris University Illinois	3,143	\$11,934,175	2,870	\$10,883,497	2,687	\$10,272,269	2,453	\$9,119,176
053	Rockford College	438	\$1,713,362	393	\$1,587,617	404	\$1,654,868	458	\$1,850,155
054	Roosevelt University	1,313	\$4,205,412	1,460	\$4,930,042	1,554	\$5,463,332	1,578	\$5,451,120
059	Shimer College	22	\$96,876	23	\$103,003	24	\$102,341	24	\$94,925
152	St. Augustine College**							997	\$3,230,731
069	St. Xavier University	1,604	\$6,239,423	1,483	\$5,940,164	1,405	\$5,734,567	1,529	\$6,109,978
144	Telshe Yeshiva	11	\$46,212	17	\$81,120	19	\$85,869	17	\$75,533
068	The School of the Art Institute	258	\$1,036,115	243	\$997,614	235	\$942,281	238	\$963,114
062	The University of Chicago	311	\$1,496,527	299	\$1,440,720	323	\$1,571,710	315	\$1,484,230
076	Trinity Christian College	365	\$1,376,875	385	\$1,450,586	400	\$1,479,477	444	\$1,641,180
080	Trinity International Univ- Reach	96	\$310,471	122	\$374,343	102	\$352,325	105	\$335,633
081	Trinity International University	204	\$911,354	180	\$771,417	168	\$746,438	157	\$689,970
057	University of St. Francis	553	\$2,293,987	545	\$2,303,573	525	\$2,220,796	586	\$2,456,690
857	University of St. Francis Professional Arts	22	\$45,673	22	\$64,384	28	\$62,201	35	\$86,136
102	Vandercook College of Music	58	\$264,991	53	\$244,517	63	\$289,320	61	\$267,941
067	Wheaton College	104	\$474,122	81	\$357,683	93	\$419,450	110	\$493,479
	Total Private 4	<u>36,569</u>	\$148,308,122	<u>35,434</u>	<u>\$145,221,116</u>	35,844	\$149,589,723	38,954	<u>\$158,081,497</u>

^{**} In FY2011, St. Augustine College changed from a Private 2-Year to a Private 4-Year Institution.

Public 2-Year

MAP	MAP		2007-2008		2008-2009		2009-2010		2010-2011	
Code	<u>Institution</u>	# Awards	\$ Payout							
103	Black Hawk College	1,563	\$1,580,865	1,278	\$1,242,729	986	\$1,007,332	1,019	\$1,020,191	
106	Carl Sandburg College	810	\$985,383	787	\$977,502	731	\$930,761	664	\$836,607	
032	College of DuPage	2,383	\$2,797,949	2,607	\$3,142,891	2,572	\$2,995,973	2,268	\$2,643,358	
074	College of Lake County	1,139	\$937,409	1,129	\$970,149	1,206	\$1,031,300	1,227	\$1,048,431	

Table 2.3b, Summary of Awards and Payout by Institution, continued 2011 ISAC Data Book

Public 2-Year, continued

MAP		2007	2007-2008		2008-2009		2009-2010		2010-2011	
Code	<u>Institution</u>	# Awards	\$ Payout							
012	Danville Area Community College	593	\$583,371	511	\$501,606	421	\$436,217	487	\$448,482	
015	Elgin Community College	1,103	\$1,079,825	1,167	\$1,196,770	1,283	\$1,282,053	1,592	\$1,548,560	
147	Frontier Community College	217	\$203,246	161	\$151,374	159	\$156,233	134	\$126,808	
114	Harold Washington College	2,725	\$2,814,178	2,429	\$2,517,151	2,631	\$2,745,228	2,777	\$2,887,209	
110	Harry S. Truman College	1,324	\$1,341,731	1,403	\$1,415,430	1,356	\$1,387,571	1,522	\$1,523,546	
124	Heartland Community College	868	\$731,643	826	\$718,301	768	\$687,106	771	\$681,080	
084	Highland Community College	506	\$501,493	519	\$513,043	463	\$497,968	471	\$501,791	
056	Illinois Central College	2,246	\$1,888,097	2,277	\$2,012,759	2,003	\$1,853,960	2,356	\$2,089,811	
028	Illinois Valley Community College	736	\$800,292	758	\$842,957	767	\$866,867	841	\$929,821	
122	John A. Logan College	1,267	\$1,316,489	1,246	\$1,274,723	961	\$1,003,347	959	\$977,649	
140	John Wood Community College	733	\$852,986	698	\$862,510	569	\$730,479	571	\$743,651	
024	Joliet Junior College	1,556	\$1,540,150	1,756	\$1,745,175	1,961	\$1,985,974	1,974	\$1,931,022	
037	Kankakee Community College	754	\$685,463	837	\$765,596	883	\$896,382	1,006	\$968,050	
800	Kaskaskia College	991	\$953,382	998	\$952,062	894	\$842,857	729	\$667,466	
116	Kennedy-King College	2,349	\$2,222,073	2,696	\$2,655,462	2,249	\$2,213,844	2,076	\$1,998,882	
009	Kishwaukee College	729	\$857,702	691	\$821,508	872	\$1,064,214	986	\$1,163,501	
105	Lake Land College	1,166	\$1,294,021	1,026	\$1,166,572	1,108	\$1,323,130	1,147	\$1,291,808	
131	Lewis & Clark Comm. College	1,369	\$1,517,593	1,309	\$1,441,610	922	\$996,763	740	\$851,705	
118	Lincoln Land Comm. College	1,441	\$1,213,473	1,367	\$1,132,279	1,150	\$1,009,885	1,110	\$930,485	
126	Lincoln Trail College	220	\$214,164	165	\$164,283	153	\$154,659	117	\$114,473	
112	Malcolm X College	1,753	\$1,720,621	1,888	\$1,822,825	1,789	\$1,695,503	1,791	\$1,695,704	
120	McHenry County College	524	\$511,356	561	\$569,575	691	\$693,544	617	\$659,753	
121	Moraine Valley Comm. College	1,831	\$1,887,142	1,997	\$2,052,780	1,875	\$1,973,440	1,720	\$1,655,785	
040	Morton College	1,089	\$940,639	837	\$668,011	780	\$680,018	844	\$701,509	
130	Oakton Community College	1,031	\$965,042	1,109	\$1,121,819	1,256	\$1,301,040	1,167	\$1,184,804	
115	Olive Harvey College	1,336	\$1,355,371	1,256	\$1,233,830	1,119	\$1,152,151	1,324	\$1,274,699	
108	Olney Central College	386	\$365,652	361	\$350,261	439	\$427,794	368	\$349,069	
107	Parkland College	1,740	\$1,959,066	1,703	\$1,932,661	1,616	\$1,907,407	1,369	\$1,556,821	
073	Prairie State College	1,448	\$1,657,752	1,238	\$1,421,772	1,073	\$1,226,301	1,096	\$1,250,614	

Table 2.3b, Summary of Awards and Payout by Institution, continued 2011 ISAC Data Book

Public 2-Year, continued

MAP	2007-2008		7-2008	2008-2009		2009-2010		2010-2011	
Code	<u>Institution</u>	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout
041	Rend Lake College	795	\$760,742	846	\$813,219	657	\$625,164	555	\$544,419
111	Richard J. Daley College	1,249	\$1,182,448	1,316	\$1,238,529	1,295	\$1,222,129	1,451	\$1,338,214
133	Richland Community College	616	\$512,291	537	\$467,660	513	\$462,539	582	\$525,210
085	Rock Valley College	1,397	\$1,144,478	1,527	\$1,264,364	1,537	\$1,286,009	1,703	\$1,396,463
088	Sauk Valley Community College	599	\$581,574	647	\$630,587	499	\$510,337	542	\$544,099
075	Shawnee Community College	668	\$534,916	512	\$409,062	397	\$311,459	376	\$289,454
063	South Suburban Coll of Cook Cty	2,669	\$2,760,337	2,465	\$2,598,087	1,876	\$1,947,990	1,706	\$1,872,150
078	Southeastern Illinois College	488	\$473,974	491	\$471,665	416	\$397,733	303	\$287,992
004	Southwestern Illinois College	2,064	\$1,631,327	1,844	\$1,490,117	1,287	\$1,072,647	1,154	\$950,094
077	Spoon River College	470	\$556,736	392	\$462,133	300	\$396,030	375	\$479,015
047	Triton College	1,513	\$1,629,378	1,581	\$1,691,148	1,687	\$1,897,019	1,845	\$1,979,941
082	Wabash Valley College	188	\$202,686	207	\$213,762	240	\$238,891	148	\$147,291
096	Waubonsee Community College	612	\$500,120	747	\$650,075	833	\$737,031	1,054	\$924,115
117	Wilbur Wright College	1,972	\$1,889,289	2,290	\$2,183,289	1,919	\$1,869,199	2,022	\$1,885,101
087	William Rainey Harper College	1,453	\$2,009,330	1,333	\$1,848,213	1,528	\$2,191,841	1,755	\$2,323,405
	Total Public 2	<u>56,679</u>	\$56,645,243	<u>56,326</u>	<u>\$56,789,887</u>	<u>52,690</u>	<u>\$54,323,319</u>	<u>53,411</u>	<u>\$53,740,110</u>

Private 2-Year

MAP		2007	2007-2008		2008-2009		2009-2010		2010-2011	
Code	Institution	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	
160	Lexington College**	32	\$115,121	28	\$113,810	31	\$126,520			
030	Lincoln College	635	\$2,377,340	624	\$2,422,737	656	\$2,593,678	679	\$2,602,947	
092	MacCormac College	95	\$309,906	71	\$201,866	51	\$143,355	74	\$184,618	
145	Morrison Institute of Tech	75	\$315,728	62	\$269,096	51	\$235,464	56	\$241,412	
061	Springfield College in Illinois	231	\$765,926	168	\$546,655	135	\$466,404	39	\$138,281	
152	St. Augustine College**	853	\$2,748,881	834	\$2,689,682	923	\$3,051,019			
	Total Private 2	<u>1,921</u>	<u>\$6,632,903</u>	<u>1,787</u>	<u>\$6,243,846</u>	<u>1,847</u>	<u>\$6,616,440</u>	<u>848</u>	<u>\$3,167,258</u>	

^{**} In FY2011, Lexington College and St. Augustine College changed from Private 2-Year to Private 4-Year Institutions.

Table 2.3b, Summary of Awards and Payout by Institution, continued 2011 ISAC Data Book

Hospital Schools

MAP		2007	-2008	2008	-2009	2009	-2010	2010	-2011
Code	<u>Institution</u>	# Awards	\$ Payout						
397	Advocate IL Masonic Schl of Rad Tech	2	\$3,000						
394	Advocate Trinity Hospital	12	\$54,648	7	\$32,292	3	\$12,786		
358	Blessing-Rieman Col of Nursing	21	\$73,042	32	\$112,967	30	\$105,758	56	\$206,271
172	Capital Area School of Prac Nursing	124	\$421,812	114	\$373,695	135	\$514,309	89	\$270,112
308	Graham Hosp/Schl of Nursing	42	\$139,406	38	\$133,772	44	\$154,483	49	\$166,094
334	Lakeview College of Nursing	74	\$268,937	87	\$304,314	84	\$310,628	76	\$279,752
312	Methodist Medical Center of Illinois	65	\$157,587	94	\$350,991	140	\$495,780	242	\$908,508
200	National University of Health Sciences	32	\$86,013	27	\$82,369	15	\$56,416	16	\$42,846
215	Rosalind Franklin Univ of Med & Science	3	\$14,242	2	\$9,108				
389	Rush University-Med Tech	24	\$111,528	16	\$79,488	11	\$51,336	17	\$70,388
335	Rush University-Nursing	30	\$139,104	11	\$49,680				
318	St. Anthony College of Nursing	76	\$270,620	82	\$302,141	75	\$275,485	74	\$264,532
321	St. Francis Medical Ctr Col of Nursing	158	\$580,772	161	\$575,041	146	\$530,512	136	\$465,389
390	St. Johns College Dept of Nursing	34	\$143,609	33	\$145,745	28	\$118,679	31	\$135,783
330	Trinity College of Nursing & Health Science	55	\$156,443	52	\$176,842	37	\$115,973	27	\$84,518
337	West Suburban College of Nursing	97	\$341,388	93	\$351,260	88	\$288,788	77	\$226,323
	Total Hospital Schools	<u>849</u>	\$2,962,152	<u>849</u>	\$3,079,706	<u>836</u>	\$3,030,932	<u>890</u>	\$3,120,515

Table 2.3b, Summary of Awards and Payout by Institution, continued 2011 ISAC Data Book

Proprietary Schools

MAP		2007-2008		2008-2009		2009-2010		2010-2011	
Code	<u>Institution</u>	# Awards	\$ Payout						
400	American Academy of Art					143	\$556,581	159	\$604,187
173	Argosy University	85	\$214,642	107	\$276,880	116	\$297,087	186	\$422,960
176	DeVry University - Chicago	3,267	\$10,101,913	3,554	\$10,991,295	3,764	\$11,406,046	3,863	\$11,852,331
180	Fox College			154	\$586,358	211	\$781,751	274	\$1,002,420
025	Kendall College *			476	\$1,603,970	530	\$1,660,945	619	\$1,882,364
174	Le Cordon Bleu College of Culinary Arts **	306	\$1,137,175	259	\$712,547	200	\$684,676	301	\$894,376
170	Midstate College	447	\$1,169,888	408	\$1,082,937	343	\$871,449	355	\$915,102
171	Northwestern College	1,421	\$3,383,183	1,197	\$2,857,243	1,298	\$3,087,759	1,042	\$2,575,279
146	The Illinois Institute of Art-Chicago	1,275	\$4,269,059	1,307	\$4,345,448	1,443	\$4,818,899	1,565	\$5,055,716
	Total Proprietary Schools	<u>6,801</u>	<u>\$20,275,860</u>	<u>7,462</u>	<u>\$22,456,677</u>	<u>8,048</u>	<u>\$24,165,194</u>	<u>8,364</u>	<u>\$25,204,735</u>
	GRAND TOTAL	145,543	\$383,817,067	144,230	\$383,892,090	141,380	\$390,465,309	147,210	\$403,295,687

^{*}In FY2009, Kendall College changed from a Private 4-year Institution to a for-profit Proprietary School. ** Formerly known as The Cooking and Hospitality Institute of Chicago

Table 2.3c of the 2011 ISAC Data Book MAP Recipients by Dependency Status and by Institution FY2011

All Schools	Dependent	Dependent Recipients		Independent Recipients		
	Number	% of Total	Number	% of Total		
	85,788	58%	61,422	42%	147,210	

Public 4-Year

MAP		Dependent 1	Recipients	Independent	Total # Paid	
Code	Institution	Number	% of Total	Number	% of Total	
010	Chicago State University	962	35%	1,792	65%	2,754
014	Eastern Illinois University	2,168	71%	872	29%	3,040
129	Governors State University	134	14%	842	86%	976
022	Illinois State University	3,375	80%	864	20%	4,239
079	Northeastern Illinois University	1,621	50%	1,642	50%	3,263
045	Northern Illinois University	4,304	71%	1,762	29%	6,066
060	Southern Illinois University-Carbondale	3,072	72%	1,212	28%	4,284
070	Southern Illinois University-Edwardsville	1,735	65%	941	35%	2,676
064	University of Illinois-Chicago	5,708	79%	1,499	21%	7,207
127	University of Illinois-Springfield	423	50%	416	50%	839
065	University of Illinois-Urbana	6,072	93%	444	7%	6,516
066	Western Illinois University	2,189	76%	694	24%	2,883
	Total Public 4-Year	31,763	71%	12,980	29%	44,743

Private 4-Year

MAP		Dependent	Recipients	Independen	t Recipients	Total # Paid
Code	<u>Institution</u>	Number	% of Total	Number	% of Total	
001	Augustana College	602	96%	25	4%	627
002	Aurora University	695	64%	394	36%	1,089
058	Benedictine University	861	61%	561	39%	1,422
005	Blackburn College	261	87%	39	13%	300
006	Bradley University	1,395	88%	194	12%	1,589
090	Columbia College	1,922	68%	893	32%	2,815
011	Concordia University	390	74%	136	26%	526

Table 2.3c, MAP Recipients by Dependency Status and by Institution, continued 2011 ISAC Data Book

MAP		Dependent	Recipients	Independen	t Recipients	Total # Paid
Code	<u>Institution</u>	Number	% of Total	Number	% of Total	
013	DePaul University	3,749	72%	1,464	28%	5,213
055	Dominican University	771	87%	114	13%	885
150	East West University	331	55%	267	45%	598
016	Elmhurst College	781	78%	216	22%	997
017	Eureka College	281	79%	73	21%	354
019	Greenville College	353	70%	152	30%	505
098	Hebrew Theological College	61	91%	6	9%	67
020	Illinois College	338	94%	20	6%	358
021	Illinois Institute of Technology	432	76%	134	24%	566
023	Illinois Wesleyan University	461	97%	16	3%	477
083	Judson University	187	52%	171	48%	358
026	Knox College	227	94%	14	6%	241
027	Lake Forest College	325	94%	22	6%	347
029	Lewis University	995	74%	345	26%	1,340
160	Lexington College	21	68%	10	32%	31
091	Lincoln Christian College			82	100%	82
031	Loyola University Chicago	2,168	84%	425	16%	2,593
034	MacMurray College	151	64%	84	36%	235
033	McKendree University	487	68%	230	32%	717
036	Millikin University	690	73%	249	27%	939
038	Monmouth College	600	95%	33	5%	633
043	National Louis University	48	8%	571	92%	619
044	North Central College	637	85%	111	15%	748
046	North Park University	396	69%	181	31%	577
048	Northwestern University	474	87%	68	13%	542
049	Olivet Nazarene University	605	74%	218	26%	823
052	Quincy University	339	70%	144	30%	483
007	Robert Morris University Illinois	1,203	49%	1,250	51%	2,453

Table 2.3c, MAP Recipients by Dependency Status and by Institution, continued 2011 ISAC Data Book

MAP	_	Dependent	Recipients	Independen	Total # Paid	
Code	<u>Institution</u>	Number	% of Total	Number	% of Total	
053	Rockford College	229	50%	229	50%	458
054	Roosevelt University	738	47%	840	53%	1,578
059	Shimer College	7	29%	17	71%	24
152	St. Augustine College	210	21%	787	79%	997
069	St. Xavier University	1,075	70%	454	30%	1,529
144	Telshe Yeshiva	17	100%			17
068	The School of The Art Institute	146	61%	92	39%	238
062	The University of Chicago	305	97%	10	3%	315
076	Trinity Christian College	246	55%	198	45%	444
080	Trinity International University Reach	1	1%	104	99%	105
081	Trinity International University	132	84%	25	16%	157
857	University of St. Francis Professional Arts	1	3%	34	97%	35
057	University of St. Francis	378	65%	208	35%	586
102	Vandercook College of Music	54	89%	7	11%	61
067	Wheaton College	101	92%	9	8%	110
	Total Private 4-Year	26,985	69%	11,969	31%	38,954

Public 2-Year

MAP		Dependent	Recipients	Independen	t Recipients	Total # Paid
Code	<u>Institution</u>	Number	% of Total	Number	% of Total	
103	Black Hawk College	402	39%	617	61%	1,019
106	Carl Sandburg College	262	39%	402	61%	664
032	College of DuPage	1,248	55%	1,020	45%	2,268
074	College of Lake County	554	45%	673	55%	1,227
012	Danville Area Community College	171	35%	316	65%	487
015	Elgin Community College	601	38%	991	62%	1,592
147	Frontier Community College	33	25%	101	75%	134
114	Harold Washington College	1,663	60%	1,114	40%	2,777
110	Harry S. Truman College	692	45%	830	55%	1,522
124	Heartland Community College	305	40%	466	60%	771

Table 2.3c, MAP Recipients by Dependency Status and by Institution, continued 2011 ISAC Data Book

Public 2-Year, continued

MAP		Dependent	Recipients	Independen	t Recipients	Total # Paid
Code	<u>Institution</u>	Number	% of Total	Number	% of Total	
084	Highland Community College	193	41%	278	59%	471
056	Illinois Central College	842	36%	1,514	64%	2,356
028	Illinois Valley Community College	344	41%	497	59%	841
122	John A. Logan College	437	46%	522	54%	959
140	John Wood Community College	222	39%	349	61%	571
024	Joliet Junior College	959	49%	1,015	51%	1,974
037	Kankakee Community College	337	33%	669	67%	1,006
008	Kaskaskia College	250	34%	479	66%	729
116	Kennedy-King College	622	30%	1,454	70%	2,076
009	Kishwaukee College	484	49%	502	51%	986
105	Lake Land College	523	46%	624	54%	1,147
131	Lewis & Clark Community College	272	37%	468	63%	740
118	Lincoln Land Community College	359	32%	751	68%	1,110
126	Lincoln Trail College	47	40%	70	60%	117
112	Malcolm X College	660	37%	1,131	63%	1,791
120	McHenry County College	321	52%	296	48%	617
121	Moraine Valley Community College	1,029	60%	691	40%	1,720
040	Morton College	494	59%	350	41%	844
130	Oakton Community College	713	61%	454	39%	1,167
115	Olive Harvey College	458	35%	866	65%	1,324
108	Olney Central College	145	39%	223	61%	368
107	Parkland College	632	46%	737	54%	1,369
073	Prairie State College	471	43%	625	57%	1,096
041	Rend Lake College	230	41%	325	59%	555
111	Richard J. Daley College	732	50%	719	50%	1,451
133	Richland Community College	166	29%	416	71%	582
085	Rock Valley College	589	35%	1,114	65%	1,703

Table 2.3c, MAP Recipients by Dependency Status and by Institution, continued 2011 ISAC Data Book

Public 2-Year, continued

MAP		Dependent	Recipients	Independen	Total # Paid	
Code	<u>Institution</u>	Number	% of Total	Number	% of Total	
088	Sauk Valley Community College	222	41%	320	59%	542
075	Shawnee Community College	168	45%	208	55%	376
063	South Suburban College of Cook County	539	32%	1,167	68%	1,706
078	Southeastern Illinois College	131	43%	172	57%	303
004	Southwestern Illinois College	392	34%	762	66%	1,154
077	Spoon River College	177	47%	198	53%	375
047	Triton College	902	49%	943	51%	1,845
082	Wabash Valley College	70	47%	78	53%	148
096	Waubonsee Community College	367	35%	687	65%	1,054
117	Wilbur Wright College	1,138	56%	884	44%	2,022
087	William Rainey Harper College	961	55%	794	45%	1,755
	Total Public 2-Year	23,529	44%	29,882	56%	53,411

Private 2-Year

MAP		Dependent	Recipients Independent Recipients		Total # Paid	
Code	<u>Institution</u>	Number	% of Total	Number	% of Total	
030	Lincoln College	469	69%	210	31%	679
092	MacCormac College	17	23%	57	77%	74
145	Morrison Institute of Tech	45	80%	11	20%	56
061	Springfield College in Illinois	15	38%	24	62%	39
	Total Private 2-Year	546	64%	302	36%	848

Table 2.3c, MAP Recipients by Dependency Status and by Institution, continued 2011 ISAC Data Book

Hospital Schools

MAP		Dependent	Recipients	Independen	t Recipients	Total # Paid
Code	<u>Institution</u>	Number	% of Total	Number	% of Total	
358	Blessing-Rieman College of Nursing	22	39%	34	61%	56
172	Capital Area School of Prac Nursing	10	11%	79	89%	89
308	Graham Hospital School of Nursing	6	12%	43	88%	49
334	Lakeview College of Nursing	33	43%	43	57%	76
312	Methodist Medical Center of Illinois	70	29%	172	71%	242
200	National University of Health Sciences	4	25%	12	75%	16
389	Rush University Medical Technology	7	41%	10	59%	17
318	St. Anthony College of Nursing	19	26%	55	74%	74
321	St. Francis Medical Center College Nursing	42	31%	94	69%	136
390	St. Johns College Dept of Nursing	11	35%	20	65%	31
330	Trinity College of Nursing & Health Science	6	22%	21	78%	27
337	West Suburban College of Nursing	22	29%	55	71%	77
	Total Hospital Schools	252	28%	638	72%	890

Proprietary Schools

MAP		Dependent	Recipients	Independent Recipients		Total # Paid
Code	<u>Institution</u>	Number	% of Total	Number	% of Total	
400	American Academy of Art	143	90%	16	10%	159
173	Argosy University	34	18%	152	82%	186
176	DeVry University-Chicago	938	24%	2,925	76%	3,863
180	Fox College	172	63%	102	37%	274
025	Kendall College	179	29%	440	71%	619
174	Le Cordon Bleu College of Culinary Arts*	83	28%	218	72%	301
170	Midstate College	13	4%	342	96%	355
171	Northwestern College	300	29%	742	71%	1,042
146	The Illinois Institute of Art	851	54%	714	46%	1,565
	Total Proprietary Schools	2,713	32%	5,651	68%	8,364

^{*}Formerly known as The Cooking and Hospitality Institute of Chicago

Table 2.3d of the 2011 ISAC Data Book Average Income by Dependency Status MAP Dependent and Independent Applicants by Institution FY2011

All Schools		DEPEN	DEPENDENTS		INDEPENDENTS	
		Eligibles Mean Income of Parents \$31,302	Non-Eligibles * Mean Income of Parents \$112,763	Eligibles <u>Mean Income</u> \$16,082	Non-Eligibles * Mean Income \$58,284	
Public 4-Year		DEPENDENTS		INDEPENDENTS		
MAP <u>Code</u>	<u>Institution</u>	Eligibles Mean Income of Parents	Non-Eligibles * Mean Income of Parents	Eligibles Mean Income	Non-Eligibles * Mean Income	
010 014 129	Chicago State University Eastern Illinois University Governors State University	\$21,302 \$35,614 \$30,986	\$87,078 \$114,978 \$04,756	\$14,251 \$15,574 \$20,645	\$35,437 \$46,106 \$56,909	
022 079	Illinois State University Northeastern Illinois University	\$39,271 \$26,768	\$94,756 \$124,112 \$95,205	\$20,643 \$13,936 \$15,870	\$36,909 \$48,047 \$44,572	
045 060	Northern Illinois University Southern Illinois University-Carbondale	\$33,548 \$30,585	\$115,048 \$110,336	\$12,120 \$11,795	\$53,962 \$48,138	
070 064	Southern Illinois University-Edwardsville University of Illinois-Chicago	\$33,530 \$33,623	\$109,920 \$114,454	\$14,033 \$12,174	\$48,867 \$47,365	
127 065	University of Illinois-Springfield University of Illinois-Urbana	\$33,741 \$40,366	\$111,305 \$137,830	\$18,257 \$9,528	\$57,244 \$38,821	
066	Western Illinois University	\$35,152	\$112,103	\$13,490	\$47,402	
	Mean Income Public 4-Year	\$34,167	\$119,612	\$14,181	\$47,131	
Private 4-Y	<u>Year</u>		NDENTS	INDEPE	NDENTS	
MAP <u>Code</u> 001	<u>Institution</u> Augustana College	Eligibles Mean Income of Parents \$51,481	Non-Eligibles * Mean Income of Parents \$149,783	Eligibles <u>Mean Income</u> \$11,075	Non-Eligibles * Mean Income \$59,636	
002 058	Aurora University Benedictine University	\$44,078 \$43,740	\$120,133 \$125,589	\$21,576 \$28,170	\$82,799 \$89,022	

Table 2.3d, Average Income by Dependency Status, continued 2011 ISAC Data Book

Private 4-	Year, continued	DEPE	DEPENDENTS		ENDENTS
MAP Code	Institution	Eligibles Mean Income of Parents	Non-Eligibles * Mean Income of Parents	Eligibles Mean Income	Non-Eligibles * Mean Income
005	Blackburn College	\$40,868	\$115,686	\$10,023	\$67,379
006	Bradley University	\$47,792	\$139,137	\$18,705	\$55,580
090	Columbia College	\$36,376	\$130,033	\$11,458	\$46,592
011	Concordia University	\$39,890	\$135,230	\$21,698	\$54,820
013	DePaul University	\$39,116	\$139,478	\$19,491	\$73,438
055	Dominican University	\$42,696	\$120,175	\$17,913	\$74,283
150	East West University	\$23,520	\$87,365	\$9,222	\$26,682
016	Elmhurst College	\$47,807	\$128,794	\$20,800	\$67,271
017	Eureka College	\$46,588	\$118,561	\$17,149	\$73,156
019	Greenville College	\$46,552	\$113,380	\$27,717	\$79,395
098	Hebrew Theological College	\$61,286	\$128,653	\$21,547	
020	Illinois College	\$46,859	\$121,834	\$14,340	\$18,400
021	Illinois Institute of Technology	\$38,826	\$124,617	\$15,579	\$59,331
023	Illinois Wesleyan University	\$51,299	\$143,602	\$7,184	\$22,107
083	Judson University	\$51,542	\$129,254	\$30,188	\$79,057
026	Knox College	\$44,763	\$137,930	\$12,113	\$45,606
027	Lake Forest College	\$44,139	\$143,978	\$11,561	\$82,548
029	Lewis University	\$43,966	\$126,404	\$25,429	\$89,649
160	Lexington College	\$31,989	\$138,111	\$18,711	\$121,426
091	Lincoln Christian University	\$42,045	\$104,759	\$25,438	\$62,457
031	Loyola University Chicago	\$42,051	\$146,339	\$16,042	\$56,196
034	MacMurray College	\$39,369	\$107,918	\$19,193	\$63,137
033	McKendree University	\$43,656	\$119,611	\$28,121	\$80,727
036	Millikin University	\$46,908	\$133,482	\$26,172	\$86,859
038	Monmouth College	\$45,489	\$126,385	\$11,066	\$19,728
043	National-Louis University	\$36,502	\$104,111	\$24,874	\$71,904
044	North Central College	\$49,356	\$134,115	\$17,640	\$66,892
046	North Park University	\$39,917	\$132,545	\$21,373	\$71,811
048	Northwestern University	\$43,844	\$153,174	\$22,896	\$72,632
049	Olivet Nazarene University	\$49,304	\$126,725	\$30,337	\$85,640
052	Quincy University	\$44,219	\$125,666	\$22,441	\$56,827

Table 2.3d, Average Income by Dependency Status, continued 2011 ISAC Data Book

Private 4-Y	<u>Year, continued</u>	DEPEN	NDENTS	INDEPENDENTS	
MAP		Eligibles Mean Income	Non-Eligibles * Mean Income	Eligibles	Non-Eligibles *
<u>Code</u>	<u>Institution</u>	of Parents	of Parents	Mean Income	Mean Income
007	Robert Morris University Illinois	\$31,792	\$115,108	\$18,633	\$72,405
053	Rockford College	\$45,175	\$113,845	\$22,594	\$69,969
054	Roosevelt University	\$37,671	\$125,875	\$20,074	\$68,392
059	Shimer College	\$28,819	\$203,647	\$12,863	\$51,643
152	St. Augustine College	\$21,610	\$81,148	\$20,400	\$58,313
069	St. Xavier University	\$40,576	\$123,402	\$19,929	\$69,182
144	Telshe Yeshiva	\$59,294	\$92,162	\$6,591	
068	The School of The Art Institute	\$38,441	\$139,402	\$9,685	\$97,211
062	The University of Chicago	\$43,312	\$143,322	\$11,570	\$114,814
076	Trinity Christian College	\$45,981	\$121,586	\$26,467	\$81,529
080	Trinity International Univ - Reach Prog	\$17,801		\$32,849	
081	Trinity International University	\$45,866	\$131,493	\$24,803	\$74,957
857	University of St. Francis - Prof Arts	\$21,579		\$51,246	
057	University of St. Francis	\$46,853	\$124,578	\$25,283	\$83,254
102	Vandercook College of Music	\$45,718	\$127,452	\$9,354	\$42,906
067	Wheaton College	\$47,584	\$146,009	\$17,062	\$49,154
	Mean Income Private 4-Year	\$41,622	\$133,184	\$20,247	\$72,649
Public 2-Y	<u>ear</u>	DEPENDENTS		INDEPENDENTS	
		Eligibles	Non-Eligibles *		
MAP		Mean Income	Mean Income	Eligibles	Non-Eligibles *
<u>Code</u>	Institution	of Parents	of Parents	Mean Income	Mean Income
103	Black Hawk College	\$25,453	\$84,762	\$16,127	\$54,296
106	Carl Sandburg College	\$25,558	\$86,627	\$15,828	\$51,579
032	College of DuPage	\$26,472	\$92,432	\$17,122	\$53,287
074	College of Lake County	\$25,323	\$89,157	\$16,306	\$59,641
012	Danville Area Community College	\$23,718	\$80,950	\$15,242	\$55,582
015	Elgin Community College	\$27,031	\$90,700	\$16,930	\$60,022
147	Frontier Community College	\$24,447	\$71,257	\$18,540	\$49,302
114	Harold Washington College	\$20,183	\$69,655	\$9,964	\$43,180
110	Harry S. Truman College	\$17,986	\$73,817	\$11,128	\$37,025
124	Heartland Community College	\$25,462	\$92,101	\$15,461	\$46,864
084	Highland Community College	\$23,384	\$81,587	\$17,353	\$55,621

Table 2.3d, Average Income by Dependency Status, continued 2011 ISAC Data Book

Public 2-Y	ear, continued	DEPENDENTS		INDEPE	INDEPENDENTS		
MAP		Eligibles Mean Income	Non-Eligibles * Mean Income	Eligibles	Non-Eligibles *		
Code	<u>Institution</u>	of Parents	of Parents	Mean Income	Mean Income		
056	Illinois Central College	\$25,071	\$91,545	\$16,002	\$51,383		
028	Illinois Valley Community College	\$26,805	\$82,440	\$18,325	\$51,278		
122	John A. Logan College	\$21,486	\$80,806	\$13,333	\$44,007		
140	John Wood Community College	\$26,440	\$87,225	\$19,227	\$50,353		
024	Joliet Junior College	\$26,010	\$90,884	\$18,194	\$64,148		
037	Kankakee Community College	\$23,717	\$80,622	\$15,809	\$58,338		
008	Kaskaskia College	\$24,530	\$81,844	\$16,906	\$49,482		
116	Kennedy-King College	\$15,964	\$67,859	\$9,893	\$59,565		
009	Kishwaukee College	\$24,218	\$88,688	\$15,604	\$61,988		
105	Lake Land College	\$27,886	\$87,662	\$18,292	\$51,808		
131	Lewis & Clark Community College	\$23,432	\$84,010	\$16,363	\$56,920		
118	Lincoln Land Community College	\$24,742	\$81,695	\$16,250	\$46,719		
126	Lincoln Trail College	\$22,286	\$101,132	\$16,988	\$51,196		
112	Malcolm X College	\$17,134	\$75,437	\$11,914	\$45,247		
120	McHenry County College	\$29,629	\$99,785	\$19,516	\$64,434		
121	Moraine Valley Community College	\$24,199	\$84,426	\$16,057	\$54,777		
040	Morton College	\$24,065	\$67,315	\$15,713	\$62,671		
130	Oakton Community College	\$22,764	\$85,687	\$15,668	\$46,060		
115	Olive Harvey College	\$16,853	\$71,976	\$11,301	\$45,926		
108	Olney Central College	\$24,413	\$78,385	\$16,985	\$65,068		
107	Parkland College	\$24,498	\$90,122	\$14,347	\$47,793		
073	Prairie State College	\$23,607	\$80,792	\$15,036	\$52,781		
041	Rend Lake College	\$23,544	\$75,343	\$14,333	\$66,185		
111	Richard J. Daley College	\$21,749	\$68,694	\$14,904	\$54,662		
133	Richland Community College	\$21,796	\$84,619	\$16,123	\$56,159		
085	Rock Valley College	\$25,155	\$84,866	\$15,606	\$71,874		
088	Sauk Valley Community College	\$26,536	\$81,871	\$17,893	\$48,511		
075	Shawnee Community College	\$20,407	\$75,723	\$13,334	\$47,181		
063	South Suburban College of Cook County	\$21,793	\$77,283	\$14,580	\$64,166		
078	Southeastern Illinois College	\$21,669	\$74,566	\$13,226	\$70,332		
004	Southwestern Illinois College	\$22,160	\$86,167	\$14,817	\$59,047		
077	Spoon River College	\$23,672	\$86,458	\$15,153	\$58,682		

Table 2.3d, Average Income by Dependency Status, continued 2011 ISAC Data Book

Public 2-Y	ear, continued	DEPEN	NDENTS	INDEPE	ENDENTS
MAP Code	<u>Institution</u>	Eligibles Mean Income of Parents	Non-Eligibles * Mean Income of Parents	Eligibles <u>Mean Income</u>	Non-Eligibles * Mean Income
047	Triton College	\$24,251	\$76,158	\$16,091	\$52,302
082	Wabash Valley College	\$24,210	\$79,984	\$18,697	\$48,350
096	Waubonsee Community College	\$26,334	\$84,112	\$18,762	\$60,894
117	Wilbur Wright College	\$21,259	\$74,901	\$14,033	\$50,581
087	William Rainey Harper College	\$27,047	\$93,034	\$17,774	\$54,201
	Mean Income Public 2-Year	\$23,465	\$86,204	\$15,124	\$55,293
Private 2-	<u>Year</u>	DEPEN	NDENTS	INDEPE	ENDENTS
		Eligibles	Non-Eligibles *		
MAP	T	Mean Income	Mean Income	Eligibles	Non-Eligibles *
Code	Institution	of Parents	of Parents	Mean Income	Mean Income
030	Lincoln College	\$31,057	\$125,549 \$112,020	\$19,807	\$76,161 \$52,506
092	MacCormac College	\$34,816	\$112,029	\$16,658	\$53,506
145 061	Morrison Institute of Tech	\$43,963 \$33,317	\$99,856 \$120,674	\$18,333 \$21,109	\$64,687
001	Springfield College in Illinois	\$33,317	\$120,074	\$21,109	\$0 4 ,067
	Mean Income Private 2-Year	\$32,253	\$121,620	\$19,375	\$69,004
Hospital S	chools	DEPEN	NDENTS	INDEPE	ENDENTS
MAP		Eligibles Mean Income	Non-Eligibles * Mean Income	Eligibles	Non-Eligibles *
Code	Institution	of Parents	of Parents	Mean Income	Mean Income
394	Advocate Trinity Hospital	\$28,560	\$72,290	\$12,351	\$59,962
358	Blessing-Rieman College of Nursing	\$48,254	\$103,472	\$20,287	\$59,661
172	Capital Area School of Prac Nursing	\$30,770	\$108,022	\$21,106	\$62,551
308	Graham Hospital School of Nursing	\$29,836	\$109,840	\$19,140	\$73,833
334	Lakeview College of Nursing	\$42,736	\$137,501	\$16,236	\$44,304
312	Methodist Medical Center of Illinois	\$39,645	\$118,495	\$26,130	\$82,372
200	National University of Health Sciences	\$41,412	\$130,012	\$22,063	\$97,699

Table 2.3d, Average Income by Dependency Status, continued 2011 ISAC Data Book

Hospital So	chools, continued	DEPEN	ENDENTS INDEPENDENTS		ENDENTS	
MAP <u>Code</u>	<u>Institution</u>	Eligibles Mean Income of Parents	Non-Eligibles * Mean Income of Parents	Eligibles <u>Mean Income</u>	Non-Eligibles * <u>Mean Income</u>	
389	Rush University Medical Technology	\$31,568	\$100,074	\$25,111	\$80,038	
318	St. Anthony College of Nursing	\$45,078	\$114,816	\$22,519	\$61,477	
321	St. Francis Medical Center College Nursing	\$47,584	\$123,172	\$26,025	\$85,518	
377	St. John's College	\$30,295	\$51,674	\$8,116		
390	St. John's College Dept of Nursing	\$43,956	\$106,263	\$19,389	\$34,817	
330	Trinity College of Nursing & Health Science	\$40,844	\$97,709	\$25,764	\$54,230	
337	West Suburban College of Nursing	\$37,955	\$110,507	\$17,054	\$61,463	
	Mean Income Hospital Schools	\$40,334	\$116,239	\$19,740	\$71,880	
Proprietar	oprietary Schools		DEPENDENTS		INDEPENDENTS	
MAP		Eligibles Mean Income	Non-Eligibles * Mean Income	Eligibles	Non-Eligibles *	
Code	<u>Institution</u>	of Parents	of Parents	Mean Income	Mean Income	
400	American Academy of Art	\$35,624	\$115,165	\$8,469	\$38,113	
173	Argosy University	\$27,656	\$114,228	\$16,845	\$64,783	
176	DeVry University-Chicago	\$29,343	\$109,436	\$22,686	\$73,523	
180	Fox College	\$33,831	\$110,346	\$15,184	\$77,229	
025	Kendall College	\$36,489	\$136,528	\$22,840	\$71,930	
174	Le Cordon Bleu College of Culinary Arts**	\$29,912	\$109,951	\$12,215	\$69,620	
170	Midstate College	\$26,704	\$92,724	\$21,343	\$75,976	
171	Northwestern College	\$25,707	\$102,574	\$16,981	\$63,616	
146	The Illinois Institute of Art	\$34,363	\$120,608	\$14,328	\$65,605	
	Mean Income Proprietary Schools	\$31,255	\$116,239	\$19,740	\$71,880	

^{*} Non-eligible mean income includes only FAFSA filers.

^{**}Formerly known as The Cooking and Hospitality Institute of Chicago

Table 2.3e of the 2011 ISAC Data Book 2010-2011 Tuition and Fees Used for Mean Weighted Tuition and Fees

Public 4-Year

MAP	ED				Tuition	
Code	Code	<u>Name</u>	Tuition	Fees	and Fees	<u>Term</u>
010	001694	Chicago State University	\$8,070	\$2,301	\$10,371	Semester
014	001674	Eastern Illinois University	\$8,128	\$2,415	\$10,543	Semester
129	009145	Governors State University	\$7,584	\$1,668	\$9,252	Semester
022	001692	Illinois State University	\$9,632	\$2,513	\$12,145	Semester
079	001693	Northeastern Il University	\$8,320	\$2,723	\$11,043	Semester
045	001737	Northern Illinois University	\$8,480	\$3,182	\$11,662	Semester
060	001758	Southern IL Univ-Carbondale - Base	\$7,290	\$3,171	\$10,461	Semester
070	001759	Southern Illinois University-Edwardsville	\$6,614	\$3,340	\$9,955	Semester
064	001776	University of Illinois at Chicago - Base	\$9,896	\$3,353	\$13,249	Semester
127	009333	University of Illinois at Springfield	\$9,134	\$4,310	\$13,444	Semester
065	001775	University of Illinois-Urbana - Base	\$10,386	\$3,122	\$13,508	Semester
066	001780	Western Illinois University	\$7,701	\$2,929	\$10,629	Semester

Private 4-Year

MAP	ED				Tuition	
Code	Code	<u>Name</u>	Tuition	Fees	and Fees	Term
001	001633	Augustana College	\$32,235	\$0	\$32,235	Quarter
002	001634	Aurora University	\$18,600	\$100	\$18,700	Semester
058	001767	Benedictine University	\$22,500	\$710	\$23,210	Semester
005	001639	Blackburn College	\$14,886	\$110	\$14,996	Semester
006	001641	Bradley University	\$25,150	\$274	\$25,424	Semester
090	001665	Columbia College	\$19,140	\$710	\$19,850	Semester
011	001666	Concordia University	\$23,918	\$488	\$24,406	Semester
013	001671	DePaul University	\$30,655	\$618	\$31,273	Quarter
055	001750	Dominican University	\$25,560	\$150	\$25,710	Semester
150	015310	East West University	\$15,000	\$795	\$15,795	Quarter
016	001676	Elmhurst College	\$28,600	\$60	\$28,660	Semester
148	001676	Elmhurst College-Management Program	\$26,048	\$60	\$26,108	Semester
017	001678	Eureka College	\$17,275	\$770	\$18,045	Semester

Table 2.3e, 2010-2011 Tuition and Fees Used for Mean Weighted Tuition and Fees, continued 2011 ISAC Data Book

MAP	ED				Tuition	
Code	<u>Code</u>	<u>Name</u>	<u>Tuition</u>	Fees	and Fees	<u>Term</u>
019	001684	Greenville College	\$20,766	\$158	\$20,924	Semester
319	001684	Greenville College Non-Traditional	\$13,120	\$0	\$13,120	Semester
098	001685	Hebrew Theological College	\$16,690	\$350	\$17,040	Semester
020	001688	Illinois College	\$22,350	\$450	\$22,800	Semester
021	001691	Illinois Institute of Technology	\$31,363	\$1,030	\$32,393	Semester
023	001696	Illinois Wesleyan University	\$35,076	\$180	\$35,256	Semester
083	001700	Judson University	\$24,150	\$630	\$24,780	Semester
026	001704	Knox College	\$32,679	\$345	\$33,024	Quarter
027	001706	Lake Forest College	\$34,885	\$640	\$35,525	Semester
029	001707	Lewis University	\$23,780	\$0	\$23,780	Semester
100	001707	Lewis University-Accelerated Programs	\$19,040	\$0	\$19,040	Semester
091	001708	Lincoln Christian College	\$14,592	\$0	\$14,592	Semester
191	001708	Lincoln Christian College	\$12,768	\$0	\$12,768	Semester
031	001710	Loyola University Chicago	\$31,040	\$1,074	\$32,114	Semester
034	001717	Macmurray College	\$18,838	\$472	\$19,310	Semester
033	001722	McKendree College	\$22,330	\$800	\$23,130	Semester
036	001724	Millikin University	\$26,780	\$645	\$27,425	Semester
038	001725	Monmouth College	\$26,900	\$0	\$26,900	Semester
043	001733	National Louis University	\$22,368	\$120	\$22,488	Quarter
044	001734	North Central College	\$27,894	\$240	\$28,134	Quarter
046	001735	North Park University	\$19,900	\$0	\$19,900	Semester
048	001739	Northwestern University, Evanston	\$39,840	\$383	\$40,223	Quarter
136	001739	Northwestern University, Chicago	\$17,160	\$0	\$17,160	Quarter
049	001741	Olivet Nazarene University	\$24,750	\$840	\$25,590	Semester
052	001745	Quincy University	\$22,240	\$760	\$23,000	Semester
252	001745	Quincy University Evening Division	\$9,600	\$480	\$10,080	Semester
007	001746	Robert Morris University Illinois	\$20,100	\$0	\$20,100	Quarter
053	001748	Rockford College	\$24,750	\$0	\$24,750	Semester
054	001749	Roosevelt University	\$23,000	\$0	\$23,000	Semester
059	001756	Shimer College	\$26,510	\$3,690	\$30,200	Semester
069	001768	St. Xavier University	\$24,790	\$730	\$25,520	Semester
144	013816	Telshe Yeshiva	\$12,000	\$0	\$12,000	Semester

Table 2.3e, 2010-2011 Tuition and Fees Used for Mean Weighted Tuition and Fees, continued 2011 ISAC Data Book

MAP	ED					
Code	Code	<u>Name</u>				<u>Term</u>
068	001753	The School of the Art Institute	\$35,550	\$570	\$36,120	Semester
062	001774	The University of Chicago	\$40,188	\$903	\$41,091	Quarter
076	001771	Trinity Christian College	\$21,398	\$120	\$21,518	Semester
080	001772	Trinity International Univ-Reach Program	\$15,040	\$50	\$15,090	Semester
081	001772	Trinity International University	\$22,970	\$400	\$23,370	Semester
057	001664	University of St. Francis	\$16,608	\$0	\$16,608	Semester
757	001664	University of St. Francis-Fast Track Nursing	\$24,292	\$450	\$24,742	Semester
857	001664	University of St. Francis-Professional Arts	\$16,608	\$0	\$16,608	Semester
102	001778	Vandercook College of Music	\$21,060	\$1,210	\$22,270	Semester
067	001781	Wheaton College	\$27,580	\$0	\$27,580	Semester

Public 2-Year

MAP	ED				Tuition	
Code	Code	<u>Name</u>	Tuition	Fees	and Fees	<u>Term</u>
103	001638	Black Hawk College	\$2,720	\$304	\$3,024	Semester
106	007265	Carl Sandburg College	\$4,304	\$0	\$4,304	Semester
032	006656	College of DuPage	\$6,345	\$955	\$7,300	Quarter
074	007694	College of Lake County	\$7,904	\$512	\$8,416	Semester
412	001669	Danville Area Community College-Cosmetology	\$11,112	\$384	\$11,496	Semester
012	001669	Danville Area Community College	\$2,848	\$384	\$3,232	Semester
015	001675	Elgin Community College	\$2,912	\$10	\$2,922	Semester
147	014090	Frontier Community College	\$2,272	\$490	\$2,762	Semester
114	001652	Harold Washington College	\$2,784	\$400	\$3,184	Semester
110	001648	Harry S. Truman College	\$2,784	\$400	\$3,184	Semester
124	030838	Heartland Community College	\$3,456	\$224	\$3,680	Semester
084	001681	Highland Community College	\$3,008	\$288	\$3,296	Semester
056	006753	Illinois Central College	\$2,976	\$0	\$2,976	Semester
028	001705	Illinois Valley Community College	\$2,188	\$246	\$2,434	Semester
122	008076	John A. Logan College	\$2,688	\$0	\$2,688	Semester
140	012813	John Wood Community College	\$3,744	\$320	\$4,064	Semester
024	001699	Joliet Junior College	\$2,432	\$864	\$3,296	Semester
037	007690	Kankakee Community College	\$2,848	\$320	\$3,168	Semester
008	001701	Kaskaskia College	\$2,464	\$352	\$2,816	Semester
116	001654	Kennedy-King College	\$2,784	\$400	\$3,184	Semester

Table 2.3e, 2010-2011 Tuition and Fees Used for Mean Weighted Tuition and Fees, continued 2011 ISAC Data Book

Public 2-Year, continued

MAP	ED					
Code	Code	<u>Name</u>				Term
416	001654	Kennedy-King College French Pastry	\$15,666	\$400	\$16,066	Semester
009	007684	Kishwaukee College	\$2,560	\$480	\$3,040	Semester
105	007644	Lake Land College	\$2,384	\$586	\$2,970	Semester
131	010020	Lewis & Clark Community College	\$2,784	\$544	\$3,328	Semester
118	007170	Lincoln Land Community College	\$9,056	\$352	\$9,408	Semester
418	007170	Lincoln Land Community College-Aviation	\$5,856	\$352	\$6,208	Semester
126	009786	Lincoln Trail College	\$2,272	\$490	\$2,762	Semester
112	001650	Malcolm X College	\$2,784	\$400	\$3,184	Semester
120	007691	McHenry County College	\$2,624	\$302	\$2,926	Semester
121	007692	Moraine Valley Community College	\$3,040	\$166	\$3,206	Semester
040	001728	Morton College	\$2,528	\$660	\$3,188	Semester
130	009896	Oakton Community College	\$2,912	\$113	\$3,025	Semester
115	001653	Olive Harvey College	\$2,784	\$400	\$3,184	Semester
108	001742	Olney Central College	\$2,272	\$490	\$2,762	Semester
107	007118	Parkland College	\$3,008	\$256	\$3,264	Semester
073	001640	Prairie State College	\$2,752	\$468	\$3,220	Semester
041	007119	Rend Lake College	\$2,720	\$96	\$2,816	Semester
111	001649	Richard J. Daley College	\$2,784	\$400	\$3,184	Semester
133	010879	Richland Community College	\$2,704	\$180	\$2,884	Semester
085	001747	Rock Valley College	\$2,272	\$366	\$2,638	Semester
088	001752	Sauk Valley Community College	\$2,944	\$224	\$3,168	Semester
075	007693	Shawnee Community College	\$2,784	\$0	\$2,784	Semester
063	001769	South Suburban College Cook County	\$3,200	\$504	\$3,704	Semester
078	001757	Southeastern Illinois College	\$2,656	\$128	\$2,784	Semester
004	001636	Southwestern Illinois College	\$2,880	\$96	\$2,976	Semester
077	001643	Spoon River College	\$2,816	\$1,184	\$4,000	Semester
047	001773	Triton College	\$2,816	\$254	\$3,070	Semester
082	001779	Wabash Valley College	\$2,272	\$490	\$2,762	Semester
096	006931	Waubonsee Community College	\$2,976	\$160	\$3,136	Semester
117	001655	Wilbur Wright College	\$2,784	\$400	\$3,184	Semester
087	003961	William Rainey Harper College	\$3,152	\$562	\$3,714	Semester
287	003961	William Rainey Harper College-Dental Hygiene	\$3,152	\$562	\$3,714	Semester

Table 2.3e, 2010-2011 Tuition and Fees Used for Mean Weighted Tuition and Fees, continued 2011 ISAC Data Book

Private 2-Year

MAP	ED				Tuition	
Code	Code	<u>Name</u>	<u>Tuition</u>	Fees	and Fees	Term
160	016942	Lexington College	\$23,500	\$1,900	\$25,400	Semester
030	001709	Lincoln College	\$21,000	\$0	\$21,000	Semester
092	001716	MacCormac College	\$12,000	\$740	\$12,740	Semester
145	008880	Morrison Institute of Technology	\$13,900	\$860	\$14,760	Semester
061	001761	Springfield College in Illinois	\$9,200	\$340	\$9,540	Semester
152	015415	St. Augustine College	\$11,200	\$0	\$11,200	Semester

Hospital Schools

MAP	ED				Tuition	
Code	Code	<u>Name</u>	Tuition	Fees	and Fees	<u>Term</u>
358	006214	Blessing Rieman College of Nursing	\$17,088	\$700	\$17,788	Semester
172	016426	Capital Area School of Practical Nursing	\$6,450	\$1,795	\$8,245	Semester
308	008938	Graham Hosp School of Nursing	\$10,934	\$380	\$11,314	Semester
334	010501	Lakeview College of Nursing	\$14,160	\$2,320	\$16,480	Semester
312	006228	Methodist College of Nursing	\$14,820	\$1,230	\$16,050	Semester
200	001732	National University of Health Sciences	\$9,376	\$146	\$9,522	Semester
389	009800	Rush University-Medical Technology	\$25,968	\$0	\$25,968	Quarter
318	009987	St. Anthony College of Nursing	\$18,886	\$458	\$19,344	Semester
321	006240	St. Francis Medical Center College Nursing	\$15,360	\$260	\$15,620	Semester
390	030980	St. Johns College-Dept. of Nursing	\$13,128	\$620	\$13,748	Semester
330	006225	Trinity College of Nursing & Health Science	\$18,144	\$948	\$19,092	Semester
337	022141	West Suburban College of Nursing	\$20,986	\$550	\$21,536	Semester

Table 2.3e, 2010-2011 Tuition and Fees Used for Mean Weighted Tuition and Fees, continued 2011 ISAC Data Book

Proprietary Schools

MAP	ED				Tuition	
Code	Code	<u>Name</u>	Tuition	Fees	and Fees	<u>Term</u>
400	001628	American Academy of Art	\$25,200	\$370	\$25,570	Semester
176	016219	DeVry University	\$16,560	\$80	\$16,640	Semester
180	016924	Fox College	\$14,580	\$0	\$14,580	Semester
025	001703	Kendall College / Day Div / Horizon (Evening)	\$22,095	\$825	\$22,920	Quarter
143	001703	Kendall College	\$12,000	\$825	\$12,825	Quarter
174	016758	Le Cordon Bleu College of Culinary Arts*	\$19,525	\$1,000	\$20,525	Semester
170	004568	Midstate College	\$13,200	\$0	\$13,200	Quarter
171	012362	Northwestern Business College - Base	\$20,640	\$420	\$21,060	Quarter
146	012584	The Illinois Institute of Art Chicago	\$23,184	\$267	\$23,451	Quarter
246	012584	The Illinois Institute of Art Schaumburg	\$22,512	\$50	\$22,562	Quarter

^{*} Formerly known as The Cooking and Hospitality Institute of Chicago



Table 2.4a of the 2011 ISAC Data Book: Monetary Award Eligibility by Income Level and Dependency Status-Historical Summary, FY2007-FY2011

		Percent of Announced Dependent Applicants Declared Eligible Y2007 FY2008 FY2009 FY2010 FY2011					Percent of Announced Independent Applicants Declared Eligible					Percent of Total Announced Applicants Declared Eligible			
Income*	FY2007	FY2008	FY2009	FY2010	FY2011	FY2007	FY2008	FY2009	FY2010	FY2011	FY2007	FY2008	FY2009	FY2010	FY2011
0- 5,000	99.0	99.3	99.2	99.5	99.5	97.2	97.4	97.6	98.4	98.6	97.6	97.8	98.0	98.7	98.8
5,001-10,000	98.2	98.9	98.7	99.3	99.3	96.8	97.3	97.6	98.2	98.5	97.1	97.6	97.9	98.4	98.7
10,001-15,000	98.2	98.9	98.8	99.4	99.4	96.7	96.8	96.9	98.4	98.5	97.2	97.5	97.5	98.7	98.8
15,001-20,000	97.5	98.4	98.4	99.1	99.2	87.4	88.9	88.7	92.6	94.4	91.2	92.4	92.2	95.0	96.2
20,001-25,000	95.4	95.3	95.1	98.5	98.7	79.4	78.9	77.6	79.6	80.0	86.3	85.7	84.9	87.3	87.9
25,001-30,000	92.5	93.8	93.7	97.9	98.4	75.2	74.9	73.7	76.0	77.6	83.7	84.2	83.3	86.1	87.7
30,001-35,000	89.8	91.1	90.3	94.0	94.8	67.3	67.4	67.7	70.3	70.8	79.7	80.5	79.8	82.4	83.8
35,001-40,000	82.9	84.8	85.5	88.8	90.6	66.7	65.7	65.9	66.7	69.7	76.8	77.3	77.5	79.3	82
40,001-45,000	75.1	76.8	76.1	80.2	84.1	63.7	66.1	64.8	67.8	69.8	71.4	73.2	72.2	75.4	78.9
45,001-50,000	66.3	67.2	67.0	71.1	75.1	56.0	58.6	59.3	66.9	70.1	63.5	64.7	64.6	69.6	73.4
50,001-55,000	55.2	57.8	55.8	60.6	64.1	45.7	49.2	49.8	61.2	67.5	52.9	55.7	54.1	60.8	65.1
55,001-60,000	43.3	46.9	46.1	48.3	55.7	34.9	39.3	38.7	50.4	61.3	41.6	45.2	44.2	48.9	57.2
60,001-65,000	33.0	35.3	35.6	39.9	45.6	26.3	28.6	30.1	39.4	54.0	31.8	34.0	34.4	39.8	47.7
65,001-70,000	24.1	27.2	28.3	30.5	35.9	15.7	21.0	20.9	29.7	42.3	22.7	26.1	26.7	30.3	37.3
OVER 70,000	5.4	5.8	5.5	6.0	8.0	6.9	8.3	7.9	10.0	15.4	5.5	6.0	5.7	6.5	8.8
,															

^{*} Reported Taxable Income--Reflects only parental AGI for dependent students.

Table 2.4b of the 2011 ISAC Data Book - Public 4-Year Institutions Monetary Award Eligibility by Income Level and Dependency Status, FY2011

PUBLIC 4-YEAR

				Average				Average				Average
Income*	# Apps	# Elig	% Elig	Award	# Apps	# Elig	% Elig	Award	# Apps	# Elig	% Elig	Award
0- 5,000	4,453	4,399	98.8%	\$4,646	10,064	9,583	95.2%	\$4,518	14,517	13,982	96.3%	\$4,558
5,001-10,000	2,091	2,069	98.9%	\$4,668	4,795	4,554	95.0%	\$4,526	6,886	6,623	96.2%	\$4,571
10,001-15,000	3,402	3,372	99.1%	\$4,681	4,057	3,853	95.0%	\$4,492	7,459	7,225	96.9%	\$4,580
15,001-20,000	3,741	3,696	98.8%	\$4,665	3,260	3,087	94.7%	\$4,150	7,001	6,783	96.9%	\$4,431
20,001-25,000	3,726	3,662	98.3%	\$4,674	2,453	1,972	80.4%	\$3,550	6,179	5,634	91.2%	\$4,280
25,001-30,000	4,128	4,076	98.7%	\$4,662	1,920	1,167	60.8%	\$4,131	6,048	5,243	86.7%	\$4,544
30,001-35,000	3,921	3,763	96.0%	\$4,550	1,396	831	59.5%	\$4,188	5,317	4,594	86.4%	\$4,485
35,001-40,000	3,668	3,477	94.8%	\$4,362	1,037	613	59.1%	\$4,303	4,705	4,090	86.9%	\$4,353
40,001-45,000	3,395	3,151	92.8%	\$4,164	744	451	60.6%	\$4,367	4,139	3,602	87.0%	\$4,189
45,001-50,000	3,302	2,887	87.4%	\$3,812	604	371	61.4%	\$4,311	3,906	3,258	83.4%	\$3,869
50,001-55,000	3,256	2,396	73.6%	\$3,505	481	302	62.8%	\$4,092	3,737	2,698	72.2%	\$3,572
55,001-60,000	3,094	1,874	60.6%	\$3,370	390	244	62.6%	\$3,746	3,484	2,118	60.8%	\$3,413
60,001-65,000	2,889	1,390	48.1%	\$3,331	341	192	56.3%	\$3,538	3,230	1,582	49.0%	\$3,356
65,001-70,000	2,852	1,031	36.2%	\$2,935	261	106	40.6%	\$3,194	3,113	1,137	36.5%	\$2,958
OVER 70,000	41,819	2,229	5.3%	\$2,527	1,310	188	14.4%	\$2,750	43,129	2,417	5.6%	\$2,544

^{*} Reported taxable income.

Table 2.4c of the 2011 ISAC Data Book - Public 2-Year Institutions Monetary Award Eligibility by Income Level and Dependency Status, FY2011

PUBLIC 2-YEAR

				Average				Average				Average
Income*	# Apps	# Elig	% Elig	Award	# Apps	# Elig	% Elig	Award	# Apps	# Elig	% Elig	Award
0- 5,000	11,237	11,222	99.9%	\$1,673	43,035	42,851	99.6%	\$1,636	54,272	54,073	99.6%	\$1,644
5,001-10,000	4,443	4,431	99.7%	\$1,689	17,902	17,836	99.6%	\$1,653	22,345	22,267	99.7%	\$1,660
10,001-15,000	7,234	7,215	99.7%	\$1,692	19,023	18,920	99.5%	\$1,703	26,257	26,135	99.5%	\$1,700
15,001-20,000	7,338	7,297	99.4%	\$1,692	15,696	14,637	93.3%	\$1,551	23,034	21,934	95.2%	\$1,598
20,001-25,000	7,395	7,315	98.9%	\$1,690	12,698	9,374	73.8%	\$1,636	20,093	16,689	83.1%	\$1,660
25,001-30,000	7,249	7,119	98.2%	\$1,675	9,877	7,381	74.7%	\$1,633	17,126	14,500	84.7%	\$1,654
30,001-35,000	6,361	5,918	93.0%	\$1,499	7,031	5,140	73.1%	\$1,742	13,392	11,058	82.6%	\$1,612
35,001-40,000	5,531	4,694	84.9%	\$1,348	5,238	3,857	73.6%	\$1,788	10,769	8,551	79.4%	\$1,546
40,001-45,000	4,566	3,245	71.1%	\$1,259	3,747	2,813	75.1%	\$1,793	8,313	6,058	72.9%	\$1,507
45,001-50,000	3,944	2,034	51.6%	\$1,173	2,962	2,208	74.5%	\$1,767	6,906	4,242	61.4%	\$1,482
50,001-55,000	3,702	1,375	37.1%	\$1,099	2,300	1,615	70.2%	\$1,672	6,002	2,990	49.8%	\$1,408
55,001-60,000	3,352	923	27.5%	\$1,003	1,862	1,101	59.1%	\$1,578	5,214	2,024	38.8%	\$1,316
60,001-65,000	2,970	497	16.7%	\$953	1,573	739	47.0%	\$1,533	4,543	1,236	27.2%	\$1,300
65,001-70,000	2,796	245	8.8%	\$875	1,258	406	32.3%	\$1,506	4,054	651	16.1%	\$1,269
OVER 70,000	19,204	287	1.5%	\$1,073	5,258	423	8.0%	\$1,343	24,462	710	2.9%	\$1,234
,	·			·	·			·	·			•

^{*} Reported taxable income.

Table 2.4d of the 2011 ISAC Data Book - Private Institutions Monetary Award Eligibility by Income Level and Dependency Status, FY2011

PRIVATE

T	Д А	# T71! -	0/ 1715-	Average	Д А	# T21! -	0/ 1711 -	Average	# 4	# Tale -	0/ EU-	Average
Income*	# Apps	# Elig	% Elig	Award	# Apps	# Elig	% Elig	Award	# Apps	# Elig	% Elig	Award
0- 5,000	3,069	3,043	99.2%	\$4,839	7,464	7,250	97.1%	\$4,819	10,533	10,293	97.7%	\$4,825
5,001-10,000	1,443	1,424	98.7%	\$4,861	3,361	3,260	97.0%	\$4,808	4,804	4,684	97.5%	\$4,824
10,001-15,000	2,318	2,293	98.9%	\$4,854	3,476	3,388	97.5%	\$4,821	5,794	5,681	98.0%	\$4,835
15,001-20,000	2,580	2,557	99.1%	\$4,847	3,054	2,978	97.5%	\$4,812	5,634	5,535	98.2%	\$4,828
20,001-25,000	2,748	2,704	98.4%	\$4,844	2,695	2,614	97.0%	\$4,781	5,443	5,318	97.7%	\$4,813
25,001-30,000	2,904	2,854	98.3%	\$4,855	2,308	2,160	93.6%	\$4,704	5,212	5,014	96.2%	\$4,790
30,001-35,000	2,764	2,686	97.2%	\$4,851	1,785	1,285	72.0%	\$4,712	4,549	3,971	87.3%	\$4,806
35,001-40,000	2,773	2,657	95.8%	\$4,854	1,459	935	64.1%	\$4,785	4,232	3,592	84.9%	\$4,836
40,001-45,000	2,487	2,353	94.6%	\$4,857	1,100	659	59.9%	\$4,779	3,587	3,012	84.0%	\$4,840
45,001-50,000	2,546	2,370	93.1%	\$4,846	916	587	64.1%	\$4,814	3,462	2,957	85.4%	\$4,839
50,001-55,000	2,476	2,207	89.1%	\$4,833	750	486	64.8%	\$4,791	3,226	2,693	83.5%	\$4,826
55,001-60,000	2,408	2,064	85.7%	\$4,782	644	409	63.5%	\$4,755	3,052	2,473	81.0%	\$4,778
60,001-65,000	2,378	1,803	75.8%	\$4,772	573	367	64.0%	\$4,781	2,951	2,170	73.5%	\$4,773
65,001-70,000	2,361	1,558	66.0%	\$4,678	494	310	62.8%	\$4,721	2,855	1,868	65.4%	\$4,685
OVER 70,000	30,101	4,642	15.4%	\$4,665	3,194	778	24.4%	\$4,651	33,295	5,420	16.3%	\$4,663

^{*} Reported taxable income.

Table 2.4e of the 2011 ISAC Data Book - Proprietary Institutions Monetary Award Eligibility by Income Level and Dependency Status, FY2011

PROPRIETARY

				Average				Average				Average
Income*	# Apps	# Elig	% Elig	Award	# Apps	# Elig	% Elig	Award	# Apps	# Elig	% Elig	Award
0- 5,000	883	878	99.4%	\$4,792	4,330	4,284	98.9%	\$4,822	5,213	5,162	99.0%	\$4,817
5,001-10,000	354	350	98.9%	\$4,798	1,878	1,816	99.1%	\$4,819	2,232	2,166	97.0%	\$4,816
10,001-15,000	501	500	99.8%	\$4,818	1,938	1,913	98.7%	\$4,819	2,439	2,413	98.9%	\$4,819
15,001-20,000	535	533	99.6%	\$4,786	1,931	1,904	98.6%	\$4,809	2,466	2,437	98.8%	\$4,804
20,001-25,000	534	531	99.4%	\$4,772	1,747	1,723	98.6%	\$4,770	2,281	2,254	98.8%	\$4,770
25,001-30,000	482	476	98.8%	\$4,789	1,524	1,421	93.2%	\$4,352	2,006	1,897	94.6%	\$4,462
30,001-35,000	440	424	96.4%	\$4,813	1,215	831	68.4%	\$4,658	1,655	1,255	75.8%	\$4,710
35,001-40,000	418	398	95.2%	\$4,793	928	635	68.4%	\$4,778	1,346	1,033	76.7%	\$4,784
40,001-45,000	373	348	93.3%	\$4,777	605	399	66.0%	\$4,819	978	747	76.4%	\$4,799
45,001-50,000	355	326	91.8%	\$4,801	518	338	65.3%	\$4,745	873	664	76.1%	\$4,772
50,001-55,000	304	263	86.5%	\$4,773	466	293	62.9%	\$4,742	770	556	72.2%	\$4,757
55,001-60,000	284	227	79.9%	\$4,573	340	230	67.6%	\$4,740	624	457	73.2%	\$4,657
60,001-65,000	264	190	72.0%	\$4,560	304	208	68.4%	\$4,663	568	398	70.1%	\$4,614
65,001-70,000	205	118	57.6%	\$4,471	252	136	54.0%	\$4,557	457	254	55.6%	\$4,517
OVER 70,000	2,195	332	15.1%	\$4,357	1,456	333	22.9%	\$4,015	3,651	665	18.2%	\$4,186

^{*} Reported taxable income.



Table 2.5a of the 2011 ISAC Data Book Monetary Award Applicant Distribution by Class Level FY2007-FY2011

FISCAL YEARS								
CLASS LEVEL	2006-07	2007-08	2008-09	2009-10	2010-11			
Freshmen	42.5%	42.5%	44.3%	47.5%	46.9%			
Sophomore	22.7%	22.6%	22.5%	21.5%	22.6%			
Junior	18.5%	18.7%	18.0%	16.9%	16.9%			
Senior	16.3%	16.2%	15.2%	14.0%	13.6%			

Table 2.5b of the 2011 ISAC Data Book Monetary Award Applicant Distribution by Age FY2007-FY2011

FISCAL YEARS								
AGE	2006-07	2007-08	2008-09	2009-10	2010-11			
18 or under	11.3%	11.3%	11.6%	11.0%	10.7%			
19	14.3%	14.6%	14.5%	14.4%	13.9%			
20	12.4%	12.6%	12.5%	12.1%	12.5%			
21	11.5%	11.2%	11.1%	10.8%	10.8%			
22-25	21.6%	21.6%	20.9%	20.4%	20.4%			
Over 25	28.9%	28.7%	29.4%	31.3%	31.7%			

Table 2.6a of the 2011 ISAC Data Book Characteristics of Announced Eligible Dependent MAP Applicants FY2007-FY2011

		FY2007	FY2008	FY2009	FY2010	FY2011
NUMBER ELIGIBLE:		104,469	106,483	113,655	131,980	150,398
MEAN ANNOUNCED MAP GRANT:	Overall	\$3,534	\$3,500	\$3,416	\$3,343	\$3,366
	Public 4-Year	\$4,229	\$4,250	\$4,261	\$4,304	\$4,348
	Public 2-Year	\$1,609	\$1,594	\$1,596	\$1,605	\$1,572
	Private 4-Year	\$4,906	\$4,902	\$4,897	\$4,906	\$4,932
	Private 2-Year	\$4,761	\$4,750	\$4,737	\$4,768	\$4,770
	Other	\$4,559	\$4,846	\$4,574	\$4,526	\$4,631
	Proprietary	\$4,857	\$4,650	\$4,820	\$4,843	\$4,911
APPLICANT DISTRIBUTION:	Public 4-Year	33%	32%	31%	29%	29%
	Public 2-Year	35%	36%	39%	42%	42%
	Private 4-Year	27%	27%	25%	24%	24%
	Private 2-Year	1%	1%	1%	1%	1%
	Other	0%	0%	0%	0%	0%
	Proprietary	4%	4%	4%	4%	4%
CLASS LEVEL:	Freshmen	44%	44%	46%	48%	47%
	Sophomores	23%	23%	23%	23%	23%
	Other Undergraduates	33%	33%	31%	29%	30%
ILLINOIS REGIONS:	Chicago (Zip 606)	27%	27%	28%	28%	27%
	Collar Area (600-605, 607, 608)	41%	42%	42%	44%	46%
	All Other Areas	32%	31%	30%	28%	27%
PARENTS:	Mean Age Oldest Parent	48	48	48	48	48
	% Married	46%	45%	42%	43%	44%
	% With Assets	70%	71%	70%	69%	68%
	Mean Assets	\$9,705	\$9,413	\$9,359	\$10,386	\$10,265
	% With Tax Income	91%	90%	90%	90%	90%
	Mean Tax Income	\$28,862	\$29,263	\$28,815	\$29,103	\$29,655
	% With Non-Tax Income **	72%	73%	74%		
HOUSEHOLD:	Mean Size	3.9	3.9	3.9	3.9	3.9
	Mean # in College	1.4	1.4	1.4	1.4	1.4
STUDENTS:	Mean Age	20	20	20	20	20
	% With Taxable Income	67%	67%	67%	64%	58%
	Mean Taxable Income	\$5,575	\$5,755	\$5,925	\$6,258	\$6,165
EXPECTED FAMILY CONTRIBUTION:	Percent Zero EFC	32%	37%	38%	51% #	50%
	Mean Federal EFC	\$2,151	\$2,009	\$1,928	\$1,546	\$1,437
	Mean ISAC Adjusted EFC	\$4,106	\$4,025	\$3,907	\$3,576	\$3,579
FFELP LOANS: *	% With ISAC Sub/Unsub Loans	27%	27%	20%	13%	*
	Mean Sub/Unsub Loan Debt if > 0	\$8,235	\$8,481	\$8,705	\$6,870	*
	% With ISAC Plus Loans	4%	4%	3%	2%	*
	Mean Plus Loan Debt	\$9,063	\$9,939	\$10,685	\$10,592	*

^{*} FFELP eliminated in 2010.

^{**}Data unavailable in FY2010. FAFSA Worksheet A, which contained several non-taxable income types, was eliminated.

[#] Reflects changes in Federal criteria for Zero EFC Students.

Table 2.6b of the 2011 ISAC Data Book Characteristics of Announced Eligible Independent MAP Applicants FY2007-FY2011

		FY2007	FY2008	FY2009	FY2010	FY2011
NUMBER ELIGIBLE:		131,837	132,972	145,678	182,218	200,790
MEAN ANNOUNCED MAP GRANT:	Overall	\$2,951	\$2,951	\$2,886	\$2,777	\$2,604
	Public 4-Year	\$4,428	\$4,423	\$4,418	\$4,439	\$4,447
	Public 2-Year	\$1,710	\$1,705	\$1,705	\$1,687	\$1,686
	Private 4-Year	\$4,897	\$4,890	\$4,889	\$4,899	\$4,959
	Private 2-Year	\$4,855	\$4,853	\$4,826	\$4,858	\$4,926
	Other	\$4,659	\$4,640	\$4,762	\$4,779	\$4,901
	Proprietary	\$4,809	\$4,803	\$4,827	\$4,838	\$4,929
APPLICANT DISTRIBUTION:	Public 4-Year	17%	16%	15%	14%	14%
	Public 2-Year	58%	58%	60%	64%	64%
	Private 4-Year	16%	16%	14%	12%	13%
	Private 2-Year	2%	2%	2%	1%	0%
	Other	0%	0%	1%	1%	1%
	Proprietary	7%	8%	8%	8%	8%
CLASS LEVEL:	Freshmen	48%	48%	49%	54%	56%
	Sophomores	23%	22%	22%	20%	23%
	Other Undergraduates	29%	30%	29%	26%	21%
ILLINOIS REGIONS:	Chicago (Zip 606)	29%	29%	29%	28%	27%
	Collar Area (600-605, 607, 608)	33%	33%	34%	35%	35%
	All Other Areas	38%	38%	37%	37%	38%
STUDENTS:	Mean Age	30	30	30	30	31
	% Married	20%	19%	18%	19%	25%
	% With Assets	55%	55%	55%	55%	53%
	Mean Assets	\$1,174	\$1,293	\$1,261	\$1,478	\$1,676
	% With Tax Income	85%	84%	85%	84%	87%
	Mean Tax Income	\$14,321	\$14,693	\$15,004	\$15,547	\$19,956
	% With Non-Tax Income **	63%	63%	63%		
EXPECTED FAMILY CONTRIBUTION:	Percent Zero EFC	57%	59%	60%	72% #	81%
	Mean Federal EFC	\$1,001	\$967	\$945	\$732	\$662
	Mean ISAC Adjusted EFC	\$2,630	\$2,608	\$2,577	\$2,425	\$2,096
HOUSEHOLD:	Mean Size	2.5	2.4	2.5	2.5	2.4
	Mean # in College	1.1	1.1	1.1	1.1	1.2
FFELP LOANS: *	% With ISAC Sub/Unsub Loans	28%	29%	24%	19%	*
	Mean Sub/Unsub Loan Debt if > 0	\$9,816	\$10,269	\$10,115	\$7,851	*

^{*} FFELP eliminated in 2010.

^{**}Data unavailable in FY2010. FAFSA Worksheet A, which contained several non-taxable income types, was eliminated.

[#] Reflects changes in Federal criteria for Zero EFC Students.

Table 2.6c of the 2011 ISAC Data Book Characteristics of Announced Eligible Dependent/Independent Combined MAP Applicants FY2007-FY2011

		FY2007	FY2008	FY2009	FY2010	FY2011
NUMBER ELIGIBLE:		236,306	239,455	259,333	314,198	351,188
MEAN ANNOUNCED MAP GRANT:	Overall	\$3,209	\$3,195	\$3,118	\$3,015	\$3,013
	Public 4-Year	\$4,306	\$4,317	\$4,322	\$4,357	\$4,400
	Public 2-Year	\$1,678	\$1,668	\$1,669	\$1,660	\$1,648
	Private 4-Year	\$4,902	\$4,897	\$4,893	\$4,903	\$4,939
	Private 2-Year	\$4,821	\$4,816	\$4,794	\$4,828	\$4,831
	Other	\$4,635	\$4,642	\$4,721	\$4,728	\$4,776
	Proprietary	\$4,823	\$4,810	\$4,825	\$4,839	\$4,889
APPLICANT DISTRIBUTION:	Public 4-Year	24%	24%	23%	20%	21%
	Public 2-Year	48%	48%	51%	56%	54%
	Private 4-Year	21%	21%	19%	17%	18%
	Private 2-Year	1%	1%	1%	1%	1%
	Other	0%	0%	0%	0%	0%
	Proprietary	6%	6%	6%	6%	6%
CLASS LEVEL:	Freshmen	46%	46%	48%	51%	50%
	Sophomores	23%	23%	22%	21%	22%
	Other Undergraduates	31%	31%	30%	28%	28%
ILLINOIS REGIONS:	Chicago (Zip 606)	28%	28%	29%	28%	28%
	Collar Area (600-605, 607, 608)	37%	37%	37%	39%	40%
	All Other Areas	35%	35%	34%	33%	32%
PARENTS OF DEPENDENT STUDENTS/	% With Assets	62%	62%	62%	61%	60%
INDEPENDENT STUDENTS:	Mean Assets	\$4,945	\$4,903	\$4,809	\$5,219	\$5,197
	% With Tax Income	87%	87%	87%	86%	85%
	Mean Tax Income #	\$20,749	\$21,172	\$21,055	\$21,241	\$21,390
	% With Non-Tax Income **	67%	68%	68%		
EXPECTED FAMILY CONTRIBUTION:	Percent Zero EFC	46%	49%	50%	63% #	64%
	Mean Federal EFC	\$1,510	\$1,431	\$1,375	\$1,074	\$984
	ISAC Adjusted EFC	\$3,282	\$3,238	\$3,160	\$2,908	\$2,881
HOUSEHOLD:	Mean Size	3.1	3.1	3.1	3.1	3.1
	Mean # in College	1.2	1.2	1.2	1.2	1.3
FFELP LOANS: *	% With ISAC Sub/Unsub Loans	28%	28%	23%	16%	*
	Mean Sub/Unsub Loan Debt if > 0	\$9,137	\$9,500	\$9,566	\$7,528	*

[#] Mean Taxable Income does not include dependent student income. * FFELP eliminated in 2010.

^{**}Data unavailable in FY2010. FAFSA Worksheet A, which contained several non-taxable income types, was eliminated.

[#] Reflects changes in Federal criteria for Zero EFC Students.

Table 2.6d of the 2011 ISAC Data Book Characteristics of Paid Dependent MAP Applicants FY2007-FY2011

		FY2007	FY2008	FY2009	FY2010	FY2011
NUMBER PAID:		78,258	78,681	79,411	79,789	85,788
MEAN MAP GRANT:	Overall	\$2,613	\$3,146	\$3,149	\$3,251	\$3,206
	Public 4-Year	\$3,641	\$3,688	\$3,758	\$3,833	\$3,761
	Public 2-Year	\$1,089	\$1,097	\$1,098	\$1,126	\$1,080
	Private 4-Year	\$4,418	\$4,430	\$4,451	\$4,490	\$4,367
	Private 2-Year	\$3,864	\$3,867	\$3,937	\$3,976	\$4,160
	Other	\$3,684	\$3,715	\$3,732	\$3,685	\$3,733
	Proprietary	\$3,412	\$3,362	\$3,378	\$3,401	\$3,364
APPLICANT DISTRIBUTION:	Public 4-Year	37%	37%	37%	36%	37%
	Public 2-Year	29%	29%	30%	29%	28%
	Private 4-Year	30%	30%	29%	31%	31%
	Private 2-Year	1%	1%	1%	1%	1%
	Other	0%	0%	0%	0%	0%
	Proprietary	3%	3%	3%	3%	3%
CLASS LEVEL:	Freshmen	38%	38%	40%	41%	39%
	Sophomores	25%	25%	25%	24%	25%
	Other Undergraduates	37%	37%	35%	35%	36%
ILLINOIS REGIONS:	Chicago (Zip 606)	26%	26%	26%	27%	27%
	Collar Area (600-605, 607, 608)	42%	43%	44%	46%	47%
	All Other Areas	32%	31%	30%	27%	26%
PARENTS:	Mean Age Oldest Parent	48	49	49	49	49
	% Married	48%	47%	45%	47%	48%
	% With Assets	74%	74%	74%	75%	74%
	Mean Assets	\$10,566	\$10,302	\$10,259	\$11,815	\$11,996
	% With Tax Income	92%	92%	92%	92%	93%
	Mean Tax Income	\$30,347	\$30,788	\$30,690	\$31,582	\$32,687
	% With Non-Tax Income **	73%	75%	76%		
HOUSEHOLD:	Mean Size	3.9	3.9	3.9	3.9	3.9
	Mean # in College	1.4	1.4	1.4	1.4	1.4
STUDENTS:	Mean Age	20	20	20	20	20
	% With Taxable Income	69%	70%	69%	67%	61%
	Mean Taxable Income	\$5,317	\$5,508	\$5,589	\$5,621	\$5,410
EXPECTED FAMILY CONTRIBUTION:	Percent Zero EFC	29%	33%	34%	45% #	44%
	Mean Federal EFC	\$2,297	\$2,160	\$2,102	\$1,780	\$1,712
	Mean ISAC Adjusted EFC	\$4,270	\$4,199	\$4,111	\$3,862	\$3,824
FFELP LOANS: *	% With ISAC Sub/Unsub Loans	31%	31%	23%	15%	*
	Mean Sub/Unsub Loan Debt if > 0	\$8,525	\$8,744	\$9,035	\$7,108	*
	% With ISAC Plus Loans	5%	5%	3%	2%	*
	Mean Plus Loan Debt	\$9,041	\$9,909	\$10,720	\$10,561	*

^{*} FFELP eliminated in 2010.

^{**}Data unavailable in FY2010. FAFSA Worksheet A, which contained several non-taxable income types, was eliminated.

[#] Reflects changes in Federal criteria for Zero EFC Students.

Table 2.6e of the 2011 ISAC Data Book Characteristics of Paid Independent MAP Applicants FY2007-FY2011

		FY2007	FY2008	FY2009	FY2010	FY2011
NUMBER PAID:		68,377	66,862	64,819	61,591	61,422
MEAN MAP GRANT:	Overall	\$2,009	\$2,038	\$2,064	\$2,129	\$2,088
	Public 4-Year	\$3,026	\$3,070	\$3,089	\$3,165	\$3,121
	Public 2-Year	\$925	\$933	\$941	\$959	\$948
	Private 4-Year	\$3,382	\$3,376	\$3,414	\$3,494	\$3,363
	Private 2-Year	\$3,071	\$3,121	\$3,154	\$3,302	\$2,966
	Other	\$3,438	\$3,410	\$3,593	\$3,605	\$3,417
	Proprietary	\$2,759	\$2,801	\$2,836	\$2,820	\$2,845
APPLICANT DISTRIBUTION:	Public 4-Year	21%	21%	21%	21%	21%
	Public 2-Year	51%	50%	50%	49%	49%
	Private 4-Year	19%	19%	19%	19%	19%
	Private 2-Year	2%	2%	2%	2%	1%
	Other	1%	1%	1%	1%	1%
	Proprietary	6%	7%	7%	9%	9%
CLASS LEVEL:	Freshmen	36%	36%	34%	35%	33%
	Sophomores	26%	25%	26%	25%	27%
	Other Undergraduates	38%	39%	40%	40%	40%
ILLINOIS REGIONS:	Chicago (Zip 606)	28%	28%	28%	29%	29%
	Collar Area (600-605, 607, 608)	33%	33%	34%	35%	36%
	All Other Areas	39%	39%	38%	36%	35%
STUDENTS:	Mean Age	30	30	30	30	30
	% Married	20%	19%	19%	19%	19%
	% With Assets	59%	60%	61%	62%	62%
	Mean Assets	\$1,230	\$1,344	\$1,387	\$1,553	\$1,646
	% With Tax Income	88%	87%	89%	88%	84%
	Mean Tax Income	\$14,655	\$15,049	\$15,460	\$16,028	\$15,824
	% With Non-Tax Income **	65%	65%	65%		
EXPECTED FAMILY CONTRIBUTION:	Percent Zero EFC	55%	57%	57%	69% #	71%
	Mean Federal EFC	\$1,064	\$1,039	\$1,048	\$832	\$748
	Mean ISAC Adjusted EFC	\$2,672	\$2,651	\$2,646	\$2,495	\$2,394
HOUSEHOLD:	Mean Size	2.4	2.4	2.4	2.4	2.3
	Mean # in College	1.1	1.1	1.1	1.1	1.1
FFELP LOANS: *	% With ISAC Sub/Unsub Loans	35%	37%	32%	25%	*
	Mean Sub/Unsub Loan Debt if > 0	\$11,434	\$11,861	\$11,796	\$9,198	*

^{*} FFELP eliminated in 2010.

^{**}Data unavailable in FY2010. FAFSA Worksheet A, which contained several non-taxable income types, was eliminated.

[#] Reflects changes in Federal criteria for Zero EFC Students.

Table 2.6f of the 2011 ISAC Data Book Characteristics of Paid Dependent/Independent MAP Applicants FY2007-FY2011

		FY2007	FY2008	FY2009	FY2010	FY2011
NUMBER PAID:		146,635	145,543	144,230	141,380	147,210
MEAN MAP GRANT:	Overall	\$2,613	\$2,637	\$2,662	\$2,762	\$2,740
	Public 4-Year	\$3,439	\$3,487	\$3,543	\$3,627	\$3,576
	Public 2-Year	\$989	\$999	\$1,008	\$1,012	\$1,006
	Private 4-Year	\$4,048	\$4,056	\$4,098	\$4,174	\$4,058
	Private 2-Year	\$3,434	\$3,453	\$3,494	\$3,582	\$3,735
	Other	\$3,490	\$3,489	\$3,627	\$3,626	\$3,506
	Proprietary	\$2,974	\$2,981	\$3,009	\$3,003	\$3,013
APPLICANT DISTRIBUTION:	Public 4-Year	30%	29%	29%	30%	30%
	Public 2-Year	39%	39%	39%	37%	36%
	Private 4-Year	25%	25%	25%	25%	26%
	Private 2-Year	1%	1%	1%	1%	1%
	Other	0%	1%	1%	1%	1%
	Proprietary	5%	5%	5%	6%	6%
CLASS LEVEL:	Freshmen	37%	37%	37%	38%	37%
	Sophomores	25%	25%	25%	25%	26%
	Other Undergraduates	38%	38%	38%	37%	37%
ILLINOIS REGIONS:	Chicago (Zip 606)	27%	27%	27%	28%	28%
	Collar Area (600-605, 607, 608)	38%	39%	39%	41%	42%
	All Other Areas	35%	34%	34%	31%	30%
PARENTS OF DEPENDENT STUDENTS/	% With Assets	67%	67%	68%	69%	69%
INDEPENDENT STUDENTS:	Mean Assets	\$6,213	\$6,187	\$6,272	\$7,345	\$7,677
	% With Tax Income	90%	90%	91%	90%	89%
	Mean Tax Income #	\$23,030	\$23,558	\$23,845	\$24,806	\$25,651
	% With Non-Tax Income **	69%	70%	71%		
EXPECTED FAMILY CONTRIBUTION:	Percent Zero EFC	41%	44%	44%	56% ##	55%
	Mean Federal EFC	\$1,722	\$1,645	\$1,628	\$1,367	\$1,310
	ISAC Adjusted EFC	\$3,525	\$3,488	\$3,453	\$3,266	\$3,227
HOUSEHOLD:	Mean Size	3.2	3.2	3.2	3.2	3.3
	Mean # in College	1.3	1.3	1.3	1.3	1.3
FFELP LOANS: *	% With ISAC Sub/Unsub Loans	33%	34%	27%	19%	*
	Mean Sub/Unsub Loan Debt if > 0	\$9,981	\$10,300	\$10,488	\$8,278	*

[#] Mean Taxable Income does not include dependent student income.

^{*} FFELP eliminated in 2010.

^{**}Data unavailable in FY2010. FAFSA Worksheet A, which contained several non-taxable income types, was eliminated.

^{##} Reflects changes in Federal criteria for Zero EFC Students.

The 2009-10 academic year was the final year for the Federal Family Education Loan Program (FFELP). Effective July 1, 2010, all new federal
student loans come directly from the U.S. Department of Education (ED) under the William D. Ford Federal Direct Loan Program (Federal Direct Loans). As a result, the Illinois Student Assistance Commission no longer originates or services FFELP Loans.
Tables 3.0 - 3.3, which reported ISAC's FFELP Loan Guarantee Volume and Borrower Characteristics prior to 2011, are no longer relevant and have been discontinued in the ISAC Data Book. Previous years' tables containing this information may be found on ISAC's website at www.isac.org under Research & Policy in the Data Book section.
PART THREE FEDERAL FAMILY EDUCATION LOAN PROGRAM

For more than 30 years, the Illinois Designated Account Purchase Program (IDAPP) helped students pay for college at schools all over Illinois
Due to the enactment of the Health Care and Education Reconciliation Act of 2010, IDAPP is no longer a Federal Family Education Loan Program (FFELP) lender for Federal Stafford, PLUS and/or GradPLUS Loans. All colleges and universities now process new loans through the U.S. Departmen of Education's William D. Ford Federal Direct Loan Program (Federal Direct Loans). Tables 4.0 - 4.2, containing historical loan program data, have been discontinued. 4Previous years' tables containing this information may be found on ISAC's website at www.collegeillinois.org in Research & Policy in the Data Book section.
PART FOUR ILLINOIS DESIGNATED ACCOUNT PURCHASE PROGRAM

ISAC administers numerous scholarship and grant programs, and one tuition waiver program. The State Scholar Program, an honorary program, recognizes academic achievement of high school seniors. Eligibility for the Student-to-Student Grant Program, the Silas Purnell Illinois Incentive for Access Program, and the Higher Education License Plate Grant Program is based on financial need. The remaining programs are financial assistance programs for students who meet special requirements.

Silas Purnell Illinois Incentive for Access Grant Program: The purpose of the Silas Purnell Illinois Incentive for Access (IIA) Program is to improve access and retention for students who have a limited ability to pay for college, and possibly to reduce the amount borrowed by these students. The program provides a \$500 or \$1,000 award (depending on funding) for freshman students who are determined through federal need analysis to have no family resources, and are attending approved Illinois colleges and universities at least half time. Funding was not appropriated for the IIA Grant Program in the FY2011 state budget, therefore, no awards were issued for this program in FY2011. Tables 5.4a and 5.4b provide historical and institutional data concerning the program.

Robert C. Byrd Honors Scholarship: The Byrd Scholarship Program is a federally-funded program administered by ISAC which provides scholarships of up to \$1,500, for a maximum of four academic years, for academically exceptional high school graduates who show promise of continued academic excellence. Byrd Scholars must become high school graduates in the same high school year in which a scholarship application is submitted, must demonstrate academic achievement through test scores and high school transcripts and be enrolled, or accepted for enrollment as full-time undergraduate students in a postsecondary institution approved by the U.S. Department of Education. This scholarship is not limited to tuition and fees. Recipients may use the scholarship at out-of-state institutions. A total of \$1,942,500 to 1,316 students was awarded in FY2011. Table 5.1 provides institutional data on this program.

Illinois Veteran Grant Program: The Illinois Veteran Grant (IVG) Program pays for tuition and mandatory fees at Illinois public universities and community colleges for qualified Illinois veterans or military service members with at least one year of active duty in the U.S. Armed Forces and who served honorably. Any member of the Illinois National Guard or a Reserve component of the U.S. Armed Forces who meets the eligibility requirements is considered a qualified applicant for the program. This grant is available for the equivalent of four academic years of full-time enrollment for undergraduate and graduate study. Veterans must have been residents of Illinois six months prior to entering the service, and must have returned to Illinois to reside within six months of leaving the service. Recipients are required to enroll for a minimum number of credit hours each term. Funding was not appropriated for the Illinois Veteran Grant Program in the FY2011 state budget.

<u>Illinois National Guard Grant Program</u>: The Illinois National Guard Grant Program pays tuition and certain fees for enlisted members and officers of the Illinois National Guard to attend public four or two-year institutions for undergraduate or graduate study. Students are eligible for eight semesters or twelve quarters of assistance. In FY2011, there were 1,294 students that participated in the program. The total dollars expended were \$2,691,922. Table 5.0b and 5.0d provides data for the program.

<u>Dependents' Grant Programs</u>: The Grant Program for Dependents of Police or Fire Officers and the Grant Program for Dependents of Correctional Officers pay the tuition and mandatory fees at any MAP-approved school for the spouse and children of police officers, fire officers, and correctional workers who were killed or permanently disabled in the line of duty. In FY2011, 87 students received \$803,741 in grant aid. Table 5.0b shows data for this program.

Merit Recognition Scholarship Program: The Merit Recognition Scholarship (MRS) Program provides a one-time \$1,000 scholarship to the top five percent of Illinois high school graduates, based on available funding each year from the General Assembly. Scholarships may be used at eligible Illinois institutions or one of the four approved United States military service academies. Recipients must use the award within one year of high school graduation, and must be enrolled for undergraduate study at least half time. The MRS Program has not been funded since the 2004-05 academic year.

PART FIVE -- SPECIALIZED SCHOLARSHIP AND GRANT PROGRAMS

(continued)

Minority Teachers of Illinois Scholarship Program: The Minority Teachers of Illinois (MTI) Scholarship Program provides scholarships of up to \$5,000 per year for up to four years to assist academically talented individuals of African American/Black, Hispanic American, Asian American, or Native American origin who plan to become teachers. Scholarships are available for undergraduate or graduate students enrolled at least half time. Students receiving this scholarship must fulfill a teaching commitment by teaching one year for each year of assistance at a nonprofit Illinois public, private, or parochial preschool, elementary or secondary school with at least 30 percent minority enrollment. If the teaching commitment is not fulfilled, the scholarship converts to a loan and the student must repay a pro-rated amount plus interest. In FY2011, 241 MTI scholarships were awarded, totaling \$1,160,922. Table 5.0c contains data by institution on this program.

<u>Illinois Future Teacher Corp Program</u>: The Illinois Future Teacher Corp Program awards scholarships to academically talented students attending an approved Illinois public or private institution, with a priority given to minority students, who plan to pursue careers as preschool, elementary and secondary school teachers in designated teacher shortage disciplines in and/or making a commitment to teach in a hard to staff school in the State of Illinois, and minority students. The scholarships are applicable only toward tuition and fees and room and board charges or commuter allowance. The annual scholarship awarded to a qualified applicant may be \$5,000 or \$10,000 depending on the teaching commitment made. Recipients must enroll at least half time as juniors or above and must fulfill the teaching commitment or repay funds received plus interest. Table 5.0b and 5.0c provides additional data on this program.

<u>Illinois Special Education Tuition Waiver Program</u>: The Illinois Special Education Teacher Tuition Waiver Program encourages current teachers and academically talented students to pursue careers in any area of special education as public, private, or parochial preschool, elementary or secondary school teachers in Illinois. Recipients must be seeking initial certification in any area of special education as undergraduate or graduate students. For non-teachers, students must be ranked in the upper half of their Illinois high school graduating class. Recipients are exempt from paying tuition and fees at an eligible institution for up to four calendar years. Recipients must fulfill a teaching requirement or repay funds received plus interest. Table 5.5 provides historical recipient data on the waiver program.

<u>Illinois Teachers Loan Repayment Program</u>: This program helps teachers who have served in low-income schools repay their student loans. Through the Illinois Teacher and Child Care Provider Loan Repayment Program, teachers who qualify for the federal Stafford Loan Cancellation for Teachers Program by teaching in an Illinois low-income school can receive an additional matching grant up to \$5,000. Childcare providers can also qualify for these federal and state programs by working full-time in a childcare facility that serves a low-income community in Illinois for at least two consecutive years. Table 5.6 contains data on the program.

Student-to-Student Grant Program: The Student-to-Student Grant Program allows voluntary student contributions to be matched dollar-for-dollar, by ISAC, and paid to participating public universities and community colleges. Need-based grants are then made available to students who qualify. In FY2011, participating institutions provided a total of \$1,147,260 and ISAC paid an equal amount for 3,103 students. Table 5.0d contains additional data on this program.

State Scholar Program: Each year ISAC selects approximately ten percent of the high school graduates in Illinois to be State Scholars. These students are chosen according to their high school class rank and the scores of ACT or SAT tests taken during the third semester prior to graduation from high school. The State Scholar program is a recognition-only program. In FY2011, there were 17,476 Illinois State Scholars. Table 5.2 gives historical data on this program.

Bonus Incentive Grant Program: The Bonus Incentive Grant (BIG) Program provides an additional financial incentive to encourage the use of Illinois College Savings Bond proceeds for attendance at Illinois colleges and universities. The incentive grants range from \$40 to \$440 per \$5,000 of compound accreted value at maturity, depending on the maturity of the bond. Students must be enrolled at least half time at an undergraduate or graduate level at a MAP-approved Illinois postsecondary institution. Table 5.3 provides historical data concerning this program.

(continued)

Golden Apple Illinois Scholars: The Golden Apple Illinois Scholars program recruits and prepares bright and talented high school graduates who represent a rich ethnic diversity for successful teaching careers in high-need schools throughout Illinois and provides scholarships to students pursuing teaching degrees. Illinois high school students must be nominated by a teacher, counselor, principal, or other non-family adult. Students may also nominate themselves. A limited number of positions in the program are also open to college sophomores in attendance at one of 54 Illinois colleges and universities that partner in the program. Scholars receive financial assistance for four years in exchange for a commitment to teach for five years in an Illinois school of need. Table 5.0b and 5.0c provides FY2011 data for this program.

Higher Education License Plate Program: The Higher Education License Plate Program provides grants to students who attend Illinois colleges for which the special collegiate license plates are available. The Illinois Secretary of State issues the license plates, and part of the proceeds are used for grants for undergraduate students attending these colleges. Program grants may be used only for tuition and mandatory fees for two semesters or three quarters in an academic year. The number of grants, as well as the individual dollar amount awarded, are subject to sufficient annual appropriations by the Illinois General Assembly and the governor. Table 5.7 provides historical data on this program.

Illinois Optometric Education Scholarship Program: An Optometric Education Scholarship Program recipient must practice in Illinois as a licensed optometrist for a period of not less than one year for each year of scholarship assistance received. The recipient must also begin practicing optometry in Illinois within one year following completion of the academic program for which he or she was awarded the scholarship and practice on a continuous basis until the obligation is fully completed. If the requirement to practice optometry in Illinois is not fulfilled, the scholarship converts to a loan and the recipient must repay the entire amount of the scholarship prorated to the fraction of the obligation not completed, plus interest at a rate of 5 percent. Table 5.8 provides program data.

Nurse Educator Scholarship Program: This scholarship pays tuition and fees for approved graduate level programs at eligible Illinois colleges for up to four calendar years for nurses and academically talented students pursuing a career in professional or practical nursing education in Illinois. In addition, qualified applicants also receive a stipend of up to \$10,000 to cover the cost of attendance, including living expenses. Recipients must sign an agreement to teach in Illinois in the field of nursing education. If the teaching commitment is not fulfilled, the amount of the proceeds received (including scholarship and stipend dollars) converts to a loan and the entire amount plus interest and reasonable collections costs must be repaid. The program was authorized only for the 2007-2010 academic year. Therefore, funding was not appropriated for this program in the FY2011 state budget.

Nurse Educator Loan Repayment Program: In an effort to address the shortage of nurses and the lack of instructors to staff courses teaching nursing in Illinois, the Nurse Educator Loan Repayment Program encourages longevity and career change opportunities. The program is intended to pay eligible loans to add an incentive to nurse educators in maintaining their teaching careers within Illinois. The annual awards to qualified nurse educators may be up to \$5,000 to repay their student loan debt, and may be received for up to a maximum of four years. Recipients must be a U.S. citizen or eligible non-citizen, an Illinois resident, meet licensing requirements, and work as a nurse instructor in an approved Illinois institution for at least 12 consecutive months prior to the date of application. Table 5.9 provides data for this program.

<u>Veterans' Home Nurse Loan Repayment Program</u>: The Veterans' Home Nurse Repayment Program provides for the payment of eligible educational loans as an incentive for nurses to pursue and continue their careers at Illinois veterans' homes. The annual award to qualified registered professional nurses and licensed practical nurses may be up to \$5,000 to repay their student loan debt. This award may be received for up to a maximum of four years. Recipients must be a U.S. citizen or eligible non-citizen, an Illinois resident, and meet licensing requirements. They must have completed the prescribed employment probationary period and be employed in good standing. For each year during which an award is received, the recipient must fulfill a separate 12 month period as a registered professional nurse practical nurse in an approved Illinois veterans' home. Table 5.10 provides data for this program.

(continued)

John R. Justice Student Loan Repayment Program: This program through the Department of Justice, is administered by the Illinois Student Assistance Commission and is used to help prosecutor and public defender offices retain and attract talented attorneys by providing student loan repayment assistance to state and federal public defenders and state prosecutors who agree to remain in their positions for at least three years. The annual awards to qualified defenders and prosecutors may be up to \$4,000, with an aggregate total of up to \$60,000, to repay their student loan debt. The number of awards made through this program, as well as the individual dollar amount awarded, are subject to sufficient annual appropriations by the federal government. Recipients must be a U.S. citizen or eligible non-citizen, an Illinois resident, and meet additional program criteria. If the employment commitment is not fulfilled, any amount received must be repaid. Table 5.12 provides summary data on this program.

Table 5.0a of the 2011 ISAC Data Book Specialized Scholarship and Grant Programs Award and Payout Summary, FY2007-FY2011

_	2	007	20	08	20	009	20	10	201	1
<u>Program</u>	# Awards	\$ Payout	# Awards	\$ Payout						
Illinois Veteran Grant Program	11,781	\$19,176,749	11,462	\$19,223,139	11,372	\$19,241,660	11,450	\$16,812,105	9,869 **	
Illinois National Guard Grant Program	1,821	\$4,462,432	1,897	\$4,470,595	1,403	\$4,479,653	1,931	\$4,729,653	1,294	\$2,691,922
Dependents' Grant Programs	71	\$454,447	74	\$469,426	70	\$465,241	86	\$710,192	87	\$803,741
Student-to-Student Grant Program	3,080	\$949,975	3,028	\$949,994	3,248	\$949,979	3,012	\$948,281	3,103	\$1,147,260
Robert C. Byrd Honors Scholarship Prog	1,037	\$1,545,500	1,067	\$1,571,593	1,224	\$1,809,465	1,322	\$1,955,876	1,316	\$1,942,500
Higher Education License Plate Program	211	\$52,000	207	\$51,675	212	\$53,075	274	\$68,425	280	\$70,000
Optometric Education Scholarship Program	10	\$50,000	10	\$50,000	10	\$50,000	10	\$50,000	10	\$50,000
Veterans' Home Nurse Loan Repayment Program			18	\$44,313	14	\$36,343	9	\$20,141	4	\$14,319
John R Justice Student Loan Repayment Program									130	\$363,859
Bonus Incentive Grant Program *#	942	\$648,920	427	\$322,220	367	324,060	262	206,440	373	\$331,100
Merit Recognition Scholarship Program *										-
<u> Teacher Programs</u>										
Ilinois Future Teachers Corp Program	564	\$3,824,127	549	\$3,823,541	518	\$3,563,367	121	\$872,992	26	\$183,420
IFTC / Golden Apple							146	\$995,111	163	\$1,138,649
Illinois Scholars (Golden Apple)	349	\$3,200,000	342	\$2,900,000	336	\$3,300,000	195	\$2,964,479	204	\$3,361,100
Minority Teachers of IL Scholarship Prog	601	\$2,891,974	583	\$2,715,787	558	\$2,662,575	459	\$2,158,821	241	\$1,160,922
Ilinois Teachers Loan Repayment Program	112	\$485,146	111	\$498,926	114	\$499,974	116	\$494,702	107	\$495,22
Ilinois Special Education Teacher Tuition Waiver	245	N/A	245	N/A	243	N/A	244	N/A	238	N/A
Nurse Educator Scholarship Program	28	\$446,792	79	\$1,202,133	75	\$1,284,314	37	\$532,068		-
Nurse Educator Loan Repayment Program			54	\$239,686	77	\$377,181	62	\$300,000	61	\$297,23

^{*} Awards provided to top 2 1/2 percent of high school graduating classes in FY1998 and FY1999, top 5 percent beginning in FY2000, and changed to top 3 1/2 percent in FY2003. The MRS Program was expanded in FY2001 to include students whose college admission test scores were in the 95th percentile of the state's scores or who ranked in the top 5 percent of their high school class. The MRS Program has not been funded since the 2005-06 academic year.

^{*#} No appropriation was initially made for the BIG Program for FY2005. The lack of initial funding led to a reduction in the number of claims.

^{**} In FY2011, the IVG Program was not funded. Awards reflect student beneficiaries of tuition waivers at institutions.

Table 5.0b of the 2011 ISAC Data Book Specialized Scholarship and Grant Programs Award and Payout Summary by Sector FY2007-FY2011

<u>Programs</u>	2	007	2	008	2	2009	2	2010	2	011
	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout
Illinois National Guard Grant Program										
Public 4-Year	833	\$3,230,805	865	\$3,244,168	645	\$3,372,923	708	\$3,241,472	533	\$1,848,422
Public 2-Year	988	\$1,231,627	1,032	\$1,226,427	758	\$1,106,729	1,223	\$1,488,181	761	\$873,500
Total All Sectors	1,821	\$4,462,432	1,897	\$4,470,595	1,403	\$4,479,652	1,931	\$4,729,653	1,294	\$2,721,922
Illinois Veteran Grant Program									*	
Public 4-Year	4,321	\$12,917,195	4,336	\$13,577,841	4,360	\$13,845,259	4,176	\$14,810,230	3,693	
Public 2-Year	7,460	\$6,259,554	7,126	\$5,645,298	7,012	\$5,396,402	7,274	\$2,001,875	6,176	
Total All Sectors	11,781	\$19,176,749	11,462	\$19,223,139	11,372	\$19,241,661	11,450	\$16,812,105	9,869	\$0
* In FY2011, the IVG Program was not funded.	Awards reflect stu	ident beneficiaries	of tuition wai	vers at institution	s.					
Robert C. Byrd Honors Scholarship Program										
Public 4-Year	226	\$337,375	230	\$336,188	252	\$367,375	269	\$407,250	260	\$383,438
Private 4-Year	161	\$237,250	163	\$237,000	199	\$290,750	237	\$355,500	249	\$361,250
Public 2-Year	3	\$4,500	4	\$5,625	6	\$8,500	5	\$6,750	3	\$4,500
Proprietary									1	\$1,500
Out of State	647	\$966,375	670	\$992,780	767	\$1,142,840	811	\$1,186,376	803	\$1,192,812
Total All Sectors & Out of State	1,037	\$1,545,500	1,067	\$1,571,593	1,224	\$1,809,465	1,322	\$1,955,876	1,316	\$1,943,500
* In FY2006, institutional recipient data becar	ne available.									
Dependents' Grant Programs										
Public 4-Year	27	\$211,345	27	\$200,218	31	\$218,158	33	\$279,937	32	\$332,965
Private 4-Year	19	\$200,306	26	\$228,550	21	\$220,932	23	\$343,802	21	\$367,888
Public 2-Year	24	\$31,436	19	\$19,869	18	\$26,151	29	\$67,399	33	\$75,534
Private 2-Year			1	6,032						
Hospital Schools	1	\$11,360								
Proprietary Schools			1	\$14,757			1	\$19,054	1	\$27,354
Total All Sectors	71	\$454,447	74	\$469,426	70	\$465,241	86	\$710,192	87	\$803,741

Table 5.0b, Specialized Programs Summary by Sector, continued 2011 ISAC Data Book

Programs	2007		2008		2009		2010		2011	
	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout
Minority Teachers of IL Scholarship Program										
Public 4-Year	385	\$1,861,547	343	\$1,623,818	334	\$1,591,368	257	\$1,222,063	139	\$667,500
Private 4-Year	198	\$947,467	232	\$1,058,340	204	\$995,002	192	\$886,758	98	\$477,422
Public 2-Year	17	\$77,960	8	\$33,629	14	\$57,871	7	\$35,000	4	\$16,000
Private 2-Year	1	\$5,000								
Proprietary					6	\$18,334	3	\$15,000		
Total All Sectors	601	\$2,891,974	583	\$2,715,787	558	\$2,662,575	459	\$2,158,821	241	\$1,160,922
Ilinois Future Teachers Corp Program										
Public 4-Year	310	\$2,107,660	288	\$2,017,474	291	\$2,032,546	66	\$479,240	11	\$75,919
Private 4-Year	254	\$1,716,467	261	\$1,806,067	223	\$1,490,821	54	\$388,752	15	\$107,501
Proprietary					5	\$40,000	1	\$5,000		
Total All Sectors	0	\$0	549	\$3,823,541	519	\$3,563,367	121	\$872,992	26	\$183,420
IFTC / Golden Apple Public 4-Year Private 4-Year Total All Sectors	 	 	 	 	105 55 160	\$262,500 \$137,500 \$400,000	91 55 146	\$632,611 \$362,500 \$995,111	85 78 163	\$553,197 \$585,452 \$1,138,649
Nurse Educator Scholarship Program Public 4-Year	_		2	\$45,386	3	\$47,706	3	\$54,872	_	_
Private 4-Year	17	\$231,344	57	\$828,672	52	\$867,295	25	\$339,411		
Hospital Schools	11	\$215,448	20	\$328,075	20	\$369,313	9	\$137,785		
Total All Sectors	28	\$446,792	79	\$1,202,133	75	\$1,284,314	37	\$532,068		
Student-to-Student Grant Program										
Public 4-Year	2,495	\$816,857	2,753	\$870,809	3,002	\$882,446	2,777	\$870,236	2,804	\$1,061,472
Public 2-Year	585	\$133,118	270	\$79,185	246	\$67,533	235	\$78,045	299	\$85,789
Total All Sectors	3,080	\$949,975	3,023	\$949,994	3,248	\$949,979	3,012	\$948,281	3,103	\$1,147,261

Table 5.0c of the 2011 ISAC Data Book Minority Teachers of Illinois Scholarship Program, Ilinois Future Teachers Corp Program IFTC/Golden Apple Illinois Scholars Program, and Nurse Educator Scholarship Program FY2011 Recipients and Payout by Institution

Public 4-Year

		Minority Teachers of Illinois Schlrshp Prog		IL Future Teacher Corp Program		IFTC / Golden Apple IL Scholars Program	
MAP							
Code	<u>Institution</u>	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout
010	Chicago State University	68	\$315,000	4	\$33,557	1	\$5,000
014	Eastern Illinois University	7	\$35,000	1	\$7,362	10	\$50,000
129	Governors State University	6	\$30,000	1	\$10,000		
022	Illinois State University	11	\$60,000			34	\$242,233
079	Northeastern Illinois University	6	\$30,000	1	\$10,000	5	\$30,000
045	Northern Illinois University	13	\$57,500	1	\$5,000	10	\$58,677
060	Southern IL University-Carbondale	2	\$10,000			3	\$24,780
070	Southern IL University-Edwardsville	10	\$50,000	1	\$2,500	4	\$30,000
064	University of IL - Chicago	6	\$30,000	2	\$7,500	3	\$12,564
127	University of IL - Springfield					1	\$2,500
065	University of IL - Urbana	9	\$45,000			12	\$80,000
066	Western Illinois University	1	\$5,000			2	\$17,443
	Total Public 4-Year	139	\$667,500	11	\$75,919	85	\$553,197

			Minority Teachers of Illinois Schlrshp Prog		e Teacher Program	IFTC / Golden Apple * IL Scholars Program		
MAP		(•		
<u>Code</u>	<u>Institution</u>	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	
001	Augustana College					1	\$5,000	
002	Aurora University	4	\$18,254			2	\$15,000	
058	Benedictine University	3	\$15,000	1	\$5,000			
006	Bradley University	5	\$25,000			5	\$45,000	

Table 5.0c, Recipients and Payout by Institution, continued 2011 ISAC Data Book

		Minority Teachers of Illinois Schlrshp Prog			e Teacher Program	IFTC / Golden Apple IL Scholars Program		
MAP Code	Institution	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	
090	Columbia College	<u>π Awarus</u>	\$5,000	# Awards	<u>Φ1 ayout</u>	$\frac{\# \text{Awards}}{2}$	\$10,000	
011	Concordia University	10	\$50,000	1	\$5,000	2	\$20,000	
013	DePaul University	6	\$26,667	1	\$1,666	9	\$46,667	
055	Dominican University	1	\$5,000		φ1,000	9	\$82,117	
016	Elmhurst College	1	\$5,000			6	\$45,000	
501	Erikson Institute							
017	Eureka College			1	\$5,000			
019	Greenville College	2	\$7,500					
020	Illinois College	1	\$5,000					
021	Illinois Institute of Technology							
023	Illinois Wesleyan University					3	\$30,000	
083	Judson University							
026	Knox College					2	\$10,000	
027	Lake Forest College							
029	Lewis University	1	\$5,000			3	\$25,000	
102	Lexington College							
031	Loyola University Chicago	1	\$5,000			7	\$60,000	
034	Macmurray College					2	\$10,000	
033	McKendree College	1	\$5,000					
036	Millikin University							
038	Monmouth College	2	\$10,000			6	\$40,000	
043	National-Louis University	13	\$70,001	6	\$63,335	5	\$56,668	
044	North Central College					2	\$10,000	
046	North Park University	2	\$10,000			1	\$5,000	
048	Northwestern University							
049	Olivet Nazarene University	1	\$5,000			1	\$5,000	
052	Quincy University	12	\$57,500	2	\$10,000			
053	Rockford College					1	\$5,000	
054	Roosevelt University	5	\$22,500	1	\$5,000	2	\$10,000	
069	St. Xavier University	4	\$20,000	2	\$12,500	4	\$30,000	
062	The University of Chicago							

Table 5.0c, Recipients and Payout by Institution, continued 2011 ISAC Data Book

		Minority Teachers of Illinois Schlrshp Prog			e Teacher Program	IFTC / Golden Apple IL Scholars Program		
MAP						-		
Code	<u>Institution</u>	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	
076	Trinity Christian College	5	\$20,000			1	10000	
081	Trinity International University							
057	University of St. Francis	4	\$20,000					
102	Vandercook College of Music	13	\$65,000			2	\$10,000	
	Total Private 4-year	98	\$477,422	15	\$107,501	78	\$585,452	

Public 2-Year

		Minority Teachers of Illinois Schlrshp Prog		IL Future Teacher Corp Program		IFTC / Golden Apple IL Scholars Program	
MAP				l —			
Code	<u>Institution</u>	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout
103	Blackhawk College						
106	Carl Sandburg Junior College						
032	College of DuPage						
074	College of Lake County	1	\$5,000				
012	Danville Area College						
015	Elgin Community College						
147	Frontier Community College						
114	Harold Washington College						
110	Harry S. Truman College						
124	Heartland Community College						
084	Highland Community College						
056	Illinois Central College						
028	Illinois Valley Community College						

Table 5.0c, Recipients and Payout by Institution, continued 2011 ISAC Data Book

Public 2-Year, continued

		Minority Teachers of Illinois Schlrshp Prog			e Teacher Program	IFTC / Golden Apple IL Scholars Program		
MAP	To atituti a a	# A1-	¢ D	# A 1 -	¢ D	# A a d a	¢ Daniert	
Code	Institution	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	
122	John A Logan College	1	\$3,500					
140	John Wood Community College							
024	Joliet Junior College							
037	Kankakee Community College							
008	Kaskaskia Junior College							
116	Kennedy-King College							
009	Kishwaukee College							
105	Lake Land College							
131	Lewis & Clark Community College							
118	Lincoln Land Community College							
126	Lincoln Trail College							
112	Malcolm X College	1	\$2,500					
120	McHenry County College							
121	Moraine Valley Community College							
040	Morton College							
130	Oakton Community College							
115	Olive Harvey College							
108	Olney Central College							
107	Parkland College							
073	Prairie State College							
041	Rend Lake College							
111	Richard J. Daley College							
133	Richland Community College							
085	Rock Valley College							
088	Sauk Valley Community College							
075	Shawnee Community College							
063	South Suburban College of Cook Cty	1	\$5,000					

Table 5.0c, Recipients and Payout by Institution, continued 2011 ISAC Data Book

Public 2-Year, continued

		•	Minority Teachers of Illinois Schlrshp Prog Corp Prog				Golden Apple * blars Program	
MAP								
Code	<u>Institution</u>	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	
078	Southeastern Illinois College							
004	Southwestern Illinois College							
077	Spoon River College							
047	Triton College							
082	Wabash Valley College							
096	Waubonsee Community College							
117	Wilbur Wright College							
087	William Rainey Harper College							
	Total Public 2-Year	4	\$16,000	0	\$0	0	\$0	

		•	Minority Teachers of Illinois Schlrshp Prog		e Teacher Program	IFTC / Golden Apple * IL Scholars Program		
MAP		-						ı
<u>Code</u>	<u>Institution</u>	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	ĺ
030	Lincoln College							ı
145	Morrison Institute							ı
061	Springfield College in Illinois							ı
								ĺ
	Total Private 2-Year	0	\$0	0	\$0	0	\$0	ĺ

Table 5.0c, Recipients and Payout by Institution, continued 2011 ISAC Data Book

Proprietary Schools

	Minority	Minority Teachers		IL Future Teacher		IFTC / Golden Apple *		
	of Illinois Scl	of Illinois Schlrshp Prog		Corp Program		s Program		
MAP		<u> </u>					ı	
Code <u>Institution</u>	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	ı	
025 Kendall College								
Total Proprietary	0	\$0	0	<u>-</u> \$0	0	\$0		

Hospital Schools

		Minority Teachers of Illinois Schlrshp Prog		IL Future Teacher Corp Program		IFTC / Golden Apple IL Scholars Program		
MAP								ĺ
Code	<u>Institution</u>	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	ĺ
389	Rush University Medicine							ĺ
321	St. Francis Med Ctr College of Nursing							ĺ
	Total Hospital Schools	0	\$0	0	\$0	0	\$0	

U.S. Service Academies

		Minority Teachers of Illinois Schlrshp Prog		IL Future Teacher Corp Program		IFTC / Golden Apple * IL Scholars Program		
MAP								
Code	<u>Institution</u>	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	
201	United States Air Force Academy							
202	United States Coast Guard Academy							
203	United States Military Academy							
204	United States Naval Academy							
	Total U.S. Service Academies	0	\$0	0	\$0	0	\$0	
	Grand Total	241	\$1,160,922	26	\$183,420	163	\$1,138,649	

^{*}Represents a portion of Illinois Scholars paid through IFTC

Table 5.0d of the 2011 Data Book
Illinois Veteran Grant Program, Illinois National Guard Grant Program
Grants for Dependents of Police/Fire/Correctional Officers Program, and the Student-to-Student Grant Program
FY2011 Recipients and Payout by Institution

Public 4-Year

		Illinois Veteran Grant Program*		Illinois National Guard		Police/Fire/Correctional Dependents Grant Prog		Student-to-Student	
		Grant Pr	ogram*	Grant	Program	Dependents	Grant Prog	Grant	Program
MAP									ISAC
Code	Institution	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	Match Funds
010	Chicago State University	212		28	\$72,654				
014	Eastern Illinois University	152		36	\$103,740	3	\$27,954		
129	Governor State University	268		13	\$30,139				
022	Illinois State University	275		68	\$265,215	8	\$81,962	585	\$158,408
079	Northeastern Illinois University	245		9	\$26,466				
045	Northern Illinois University	431		41	\$130,165	4	\$41,410	337	\$176,527
060	Southern Illinois UnivCarbondale	611		76	\$279,801	3	\$18,062	173	\$85,887
070	Southern Illinois Univ -Edwardsville	311		81	\$206,124	2	\$14,606	223	\$124,644
064	University of Illinois-Chicago	351		38	\$253,355	4	\$45,252	167	\$82,956
127	University of Illinois-Springfield	281		35	\$88,069			65	\$38,899
065	University of Illinois-Urbana	243		30	\$155,472	7	\$95,530	1,254	\$394,151
066	Western Illinois University	313		78	\$237,221	1	\$8,189		
	Total Public 4-Year	3,693	\$0	533	\$1,848,422	32	\$332,965	2,804	\$1,061,472

		Illinoi	Illinois Veteran		Illinois National Guard		Correctional	Student-to-Student	
		Grant	Grant Program*		Grant Program		Grant Prog	Grant Program	
MAP									ISAC
Code	<u>Institution</u>	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	Match Funds
001	Augustana					1	\$18,236		
002	Aurora University								
058	Benedictine University								
005	Blackburn College								
006	Bradley University					1	\$20,956		

Table 5.0d, Recipients and Payout by Institution, continued 2011 ISAC Data Book

							ice/Fire/Correctional pendents Grant Prog		to-Student t Program
MAP									ISAC
Code	<u>Institution</u>	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	Match Funds
090	Columbia College					1	\$18,236		
011	Concordia University								
013	DePaul University					6	\$88,544		
055	Dominican University								
150	East West University								
016	Elmhurst College					3	\$54,708		
017	Eureka College								
019	Greenville College								
098	Hebrew Theological College								
020	Illinois College								
231	Illinois Institute of Technology					1	\$27,236		
023	Illinois Wesleyan University					1	\$18,236		
083	Judson University								
025	Kendall College								
026	Knox College					1	\$18,236		
027	Lake Forest College								
029	Lewis University					1	\$18,236		
091	Lincoln Christian College								
031	Loyola University Chicago					1	\$18,236		
034	MacMurray College								
033	McKendree College								
036	Millikin University								
038	Monmouth College								
043	National Louis University								
044	North Central College								
046	North Park University								
048	Northwestern University								
049	Olivet Nazarene College								
052	Quincy University								
007	Robert Morris College								
053	Rockford College								
054	Roosevelt University					1	20,106		

Table 5.0d, Recipients and Payout by Institution, continued 2011 ISAC Data Book

Private 4-Year, continued

			Illinois Veteran Grant Program*		Illinois National Guard Grant Program		Police/Fire/Correctional Dependents Grant Prog		to-Student Program
MAP									ISAC
Code	<u>Institution</u>	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	Match Funds
059	Shimer College								
069	St. Xavier University					3	\$46,922		
068	The School of the Art Institute								
062	The University of Chicago								
076	Trinity Christian College								
081	Trinity International University								
057	University of St. Francis								
102	Vandercook College of Music								
067	Wheaton College								
	Total Private 4-year	0	\$0	0	\$0	21	\$367,888	0	\$0

Public 2-Year

			Illinois Veteran		Illinois National Guard		Police/Fire/Correctional		to-Student
		Grant P	rogram*	Grant Program		Dependents Grant Prog		Grant Program	
MAP									ISAC
Code	<u>Institution</u>	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	Match Funds
103	Black Hawk College	148		20	\$27,797	1	\$4,417		
106	Carl Sandburg College	61		16	\$25,453				
032	College of DuPage	387		15	\$18,569	4	\$10,960	299	\$85,789
074	College of Lake County	246		12	\$12,501	2	\$7,894		
012	Danville Area College	50							
015	Elgin Community College	131		6	\$6,218	2	\$3,531		
147	Frontier Community College	9							
114	Harold Washington College	96		12	\$13,118				
110	Harry S. Truman College	83							
124	Heartland Community College	141		37	\$55,502	1	\$2,830		
084	Highland Community College	57		8	\$8,997	1	\$2,987		
056	Illinois Central College	415		74	\$73,688	1	\$1,792		

Table 5.0d, Recipients and Payout by Institution, continued 2011 ISAC Data Book

Public 2-Year, continued

		Illinois ' Grant Pr		Illinois Natio Grant P		Police/Fire/C Dependents		10 1111111	to-Student Program
MAP									ISAC
Code	<u>Institution</u>	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	Match Funds
028	Illinois Valley Community College	110		9	\$8,538				
122	John A. Logan College	111		37	\$44,690				
140	John Wood Community College	73		9	\$14,038				
024	Joliet Junior College	250		15	\$12,614				
037	Kankakee Community College	97		10	\$10,633	1	\$2,574		
008	Kaskaskia Junior College	73		22	\$25,956				
116	Kennedy-King College	93		1	\$622				
009	Kishwaukee College	102		14	\$17,580	2	\$2,368		
105	Lake Land College	139		50	\$49,007				
131	Lewis & Clark Community College	105		14	\$18,037				
118	Lincoln Land Community College	239		81	\$70,963	1	\$1,008		
126	Lincoln Trail College	10		1	\$1,811				
112	Malcolm X College	100		3	\$2,836				
120	McHenry County College	126		9	\$6,610	1	\$2,663		
121	Moraine Valley Community College	233		15	\$21,191	8	\$19,417		
040	Morton College	60		9	\$8,081				
130	Oakton Community College	64		3	\$4,467				
115	Olive Harvey College	50		3	\$3,906				
108	Olney Central College	18		2	\$1,792				
107	Parkland College	191		43	\$75,717	1	\$1,326		
073	Prairie State College	159		10	\$9,055	1	\$883		
041	Rend Lake College	96		10	\$12,970				
111	Richard J. Daley College	76		9	\$7,302				
133	Richland Community College	126		10	\$10,343				
085	Rock Valley College	237		35	\$35,213				
088	Sauk Valley Community College	76		20	\$22,401				

Table 5.0d, Recipients and Payout by Institution, continued 2011 ISAC Data Book

Public 2-Year, continued

		Illinois V	Veteran	Illinois Natio	onal Guard	Police/Fire/C	Correctional	Student-	to-Student
		Grant Pr	ogram*	Grant P	rogram	Dependents	Grant Prog	Grant	Program
MAP				-					ISAC
Code	<u>Institution</u>	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	Match Funds
075	Shawnee Community College	31		1	\$1,131				
063	South Suburban College of Chicago	119		2	\$1,878				
078	Southeastern Illinois College	24		5	\$4,959				
004	Southwestern Illinois College	436		56	\$60,762				
077	Spoon River College	35		14	\$21,749				
047	Triton College	174		11	\$16,328				
082	Wabash Valley College	35		2	\$1,988				
096	Waubonsee Community College	158		17	\$16,664	3	\$4,410		
117	Wilbur Wright College	124				1	\$1,331		
087	William Rainey Harper College	202		9	\$9,827	2	\$5,144		
	Total Public 2-Year	6,176	\$0	761	\$873,500	33	\$75,534	299	\$85,789

			Illinois Veteran Grant Program*		onal Guard Trogram	Police/Fire/Correctional Dependents Grant Prog		Student-to-Student Grant Program	
MAP									ISAC
Code	<u>Institution</u>	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	Match Funds
030	Lincoln College								
092	MacCormac College								
145	Morrison Institute of Technology								
061	Springfield College in Illinois								
	Total Private 2-Year	0	\$0	0	\$0	0	\$0	0	\$0

Table 5.0d, Recipients and Payout by Institution, continued 2011 ISAC Data Book

Hospital Schools

		Illinois '	Veteran	Illinois Natio	onal Guard	Police/Fire/C	Correctional	Student-	to-Student
		Grant Pr	rogram*	Grant P	rogram	Dependents	Grant Prog	Grant	Program
MAP									ISAC
Code	<u>Institution</u>	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	Match Funds
358	Blessing Rieman College of Nursing								
308	Graham Hospital School of Nursing								
334	Lakeview College of Nursing								
200	National Univ of Health Sciences								
389	Rush University Medical Technology								
335	Rush University Nursing								
318	St. Anthony College of Nursing								
321	St. Francis Med Ctr College of Nursing								
390	St. John's College of Nursing								
337	West Suburban College								
	Total Hospital	0	\$0	0	\$0	0	\$0	0	\$0

Proprietary

			Illinois Veteran Grant Program*		ional Guard Program	Police/Fire/Correctional Dependents Grant Prog		Student-to-Student Grant Program	
MAP									ISAC
Code	<u>Institution</u>	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	Match Funds
<u>173</u>	Argosy University					1	\$27,354		
176	DeVry University								
146	Illinois Institute of Art								
170	Midstate College								
171	Northwestern Business College								
	Total Proprietary	0	\$0	0	\$0	1	\$27,354	0	\$0
	Grand Total	9,869	\$0	\$1,294	\$2,721,922	87	\$803,741	3,103	\$1,147,261

^{*} In FY2011, the IVG Program was not funded. Awards reflect student beneficiaries of tuition waivers at institutions.

Table 5.1 of the 2011 ISAC Data Book Robert C. Byrd Honors Scholarship Program Summary of Recipients and Payout by Institution, FY2008-FY2011

Public 4-Year

MAP		FY2008	FY2008	FY2009	FY2009	FY2010	FY2010	FY2011	FY2011
Code	<u>Institution</u>	<u>Number</u>	Dollars	<u>Number</u>	Dollars	<u>Number</u>	Dollars	<u>Number</u>	Dollars
014	Eastern Illinois University	4	\$6,000	5	\$6,750	4	\$6,000	4	\$6,000
022	Illinois State University	2	\$3,000	6	\$6,000	8	\$12,000	5	\$7,500
045	Northern Illinois University	3	\$4,500	2	\$3,000			1	\$1,500
060	Southern IL University-Carbondale	5	\$7,500	4	\$6,000	6	\$9,750	6	\$9,000
070	Southern IL University-Edwardsville	5	\$7,500	10	\$15,000	8	\$12,000	10	\$15,000
064	University of Illinois-Chicago	5	\$7,500	9	\$13,500	13	\$19,500	11	\$17,250
127	University of Illinois-Springfield			1	\$1,500	1	\$1,500	1	\$1,500
065	University of Illinois-Urbana	205	\$298,688	213	\$312,625	228	\$345,000	221	\$324,188
066	Western Illinois University	1	\$1,500	2	\$3,000	1	\$1,500	1	\$1,500
	Total Public 4-Year	230	\$336,188	252	\$367,375	269	\$407,250	260	\$383,438

MAP Code	Institution	FY2008 Number	FY2008 Dollars	FY2009 Number	FY2009 Dollars	FY2010 Number	FY2010 Dollars	FY2011 Number	FY2011 Dollars
	·	<u>Ivallibel</u>		<u>Ivallibel</u>		<u>Ivuilibei</u>		Number	
001	Augustana College	6	\$9,000	5	\$7,500	3	\$4,500	3	\$4,500
058	Benedictine University	1	\$1,500	1	\$1,500			1	\$1,500
006	Bradley University	6	\$9,000	8	\$12,000	10	\$15,000	9	\$13,500
011	Concordia University			1	\$1,500	1	\$1,500	1	\$1,500
013	DePaul University			1	\$1,500	1	\$1,500	1	\$1,500
016	Elmhurst College			1	\$1,500	1	\$1,500	1	\$1,500
017	Eureka College	2	\$3,000	2	\$3,000	1	\$1,500	1	\$1,500
019	Greenville College	1	\$1,500	1	\$1,500				
020	Illinois College	1	\$1,500	3	\$4,500	3	\$4,500	5	\$7,500
021	Illinois Institute of Technology	7	\$10,500	8	\$12,000	10	\$15,000	9	\$13,500
023	Illinois Wesleyan University	11	\$16,500	13	\$19,500	18	\$27,000	15	\$22,500
083	Judson University								
026	Knox College	6	\$8,500	6	\$8,000	5	\$7,500	5	\$7,500
027	Lake Forest College								
029	Lewis University	1	\$1,500	1	\$1,500	1	\$1,500		
091	Lincoln Christian College	1	\$1,500						

Table 5.1, Summary of Robert C. Byrd Honors Scholarship Program 2011 ISAC Data Book

MAP Code	Institution	FY2008 Number	FY2008 Dollars	FY2009 Number	FY2009 Dollars	FY2010 Number	FY2010 <u>Dollars</u>	FY2011 Number	FY2011 Dollars
031	Loyola University Chicago	3	\$4,500	11	\$15,750	14	\$21,000	14	\$21,000
033	McKendree College	2	\$3,000	2	\$3,000	2	\$3,000	3	\$3,750
036	Millikin University	2	\$3,000	3	\$4,500	2	\$3,000	1	\$1,500
044	North Central College			2	\$3,000	5	\$7,500	6	\$9,000
048	Northwestern University	58	\$86,000	66	\$98,500	85	\$127,500	103	\$147,000
049	Olivet Nazarene University	5	\$6,750	3	\$4,500	3	\$4,500	5	\$6,750
052	Quincy University								
053	Rockford College							1	\$1,500
081	Trinity International University			1	\$1,500	1	\$1,500		
062	University of Chicago	45	\$63,000	54	\$75,500	61	\$91,500	57	\$83,000
057	University of St. Francis					1	\$1,500	1	\$1,500
067	Wheaton College	5	\$6,750	6	\$9,000	9	\$13,500	7	\$9,750
	Total Private 4-Year	163	\$237,000	199	\$290,750	237	\$355,500	249	\$361,250

Public 2-Year

MAP Code	Institution	FY2008 Number	FY2008 Dollars	FY2009 Number	FY2009 Dollars	FY2010 Number	FY2010 Dollars	FY2011 Number	FY2011 Dollars
		<u>rtuinoer</u>	Donars	1		1		rumber	Donars
124	Heartland Community College			1	\$1,500	1	\$1,500		
084	Highland Community College	1	\$1,500						
056	Illinois Central College			1	\$1,500			\$1	\$1,500
028	Illinois Valley Community College								
122	John A Logan College			1	\$1,500	1	\$1,500	1	\$1,500
131	Lewis & Clark Community College			1	\$1,500				
120	McHenry County College					1	\$1,500	1	\$1,500
107	Parkland College	1	\$1,125						
088	Sauk Valley Community College					1	\$1,500		
078	Southeastern Illinois College	1	\$1,500						
004	Southwestern Illinois College	1	\$1,500	1	\$1,500				
087	William Rainey Harper			1	\$1,000	1	\$750		
	Total Public 2-Year	4	\$5,625	6	\$8,500	5	\$6,750	3	\$4,500

Table 5.1, Summary of Robert C. Byrd Honors Scholarship Program 2011 ISAC Data Book

Proprietary

MAP Code	<u>Institution</u>	FY2008 <u>Number</u>	FY2008 Dollars	FY2009 <u>Number</u>	FY2009 <u>Dollars</u>	FY2010 <u>Number</u>	FY2010 <u>Dollars</u>	FY2011 <u>Number</u>	FY2011 <u>Dollars</u>
146	Illinois Institute of ArtChicago							1	\$1,500
	Total Proprietary							1	\$1,500

	FY	<u> 2008</u>	<u>FY</u>	2009	<u>FY</u>	<u> 2010</u>	<u>FY</u>	<u> 2011</u>	_
Total Recipients and Dollars for all sectors Total Out-of-State Institutions	397 670	\$578,813 \$992,780	457 767	\$666,625 \$1,142,840	511 811	\$769,500 \$1,186,376	513 803	\$750,688 \$1,192,812	
Grand Total All Sectors & Out-of-State	1,067	\$1,571,593	1,224	\$1,809,465	1,322	\$1,955,876	1,316	\$1,943,500	I

Table 5.2 of the 2011 ISAC Data Book Historical Summary of State Scholar Selection Scores, AY1979-80 to AY2011-12

Year Entering College	Number of State Scholars	Mean High School Rank Percentile	Mean Rank Standard Score	Mean Test Score	Mean Selection Score	Selection Score Cutoff	95th Percentile Test Score Cutoff	Students at or above 95% ACT/SAT **
1979-1980	9,862	94.35	23.5	26.9	464.0	415		
1980-1981	10,013	94.35	23.5	26.7	461.0	415		
1981-1982	10,176	94.40	23.1	27.5	476.0	425		
1982-1983	10,105	94.57	23.6	27.8	477.0	425		
1983-1984	10,404	91.04	23.4	28.1	79.6 *	73 *		
1984-1985	10,446	90.84	23.3	28.3	79.8	73		
1985-1986	10,133	91.87	23.7	28.7	81.1	75		
1986-1987	13,917	89.95	23.0	27.9	78.8	72		
1987-1988	14,242	89.94	23.1	27.6	78.2	71		
1988-1989	14,278	90.32	23.2	27.7	78.6	72		
1989-1990	14,634	89.72	23.0	27.6	78.2	71		
1990-1991	13,155	90.33	23.2	27.9	78.9	72		
1991-1992	11,837	90.82	23.3	28.4	80.1	73		
1992-1993	12,026	90.36	23.1	28.4	79.5	73	29	5,507
1993-1994	12,706	90.26	23.1	28.5	79.5	73	29	5,851
1994-1995	13,141	90.14	23.1	28.5	79.5	73	30	6,131
1995-1996	12,401	90.91	23.4	28.8	80.4	74	30	6,491
1996-1997	12,430	90.48	23.1	29.0	80.3	74	30	6,714
1997-1998	12,818	90.18	23.0	28.9	79.9	74	30	4,592
1998-1999	12,924	90.50	23.1	29.3	80.8	75	30	5,626
1999-2000	13,167	91.83	23.5	28.9	52.3 #	48	30	5,314
2000-2001	13,496	91.97	23.5	28.9	52.3	48	30	5,571
2001-2002	13,731	91.44	23.4	29.0	52.3	48	30	6,202
2002-2003	15,407	91.07	23.3	29.2	52.3	48	30	6,996
2003-2004	16,434	90.89	23.2	29.2	52.4	48	30	7,717
2004-2005	15,951	90.99	23.3	29.1	52.3	48	30	7,152
2005-2006	16,516	90.53	23.2	29.3	52.3	48	30	7,988
2006-2007	16,630	90.86	23.2	29.2	52.3	48	30	7,791
2007-2008	17,093	90.27	23.1	29.5	52.4	48	30	8,728
2008-2009	18,178	89.57	23.0	29.5	52.3	48	30	9,450
2009-2010	19,331	89.07	22.9	29.6	52.3	48	30	10,386
2010-2011	19,432	88.89	22.8	29.6	52.3	48	30	10,602
2011-2012	17,548	88.83	22.9	30.1	52.9	49	30	11,002

^{**} Beginning in 1992-93, students who scored in Illinois' 95th percentile or higher on their ACT Assessment or SAT I tests were automatically designated a State Scholar. Thus, these students are excluded from all mean calculations except the mean test score.

[#] Beginning in 1999-00, the selection formula was changed to weight the test and rank scores equally.

Table 5.3 of the 2011 ISAC Data Book: Bonus Incentive Grant Program Recipients and Payout, FY2002-FY2011

	FY2002	FY2003	FY2004	FY2005*	<u>FY2006</u>	FY2007	<u>FY2008</u>	FY2009	FY2010	<u>FY2011</u>
Recipients	1,175	1,436	1,252	399	1,071	942	427	367	262	373
Payout	\$573,720	\$645,540	\$649,880	\$219,000	\$648,820	\$648,920	\$322,220	\$324,060	\$206,440	\$331,100
Maximum Grant Per Bond	\$260	\$280	\$300	\$320	\$340	\$360	\$380	\$400	\$400	\$420

^{*} No appropriation was initially made for this program for FY2005. The lack of initial funding led to a reduction in the number of claims.

Table 5.4a of the 2011 ISAC Data Book Silas Purnell Illinois Incentive for Access Grant Program (IIA) Summary of Awards and Payout by Sector FY2007-FY2011

	2006-2007		 2007-2008		_	2008-2009			2009-2010			2010-2011	
<u>Sector</u>	# Awards	\$ Payout	# Awards	\$ Payout		# Awards	\$ Payout		# Awards	\$ Payout		# Awards	\$ Payout
Public 4-Year	2,577	\$1,143,750	3,087	\$1,387,500		3,304	\$1,517,750		4,150	\$1,037,500			
Private 4-Year	1,793	\$787,250	1,969	\$886,250		1,972	\$898,000		2,016	\$504,000			
Public 2-Year	14,801	\$5,488,750	14,841	\$5,542,250		13,124	\$4,988,000		11,348	\$2,837,000			
Private 2-Year	348	\$141,250	333	\$135,250		380	\$156,250		368	\$92,000			
Hospital	57	\$20,500	77	\$27,250		64	\$23,750		51	\$12,750			
Proprietary	1,133	\$429,000	1,028	\$385,000		1,053	\$348,250		941	\$235,250			
All Sector Total	20,709	\$8,010,500	21,335	\$8,363,500		19,897	\$7,932,000]	18,874	\$4,718,500			

Program Not Funded

Table 5.4b of the 2011 ISAC Data Book Silas Purnell Illinois Incentive for Access (IIA) Grant Program Award and Payout Summary by Institution, FY2008-FY2011

Public 4-Year

MAP		2007-	2008	2008	3-2009	2009	9-2010	NOT FUNDED 2010-2011	
<u>Code</u>	Institution	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout
010	Chicago State University	308	\$125,000	338	\$143,750	302	\$75,500		
014	Eastern Illinois University	189	\$85,000	197	\$91,000	269	\$67,250		
022	Illinois State University	153	\$67,750	175	\$81,750	261	\$65,250		
079	Northeastern Illinois University	302	\$122,000	344	\$149,500	290	\$72,500		
045	Northern Illinois University	396	\$188,750	452	\$214,250	741	\$185,250		
060	Southern Illinois-Carbondale	507	\$225,500	506	\$230,000	553	\$138,250		
070	Southern Illinois-Edwardsville	215	\$95,500	198	\$87,750	267	\$66,750		
064	University of Illinois-Chicago	475	\$219,250	465	\$220,000	684	\$171,000		
127	University of Illinois-Springfield	34	\$15,750	46	\$21,500	41	\$10,250		
065	University of Illinois-Urbana	356	\$176,000	403	\$196,250	525	\$131,250		
066	Western Illinois University	152	\$67,000	180	\$82,000	217	\$54,250		
	Total Public 4-Year	3,087	1,387,500	3,304	\$1,517,750	4,150	\$1,037,500		

								NOT FU	UNDED
MAP		2007-2008		2008-2009		2009-2010		2010-2011	
Code	<u>Institution</u>	# Awards	\$ Payout						
001	Augustana College	25	\$12,500	17	\$8,500	27	\$6,750		
002	Aurora University	19	\$9,000	34	\$16,000	53	\$13,250		
058	Benedictine University	26	\$12,250	53	\$25,250	32	\$8,000		
005	Blackburn College	26	\$11,750	33	\$14,250	21	\$5,250		
006	Bradley University	52	\$26,000	46	\$22,500	93	\$23,250		
090	Columbia College	138	\$63,250	141	\$64,750	125	\$31,250		
011	Concordia University	15	\$5,250	23	\$11,000	36	\$9,000		

Table 5.4b, Silas Purnell IIA Summary of Awards and Payout by Institution, continued 2011 ISAC Data Book

								NOT FU	JNDED
MAP		2007-2008		2008-2009		2009-	2010	2010-	2011
Code	<u>Institution</u>	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout
013	DePaul University	189	\$92,250	182	\$90,000	313	\$78,250		
055	Dominican University	45	\$21,000	55	\$25,500	84	\$21,000		
150	East-West University	281	\$110,500	246	\$92,500	103	\$25,750		
016	Elmhurst College	29	\$12,500	33	\$15,500	41	\$10,250		
017	Eureka College	14	\$6,750	10	\$4,000	19	\$4,750		
019	Greenville College	22	\$9,750	19	\$9,500	15	\$3,750		
098	Hebrew Theological College	2	\$1,000	4	\$2,000	7	\$1,750		
020	Illinois College	22	\$10,500	18	\$9,000	18	\$4,500		
021	Illinois Institute of Tech.	16	\$7,500	29	\$14,250	28	\$7,000		
023	Illinois Wesleyan University	14	\$7,000	8	\$4,000	19	\$4,750		
083	Judson University	9	\$4,000	5	\$2,000				
025	Kendall College *	7	\$2,500						
026	Knox College	10	\$4,750	10	\$4,750	13	\$3,250		
027	Lake Forest College	12	\$5,500	17	\$8,000	30	\$7,500		
029	Lewis University	47	\$22,250	69	\$32,000	48	\$12,000		
160	Lexington College**								
091	Lincoln Christian College	12	\$5,250	7	\$2,500	5	\$1,250		
031	Loyola University Chicago	128	\$60,000	124	\$58,000	81	\$20,250		
034	MacMurray College	14	\$6,500	12	\$4,750	23	\$5,750		
033	McKendree College	36	\$15,250	38	\$17,250	45	\$11,250		
036	Millikin University	35	\$16,750	45	\$21,750	43	\$10,750		
038	Monmouth College	40	\$19,500	36	\$16,500	41	\$10,250		
043	National Louis University	10	\$3,500	8	\$3,250	3	\$750		
044	North Central College	16	\$7,000	29	\$13,500	39	\$9,750		
046	North Park College	25	\$12,250	33	\$15,250	31	\$7,750		
048	Northwestern University	23	\$11,250	14	\$7,000	31	\$7,750		
049	Olivet Nazarene College	33	\$14,750	33	\$15,250	52	\$13,000		
052	Quincy University					29	\$7,250		

^{*} Kendall College became a Proprietary Institution beginning in 2008-2009.

** In FY2011, Lexington College changed from a Private 2-Year to a Private 4-Year Institution.

 $\begin{tabular}{ll} Table 5.4b, Silas Purnell IIA Summary of Awards and Payout by Institution, continued 2011 ISAC Data Book \\ \end{tabular}$

								NOT FU	JNDED
MAP		2007-2008		2008-2009		2009-2010		2010-2011	
Code	<u>Institution</u>	# Awards	\$ Payout						
007	Robert Morris University Illinois	335	\$148,750	285	\$130,250	217	\$54,250		
053	Rockford College	11	\$5,000	10	\$5,000	7	\$1,750		
054	Roosevelt University	23	\$9,500	63	\$27,000	77	\$19,250		
059	Shimer College	2	\$1,000	1	\$500				
152	St. Augustine College**								
069	St. Xavier University	101	\$43,250	74	\$33,750	69	\$17,250		
144	Telshe Yeshiva	3	\$1,500						
068	The School of the Art Institute	2	\$750	7	\$3,000	7	\$1,750		
062	The University of Chicago	17	\$8,500	18	\$9,000	25	\$6,250		
076	Trinity Christian College	17	\$8,000	17	\$8,000	18	\$4,500		
081	Trinity International University	23	\$10,000	7	\$3,000	16	\$4,000		
080	Trinity International University - Reach			3	\$1,500				
057	University of St. Francis	22	\$10,000	18	\$8,750	16	\$4,000		
102	Vandercook College of Music	0	\$0	3	\$1,500	7	\$1,750		
067	Wheaton College	2	\$1,000	1	\$500	9	\$2,250		
	Total Private 4-Year	1,969	\$886,250	1,972	\$898,000	2,016	\$504,000		

^{**} In FY2011, St. Augustine College changed from a Private 2-Year to a Private 4-Year Institution.

Public 2-Year

								NOT FU	UNDED
MAP		2007-2008		2008-2009		2009-2010		2010-2011	
Code	<u>Institution</u>	# Awards	\$ Payout						
103	Black Hawk College	424	\$159,750	269	\$99,000	217	\$54,250		
106	Carl Sandburg College	192	\$72,500	226	\$86,250	159	\$39,750		
032	College of DuPage	484	\$191,750	473	\$198,250	376	\$94,000		
074	College of Lake County	261	\$97,750	246	\$89,500	262	\$65,500		
412	Danville Area Community College			1	\$500				
012	Danville Area Community College	136	\$55,250	90	\$37,000	99	\$24,750		
015	Elgin Community College	217	\$86,000	190	\$79,750	202	\$50,500		
147	Frontier Community College	35	\$12,500	30	\$11,000	21	\$5,250		

Table 5.4b, Silas Purnell IIA Summary of Awards and Payout by Institution, continued 2011 ISAC Data Book

Public 2-Year, continued

IC 2-1	car, continucu							NOT FU	JNDED
MAP		2007-2	2008	2008-	2009	2009	-2010	2010-	
Code	<u>Institution</u>	# Awards	\$ Payout						
114	Harold Washington College	963	\$360,750	707	\$270,250	648	\$162,000		
110	Harry S. Truman College	414	\$153,250	409	\$155,000	353	\$88,250		
124	Heartland Community College	227	\$84,000	181	\$67,750	156	\$39,000		
084	Highland Community College	114	\$41,250	115	\$44,250	100	\$25,000		
056	Illinois Central College	189	\$75,250	244	\$102,500	444	\$111,000		
028	Illinois Valley Community College	164	\$62,500	168	\$67,750	156	\$39,000		
122	John A. Logan College	386	\$142,500	255	\$98,000	222	\$55,500		
140	John Wood Community College	161	\$58,250	132	\$51,750	96	\$24,000		
024	Joliet Junior College	450	\$172,500	453	\$175,000	437	\$109,250		
037	Kankakee Community College	213	\$75,500	240	\$89,000	166	\$41,500		
008	Kaskaskia College	238	\$87,750	192	\$68,250	167	\$41,750		
116	Kennedy-King College	949	\$334,750	922	\$330,250	582	\$145,500		
009	Kishwaukee College	155	\$61,750	104	\$40,000	193	\$48,250		
105	Lake Land College	194	\$57,500	180	\$56,250	168	\$42,000		
131	Lewis & Clark Community College	339	\$127,500	252	\$99,500	185	\$46,250		
118	Lincoln Land Community College	446	\$170,000	380	\$142,250	294	\$73,500		
418	Lincoln Land Com College - Aviation			1	\$500				
126	Lincoln Trail College	57	\$20,250	21	\$6,750	36	\$9,000		
112	Malcolm X College	561	\$203,250	596	\$218,750	466	\$116,500		
120	McHenry County College	103	\$37,750	120	\$44,750	140	\$35,000		
121	Moraine Valley Community College	549	\$216,250	480	\$187,250	392	\$98,000		
040	Morton College	300	\$111,750	218	\$80,250	168	\$42,000		
130	Oakton Community College	192	\$79,500	171	\$73,500	145	\$36,250		
115	Olive Harvey College	407	\$146,750	380	\$140,500	293	\$73,250		
108	Olney Central College	84	\$32,500	65	\$25,500	67	\$16,750		
107	Parkland College	473	\$184,750	435	\$173,250	372	\$93,000		
073	Prairie State College	343	\$128,000	302	\$115,250	160	\$40,000		
041	Rend Lake College	176	\$67,500	184	\$70,500	137	\$34,250		
111	Richard Daley College	337	\$123,250	329	\$120,750	268	\$67,000		
133	Richland Community College	172	\$69,000	48	\$18,750	112	\$28,000		
085	Rock Valley College	408	\$144,000	432	\$157,250	366	\$91,500		

Table 5.4b, Silas Purnell IIA Summary of Awards and Payout by Institution, continued 2011 ISAC Data Book

Public 2-Year, continued

								NOT FU	UNDED
MAP		2007	-2008	2008	3-2009	2009	9-2010	2010-	-2011
Code	<u>Institution</u>	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout
088	Sauk Valley CommunityCollege	143	\$54,750	152	\$55,250	106	\$26,500		
075	Shawnee Community College	175	\$66,750	121	\$45,000	93	\$23,250		
063	South Suburban College of Cook Cty	772	\$272,000	521	\$184,250	425	\$106,250		
078	Southeastern Illinois College	101	\$35,750	122	\$42,750	97	\$24,250		
004	Southwestern Illinois College	563	\$211,750	448	\$180,750	305	\$76,250		
077	Spoon River College	98	\$39,250	95	\$38,500	81	\$20,250		
047	Triton College	365	\$136,500	337	\$125,750	366	\$91,500		
082	Wabash Valley College	44	\$19,000	45	\$15,750	46	\$11,500		
096	Waubonsee Community College	166	\$58,250	188	\$74,250	191	\$47,750		
117	Wilbur Wright College	598	\$223,000	634	\$238,750	516	\$129,000		
087	William Rainey Harper College	303	\$120,250	219	\$94,000	297	\$74,250		
287	William Rainey Harper College			1	\$500				
	Total Public 2-Year	14,841	\$5,542,250	13,124	\$4,988,000	11,348	\$2,837,000		

Private 2-Year

MAP		2007-	2008	2008-	2009	2009.	-2010		UNDED -2011
Code	<u>Institution</u>	# Awards	\$ Payout						
160	Lexington College**	2	\$1,000	5	\$2,000	4	\$1,000		
030	Lincoln College	110	\$48,500	197	\$84,250	142	\$35,500		
092	MacCormac College	9	\$3,250	7	\$3,000	2	\$500		
145	Morrison Institute of Technology	7	\$2,750	8	\$3,500	7	\$1,750		
061	Springfield College in Illinois	15	\$7,250	6	\$2,750	9	\$2,250		
152	St. Augustine College**	190	\$72,500	157	\$60,750	204	\$51,000		
	Total Private 2-Year	333	\$135,250	380	\$156,250	368	\$92,000		

^{**} In FY2011, Lexington College and St. Augustine College changed from Private 2-Year to Private 4-Year Institutions.

Table 5.4b, Silas Purnell IIA Summary of Awards and Payout by Institution, continued 2011 ISAC Data Book

Hospital Schools

MAP		2007-2	2008	2008-	2009	2009-	2010	NOT FU 2010-	-
Code	Institution	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout
394	Advocate Trinity Hospital	2	\$1,000	1	\$500	1	\$250		
172	Capital Area School of Nursing	71	\$24,250	58	\$20,750	48	\$12,000		
308	Graham Hosp/School of Nursing	4	\$2,000	5	\$2,500	2	\$500		
	Total Hospital Schools	77	\$27,250	64	\$23,750	51	\$12,750		

Proprietary Schools

MAP	·	2007	-2008	2008	3-2009	2009	9-2010	NOT FO 2010	UNDED -2011
<u>Code</u>	<u>Institution</u>	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout	# Awards	\$ Payout
400	American Academy of Art					27	\$6,750		
176	DeVry University-Chicago	408	\$133,500	519	\$134,000	451	\$112,750		
180	Fox College			23	\$10,000	19	\$4,750		
25	Kendall College*								
170	Midstate College	68	\$29,000	44	\$19,000	29	\$7,250		
171	Northwestern College	410	\$164,750	278	\$114,250	232	\$58,000		
147	Le Cordon Bleu College of Culinary Arts**	98	\$40,500	62	\$17,250	52	\$13,000		
146	The Illinois Institute of Art-Chicago	29	\$11,250	104	\$44,500	131	\$32,750		
246	The Illinois Institute of Art-Schaumburg	15	\$6,000	18	\$7,250				
	Total Proprietary Schools	1,028	\$385,000	1,048	\$346,250	941	\$235,250		
	GRAND TOTAL	21,335	\$8,363,500	19,897	\$7,932,000	18,874	\$4,718,500		

^{*} Kendall College became a Proprietary Institution beginning in 2008-2009.

^{**} Previously know as The Cooking and Hospitality Institute of Chicago.

Table 5.5 of the 2011 ISAC Data Book Illinois Special Education Teacher Tuition Waiver Program Summary of Recipients FY1996-FY2011

Fiscal Year	Recipients
1996	174
1997	199
1998	232
1999	212
2000	197
2001	238
2002	244
2003	229
2004	236
2005	248
2006	236
2007	245
2008	245
2009	243
2010	244
2011	238

Table 5.6 of the 2011 ISAC Data Book
Illinois Teacher Loan Repayment Program
Summary of Recipients and Payout FY2003-FY2011

Fiscal <u>Year</u>	Recipients		Payout .
			
2003	27	\$154,076	(includes \$19,568 in admin expenses)
2004	29	\$148,886	(includes \$11,448 in admin expenses)
2005	48	\$230,791	
2006	102	\$497,639	
2007	112	\$485,146	
2008	111	\$498,926	
2009	114	\$499,974	
2010	116	\$494,702	
2011	107	\$495,221	

Table 5.7 of the 2011 ISAC Data Book Higher Education License Plate Program (HELP) Summary of Recipients and Payout FY1998-FY2011

Fiscal			
<u>Year</u>	Recipients*	Payout	
1998	29	\$7,250	
1999	124	\$30,950	
2000	145	\$36,175	
2001	175	\$43,850	
2002	183	\$45,725	
2003	213	\$53,325	
2004	212	\$53,100	
2005	206	\$51,650	
2006	207	\$51,750	
2007	211	\$52,000	
2008	207	\$51,675	
2009	212	\$53,075	
2010	274	\$68,425	
2011	280	\$70,000	

^{*} Based on a \$250 award

Table 5.8 of the 2011 ISAC Data Book Optometric Education Scholarship Program Summary of Recipients and Payout FY2004-FY2011

Fiscal <u>Year</u>	<u>Recipients</u>	Payout
2004	10	\$50,000
2005	10	\$50,000
2006	10	\$50,000
2007	10	\$50,000
2008	10	\$50,000
2009	10	\$50,000
2010	10	\$50,000
2011	10	\$50,000

Table 5.9 of the 2011 ISAC Data Book Nurse Educator Loan Repayment Program (NELR) Summary of Recipients and Payout FY2008-FY2011

Fiscal		
<u>Year</u>	<u>Recipients</u>	Payout
2008	54	\$239,686
2009	77	\$377,181
2010	62	\$300,000
2011	61	\$297,236

Table 5.10 of the 2011 ISAC Data Book Veterans' Home Nurse Loan Repayment Program Summary of Recipients and Payout FY2008-FY2011

Fiscal <u>Year</u>	Recipients	<u>Payout</u>
2008	18	\$44,313
2009 *	14	\$36,343
2010	9	\$20,141
2011	4	\$14,319

^{*} Data for 2009 corrected.

Table 5.11 of the 2011 ISAC Data Book Nurse Educator Scholarship Program Summary of Recipients and Payout FY2007-FY2011

Fiscal		
<u>Year</u>	Recipients	Payout
2007	28	\$446,792
2008	79	\$1,202,133
2009	75	\$1,284,314
2010 *	37	\$532,070
2011		

^{*}Last year of program

Table 5.12 of the 2011 ISAC Data Book John R. Justice Student Loan Repayment Program Summary of Recipients and Payout FY2011

Fiscal <u>Year</u>	<u>Recipients</u>	<u>Payout</u>
2011	130	\$363,859

The Illinois Student Ass the program, visit <a href="https://www.529pre-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purple:super-purp-purp-purp-purp-purp-purp-purp-</th><th>sistance Commission administers the College Illinois!® Prepaid Tuition Program. For more information about paidtuition.org Table 6.0 provides information on program enrollment and payout.	
	PART SIX COLLEGE ILLINOIS!® PREPAID TUITION PROGRAM

Table 6.0 of the 2011 ISAC Data Book: *College Illinois!* The Illinois Prepaid Tuition Program FY2007-FY2011 Program Enrollment and Payout

Program Enrollment and Payout

NUMBER CONTRACTS SOLD:		FY2007 4,113	<u>FY2008</u> 4,545	FY2009 3,458	FY2010 2,166	FY2011 1,746
TOTAL CONTRACT SALES:		\$119,276,422	\$130,874,606	\$120,302,509	\$81,734,173	\$69,297,250
CONTRACTS BY TYPE:	University 8 or more Semesters	1,407	1,358	523	377	321
	University 1-7 Semesters	2,162	2,633	1,111	772	618
	Community College 1- 4 Semesters	278	288	315	181	177
	University Plus 8 or more Semesters*			501	257	199
	University Plus 1 -7 Semesters*			755	455	342
	Univ. & Comm. College 4 + 4 Semesters	266	266	217	106	7
	Univ. Plus & Comm. College 4 + 4 Semesters*			36	18	13
CONTRACTS BY BENEFACTORS:	Parents	3,587	3,989	3,057	1,842	1,48
	Grandparents	407	398	302	238	21
	Other	119	158	99	86	4
BENEFICIARIES BY AGE:	0-5 years	44.2%	42.6%	42.6%	46.5%	48.1%
BENEFICIARIES BT AGE.	6-13 years	59.8%	50.3%	51.4%	47.2%	45.89
	14 and over	6.0%	7.0%	6.3%	6.2%	6.19
PAYOUT OF BENEFITS:		\$28,308,969	\$38,789,324	\$53,894,155	\$64,934,779	\$78,825,586

^{*} University semesters were offered as University plans or University Plus plans starting with the 2008-2009 enrollment period.

The primary goal of ISAC's outreach efforts is to help families navigate the steps of the college admission and financial aid processes. Staff offers or participates in a number of college fairs, financial aid nights, financial literacy events and FAFSA (Free Application for Federal Student Aid) completion workshops throughout the State of Illinois.

ISAC's outreach efforts also include personal one-on-one contact with students to help guide them in their efforts to attend college after high school. In 2009, the Illinois Student Assistance Commission introduced the College Illinois! Corps. In FY2011, the name was changed to the ISAC Corps. Corps members, who are recent college graduates, are placed in communities throughout Illinois to help high school students navigate the process of applying to and entering college. Corp members assist in the completion of FAFSAs and help parents and students throughout the high school experience in preparing for college. Table 7.0 provides a summary outreach activities for 2009-2011

PART SEVEN – ISAC STATEWIDE OUTREACH

Table 7.0 of the 2011 ISAC Data Book Summary of ISAC Outreach Activities FY2009-FY2011

<u>Activity</u>	FY2009	FY2010	FY2011
Professional Development Workshops	41	78	90
Financial Aid Presentations	180	582	450
FAFSA Completion Workshops	147	373	518
College Fairs	123	199	148
College Preparation Sessions	0	0	574
Attendance at Outreach Events			
Students	26,675	47,869	66,958
Parents	8,381	24,690	24,263
Professionals	3,609	7,564	8,811